DIARRHEA DURING CRITICAL ILLNESS

DIARRHEA DURING CRITICAL ILLNESS

By JOANNA C. DIONNE, B.N., M.D., M.Sc. FRCP(C)

A Thesis Submitted to the School of Graduate Studies in Partial Fulfillment of the Requirements for the Degree Doctor of Philosophy

McMaster University © Copyright by Joanna C. Dionne, October 2021

McMaster University

DOCTOR OF PHILOSOPHY (2021)

Hamilton, Ontario (Department of Health Research Methods, Evidence and Impact)

TITLE: Diarrhea During Critical Illness

AUTHOR: Joanna C. Dionne, B.N. (University of New Brunswick), M.D. (McMaster University), M.Sc. (McMaster University)

SUPERVISOR: Deborah Cook, M.D., M.Sc., FRCPC OC

NUMBER OF PAGES: 169

ABSTRACT

Diarrhea is common during critical illness; however, the etiology, definitions, incidence and risk factors for diarrhea and its impact on patient important outcomes require further investigation. There are many possible etiologies of diarrhea, including iatrogenic causes such as laxative medications, often administered as part of bowel protocols, as well as *Clostriodiodes difficile* associated diarrhea (CDAD).

This thesis includes 6 chapters that address the knowledge gaps in the literature regarding the epidemiology of diarrhea in the intensive care unit (ICU), the impact of bowel protocols on diarrhea, and CDAD in critically ill adults.

Chapter 1 provides an introduction to gaps in the literature that are addressed by the studies included in this thesis.

Chapter 2 outlines the methodology used to inform the protocol for the Diarrhea, Incidence, Consequences and Epidemiology in the Intensive Care Unit (DICE-ICU) Study.

Chapter 3 reports on the findings of DICE-ICU including the incidence, risk factors, definitions, and outcomes of patients who develop diarrhea in the ICU.

Chapter 4 provides a content analysis of bowel protocols used in multiple ICUs.

Chapter 5 summarizes a nested cohort study addressing the incidence, prevalence,

timing, treatments, and outcomes of CDAD in critically ill patients enrolled in the

PROSPECT Trial.

Chapter 6 summarizes the work and discusses the strengths and limitations, implications and conclusions presented in this PhD thesis.

iii

ACKNOWLEDGEMENTS

It took a global village to make this dissertation and PhD happen!! I have lots of people to thank!

I would not be here and finishing a PhD without the devotion and love from my Mum and Dad, Helen and Robert Dionne. Mum, thank you so much for the never-failing support, love, encouragement, laughs, toodling, funsies and nudging when I needed it! I would be lost without you, Command Central!! I love you to the moon and back. Dad, thank you for your steadfast support, love, cheerleading, funny emails, chats, trips and fixing all the things I can't fix!!! I repeat your advice "Let it Go" a lot. Thank you for thinking I could do anything. I could not have done this without you. I would be lost without you both!

I owe a lot to my big boisterous family! I want to first celebrate those who have helped me grow into the person I am, and who have crossed the Rainbow Bridge. Thank you to my grandparents, Kathleen Smith and Roger Dionne, who I'm sure thought I would end up being a doctor or a Broadway starlet. Thank you for showing me what real work ethic is, and how to laugh, even during the hard times. When in doubt have a cup of tea and a maple donut! Thank you to my uncle, Gerald Smith, who taught me to appreciate an excellent book or script, and to appreciate academia.

To the rest of the FamJAM: Thank you to Jordan Dionne, my youngest brother, who always makes me laugh, even when it's at myself, and who drops everything on the regular to help his sister out. Thank you for thinking I'm awesome when I don't. I owe a huge thank you to my sister Dr. Jennifer Legon for her keen eye, fastidious edits, recipes, funny texts, support and always being the best sister, a gal could ask for. Thank you to my other brothers Jeff Dionne and Jonathan Dionne for their encouragement during the past 5 years. Thank you to my brother-in law Dr. Wynn Legon for the helpful defense tips and support. Thank you to my sister-in-law Annemarie for being the President of my Fan Club! To my niece Shea Legon, thank you for being so joyful and I live for our FaceTime American Girl Doll play sessions. To Brady, Amy, Scott, Bella, Donovan, and Jackson thank you for making your aunt laugh and ensuring I stay "Cool".

To my supervisor Dr. Deborah Cook- Thank you for believing in me. Thank you for the canoe PhD planning rides, dance parties and laughs. Thank you for your mentorship over the last 7 years and guidance on undertaking this mammoth work. Throughout this process I have made a lifelong friend in you!

To my committee: Dr. Lehana Thabane, Dr. David Armstrong, Dr. Roman Jaeschke, and Dr. Paul Moayyedi. Thank you for your continued support and encouragement over the past 5 years (some of you it's been 12 years!!!). Thank you for believing in a project that forged a new discipline in critical care research.

To the DICE-ICU Co-investigators: THANK YOU, THANK YOU, THANK YOU for taking a chance on a newbie investigator and believing in this work.

To my mentors: Dr. Bob Bulat: Thank you Bob for your mentorship, guidance, support, and encouragement. Thank you for always "checking in", your calls, texts and visits always come when I need them the most. Dr. Zena Samaan: Thank you Zena for being such a strong mentor, sponsor and friend. I couldn't have gotten through the last 2 years without you. Thank you for seeing me and encouraging me when I was wavering. Dr. Laura Wagner, I would never have started in research if it weren't for you. Thank you for taking a chance on me all those years ago. You have been a shining example of what is possible. It was so meaningful to me for you to attend my defense!

To my friend and collaborator Dr. Lawrence Mbaugbaw- you have become my "brother from a different mother". I owe you are world of thanks for the friendship, mentorship, support and ALL THE REVISIONS!!! I look forward to many more future collaborations!

To Lorraine Carroll: You could never get rid of me!!! Thank you for your patience, understanding and unbelievable support.

To my besties: Dr. Emilie Belley-Cote- Thank you for the girls' nights, St. Lucia trips, laughs, dinners, and of course "Pink Bubbles" celebrations. Thank you for cheering me on and being my sister of the heart. Thank you to Dr. Emma Mazurek, from the very first day we met in medical school in 2009 to now you have been a steadfast friend, always there with a warm heart and attentive ear, thank you for our girls' nights, spa days and adventures!!

Thank you to Sheila and Roger Legon for reminding me what is truly important, especially in the hard times. Love you both so much!

To my urban family; thank you to Dr. Heather Thomas for putting up with me even when I'm a "disaster", to Dr. Rucha Utgikar for the "Cocktails and Chat" nights, and Dr. Liz Wilcox for the texts and dog visits. Also thank you to my Irish urban family Dr. Andrew Smyth and Mrs. Anne Smyth, thank you for everything. Thank you to Amanda Burns who always thought I would get here, even when we were 15! Thank you to Kathleen Harquail for being a second sister and helping me get through the rough times.

My research family- Thank you to Nicole Zytaruk, France Clarke, Suzanne Duschesne, Diane Heels-Ansdell and Lois Saunders. Thank you for your friendship and answering my million questions.

To my work family- thank you to my work brothers- Dr. Simon Oczkowski, Dr. John Centofanti, Dr. Bram Rochwerg, Dr. Tim Karachi, Dr. Faizan Amin, Dr. Farzin Visram, Dr. Graham Jones, and Dr. Corey Sawchuk. Thank you for the encouragement, friendship and for covering shifts when I needed to finish this beast (AKA my dissertation). I have to also thank my furry family members. "When I needed a hand to hold I found your paw", Thank you Coco, Piper, Pixie, Cito, Zelda and Matilda. You are always the best part of my day.

I dedicate this dissertation to my Mum and Dad, Helen and Robert Dionne. I also dedicate this in loving memory of Roger Legon.

CONTRIBUTIONS BY OTHERS

At the end of each chapter is a full account of authors' contributions.

TABLE OF CONTENTS

Title Pagei
Descriptive Noteii
Abstractiii
Acknowledgementsiv
Contributionsvii
Table of Contentsviii
List of Abbreviationsix
Chapter 1: Introduction1
Chapter 2: Diarrhea: Interventions, Consequences and Epidemiology in the Intensive Care Unit: (DICE-ICU) – A protocol for a Prospective Multicenter Cohort Study
Chapter 3: Diarrhea During Critical Illness: A Multicenter Cohort Study48
Chapter 4: Content Analysis of Bowel Management Protocols for the Management of Constipation in Adult Critically III Patients
Chapter 5: <i>Clostridioides Difficile</i> Infection in Mechanically Ventilated Critically III Patients: A Nested Cohort Study113
Chapter 6: Conclusions152

LIST OF ABBREVIATIONS

ACG= American College of Gastroenterology

AIC= Akaike information criteria

APACHE II= Acute Physiology and Chronic Health Evaluation II Score

Bliss= Bliss Stool Classification System

Bristol= Bristol Stool Chart

CAG= Canadian Association of Gastroenterology

CCCTG= Canadian Critical Care Trials Group

CDAD= Clostridioides difficile associated diarrhea

CDI = Clostridioides difficile infection

CI= Confidence Intervals

CIHR= Canadian Institute for Health Research

DICE-ICU= Diarrhea, Interventions, Consequences and Epidemiology in the Intensive

Care Unit study

ELISA= Enzyme Linked Immunosorbent Assay

EN= Enteral Nutrition

ESCMID= European Society of Clinical Microbiology and Infections Disease

ESICM= European Society of Intensive Care Medicine

H2RA= Histamine-2 Receptor Agonists

HR= Hazard Ratio

IDSA= Infectious Diseases Society of America

ICU= Intensive Care Unit

IQR= Interquartile Range

LAMP= Loop Mediated Isothermal Amplifications

- LOS= Length Of Stay
- MD= Mean Difference
- NAAT= Nucleic Acid Amplification Test
- OR=Odds Ratio
- PCR= Polymerase Chain Reaction
- PEG= Polyethylene Glycol
- PPI= Proton Pump Inhibitor
- PROSPECT=Probiotics to prevent Severe Pneumonia and Endotracheal Colonization

<u>T</u>rial

- PSI= Physicians Services Incorporated
- RCT= Randomized Controlled Trials
- RMA= Regional medical Associates
- RR= Risk Ratio
- SD= Standard Deviation
- SHEA= Society for Healthcare Epidemiology of America
- STROBE= Strengthening The Reporting of Observational Studies in Epidemiology
- VAP= Ventilator Associated Pneumonia
- WHO= World Health Organization

CHAPTER 1 - Introduction

1.1 Background

Critical illness results in dysregulation of many organ systems; however, optimization of cardio-pulmonary function and normalization of hemodynamics often supersedes the management of gastrointestinal dysfunction in this setting and the impact of critical illness on gastrointestinal function remains under-investigated.

Gastrointestinal dysfunction can include issues ranging from vomiting, feeding intolerance and diarrhea to constipation and ileus[1]. These symptoms are common in the intensive care unit (ICU) and are frequent concerns discussed on medical rounds. However, how the interdisciplinary team defines, monitors, and intervenes to address each of these issues is variable. To meet the needs of critically ill patients experiencing gastrointestinal dysfunction, research has sought to examine different aspect of this dysfunction and its prevention and treatment including the determination of risk factors for diarrhea, the development of protocols to mitigate constipation and feeding intolerance and the definition, diagnosis and treatment of *Clostridioides difficile* associated diarrhea (CDAD). This thesis aims to address the gaps in the literature surrounding those issues, including the development and execution of a large international study on the risk factors and consequences of diarrhea in critical illness; performing a content analysis of bowel protocols in the ICU; and conducting a cohort study of CDAD in a large population of critically ill patients.

The remainder of this chapter lays down the context of each of the issues addressed in this thesis, outlining what we know, where the gaps in knowledge exist, and how the thesis attempts to fill the gaps in subsequent chapters 2-5.

1.2 Diarrhea in the critically ill patients

Many challenges exist in investigating diarrhea in the critically ill, including inattention to the problem, variations in definitions applied across studies, and inconsistent research findings on the impact of diarrhea on patient important outcomes. This has had a direct impact on estimating the incidence of prevalence of this problem in the ICU.

The incidence of diarrhea in the ICU has ranged in the literature from 2%-95% [2-4]. This wide variation may in part be to a lack of consistent definitions applied across studies, and ineffective metrics in monitoring patient's bowel habits in the ICU. A systematic review by Hay and colleagues 2019[4] reported that definitions used across studies were variable. Three of 8 studies reported that diarrhea was associated with a greater length of ICU stay, while only 1 of the 8 studies reported a higher mortality[4]. Half of the studies were retrospective cohorts, and only 1 was multicentre [4].

Taito and colleagues[5] performed a systematic review of diarrhea in the ICU examining short term outcomes including ICU and hospital mortality, and length of stay. The authors identified 12 studies including a total sample size of 13,140; 8 of the 12 studies were prospective cohort designs[5]. This meta-analysis found that diarrhea was associated with an increased risk of ICU mortality (risk ratio [RR] 1.43, 95%CI 1.00,1.98), as well as greater lengths of ICU stay (RR 8.08, 95%CI 5.85,10.32) and total hospital stay (RR 9.67, 95%CI 2.17,17.16)[5].

With differences in definitions used, incidence and prevalence reported, and variable findings addressing patient important outcomes associated with diarrhea in the ICU, I designed a multicenter, multinational prospective cohort study to address these

gaps in the literature – the Diarrhea, Incidence, Consequences and Epidemiology in the intensive care unit (DICE-ICU) Study. Specifically, the objectives were to determine the true incidence of diarrhea, assess risk factors for diarrhea, validate a definition of diarrhea in this patient population, and determine its impact on patient outcomes. In this thesis, I will summarize the design and results of this study in 2 separate chapters. In Chapter 2, I will present the protocol for the Diarrhea, Interventions, Consequences and Epidemiology in the Intensive Care Unit (DICE-ICU) Study and, in Chapter 3, I will present the DICE-ICU Study.

1.3 Bowel Protocols in the Intensive Care Unit

Bowel protocols are increasingly being used in the ICU to minimise feeding intolerance and constipation. However, the evidence about whether bowel protocols are beneficial or positively influence patient important outcomes is conflicting. A systematic review of bowel protocols in the ICU [6] identified 4 trials of 534 critically ill patients examining their impact on constipation, feeding intolerance and duration of mechanical ventilation. The interventions tested in the 4 trials included lactulose or polyethylene glycol (PEG) compared to usual care or placebo. The results of this meta-analysis showed that bowel protocols were associated with no reduction in feeding intolerance (RR 0.94, 95%CI 0.62-1.42), constipation (RR 0.50, 95%CI 0.25,1.01) or days of mechanical ventilation (mean difference (MD) 0.01 days, 95%CI –2.67,2.69 days) based on low certainty of evidence [6].

An updated systematic review and meta-analysis [4] examining constipation, diarrhea and bowel protocols in the ICU identified 6 studies, including 3 trials and 3 prospective cohort studies. The studies varied in terms of the laxatives used, including

senna, lactulose, PEG or glycerine compared to placebo, or usual care. The authors found similar results to Oczkowski and colleagues[6]; bowel protocols did not decrease constipation (RR 0.39, 95%CI 0.14,1.05) or duration of mechanical ventilation (weighted MD 0.18, 95%CI –3.25,3.61), but did result in an increased risk of diarrhea (weighted MD 1.58, 95%CI 1.22,2.04). Hays and colleagues[4] did not find a difference in length of ICU stay (weighted MD -0.76, 95%CI –2.27,0.75) or mortality (RR 0.83, 95%CI 0.56,1.22) associated with bowel protocols.

With the growing use of bowel protocols, evidence currently showing no benefit of bowel protocol regimens, and their potential to induce diarrhea, I performed a content analysis of bowel protocols in an international sample of ICUs which were participating in a randomized controlled trial of the probiotic *Lactobacillus rhamnosus GG* compared to placebo. The Probiotics: Prevention of Severe Pneumonia and Endotracheal Colonization Trial (PROSPECT) found no effect for bowel protocols in preventing ICU acquired infections including ventilator associated pneumonia (VAP), other nosocomial infections, or diarrhea [NCT02462590}[7]. The objective of the study was to analyze the use of bowel protocols in clinical practice in centers participating in the trial, specifically, describing the initiation, medications used, escalation approaches, discontinuation criteria, stool assessment methods and contraindications. In chapter 4, I will summarize the design and results of this study.

1.4 CDAD in the Intensive Care Unit

Clostridioides difficile infection (CDI) is associated with significant morbidity and mortality in hospitalized patients [8], including ICU patients who are at an increased risk of contracting the infection [9]. CDI is associated with significant costs to the healthcare

system. Costs of \$8,911 USD to \$30,049 USD [10] for patients with CDI, with an annual estimated cost of 1.1 to 3.2 billion USD per year [11] for CDI associated care. ICU CDI costs have been demonstrated to be as high as \$11,353 USD compared to \$6,028 for patients who do not have CDI [11]. Given the impact on patient outcomes and costs to the system, CDI remains a research priority.

As highlighted previously, diarrhea is common in the ICU, and increases a patient's risk of skin breakdown and dehydration. Although the etiology of diarrhea in this patient population is multifactorial [9, 12], the most concerning diagnosis is CDAD. Common symptoms associated with CDAD, including pain, fever, and leukocytosis [9, 13], may be either masked in the ICU patient, or be contributed to other disease processes, making the diagnosis of CDAD challenging. This can result in unnecessary testing for CDAD and unnecessary contact precautions from an infection control perspective, which may limit patient contact with health care providers and family members, with diverse attendant consequences.

Risk factors for CDAD include antibiotic exposure, morbid state, age, mechanical ventilation, length of stay and acid suppression exposure [9]. Although the effect of some of these predisposing factors such as acid suppression on the risk of CDI is debated, many of these risk factors are frequent in ICU patients[9, 14, 15].

Timing of development of CDAD and quantifying the true incidence of CDAD in the ICU remains a gap in the CDAD literature, as most studies conducted in this population are comprised of retrospective cohorts and cross-sectional studies. More research is required to describe the proportion, including incidence and prevalence, of

patients who develop CDAD prior, during or after ICU admission and its impact on patient important outcomes including mortality.

Severity of CDAD can range from mild to severe complicated CDAD. Different scores exist to determine the severity based on different clinical factors, including the Infectious Disease Society of American (IDSA)[16], the American College of Gastroenterology (ACG)[17] and the European Society of Clinical Microbiology and Infectious Diseases (ESCMID)[18]. Without an ICU specific definition for CDAD, there is not a gold standard for defining severity in this population either. Therefore, I sought to assess the severity of CDAD using these 3 accepted scoring systems in the critically ill. In chapter 5, I will summarize the design and results of this study to determine the incidence, prevalence, severity, timing and treatment of CDAD in the ICU within a nested cohort study with of the PROSPECT trial.

1.5 Conclusions

In Chapter 6, I will summarize the work of this dissertation, strengths and limitations including methodologic challenges, and future work in this research area.

References

- Reintam A, Parm P, Redlich U, Tooding LM, Starkopf J, Kohler F, Spies C, Kern H, (2006) Gastrointestinal failure in intensive care: a retrospective clinical study in three different intensive care units in Germany and Estonia. BMC Gastroenterol 6: 19
- Whelan K, Judd PA, Taylor MA, (2004) Assessment of fecal output in patients receiving enteral tube feeding: validation of a novel chart. Eur J Clin Nutr 58: 1030-1037
- 3. Ronan Thibault SG, Aurelie Cler, Nathalie Delieuvin, Claudia Paula Heidegger, Claude Pichard, (2013) Diarrhoea in the ICU: respective contribution of feeding and antibiotics. Critical Care 17: 1-8
- 4. Hay T, Bellomo R, Rechnitzer T, See E, Ali Abdelhamid Y, Deane AM, (2019) Constipation, diarrhea, and prophylactic laxative bowel regimens in the critically ill: A systematic review and meta-analysis. Journal of Critical Care
- 5. Taito Š, Kawai Y, Liu K, Ariie T, Tsujimoto Y, Banno M, Kataoka Y, (2019) Diarrhea and patient outcomes in the intensive care unit: Systematic review and meta-analysis. J Crit Care 53: 142-148
- Oczkowski SJW, Duan EH, Groen A, Warren D, Cook DJ, (2017) The Use of Bowel Protocols in Critically III Adult Patients: A Systematic Review and Meta-Analysis. Crit Care Med 45: e718-e726
- 7. Johnstone J, Meade M, Lauzier F, Marshall J, Duan E, Dionne J, Arabi YM, Heels-Ansdell D, Thabane L, Lamarche D, Surette M, Zytaruk N, Mehta S, Dodek P, McIntyre L, English S, Rochwerg B, Karachi T, Henderson W, Wood G, Ovakim D, Herridge M, Granton J, Wilcox ME, Goffi A, Stelfox HT, Niven D, Muscedere J, Lamontagne F, D'Aragon F, St-Arnaud C, Ball I, Nagpal D, Girard M, Aslanian P, Charbonney E, Williamson D, Sligl W, Friedrich J, Adhikari NK, Marquis F, Archambault P, Khwaja K, Kristof A, Kutsogiannis J, Zarychanski R, Paunovic B, Reeve B, Lellouche F, Hosek P, Tsang J, Binnie A, Trop S, Loubani O, Hall R, Cirone R, Reynolds S, Lysecki P, Golan E, Cartin-Ceba R, Taylor R, Cook D, Prevention of Severe P, Endotracheal Colonization Trial I, the Canadian Critical Care Trials G, (2021) Effect of Probiotics on Incident Ventilator-Associated Pneumonia in Critically III Patients: A Randomized Clinical Trial. JAMA 326: 1024-1033
- 8. Dodek PM, Norena M, Ayas NT, Romney M, Wong H, (2013) Length of stay and mortality due to Clostridium difficile infection acquired in the intensive care unit. Journal of Critical Care 28: 335-340
- 9. Riddle DJ, Dubberke ER, (2009) Clostridium difficile infection in the intensive care unit. Infect Dis Clin North Am 23: 727-743
- 10. Gupta A, Ananthakrishnan AN, (2021) Economic burden and cost-effectiveness of therapies for Clostridiodes difficile infection: a narrative review. Therap Adv Gastroenterol 14: 17562848211018654
- 11. Bobo LD, Dubberke ER, Kollef M, (2011) Clostridium difficile in the ICU: the struggle continues. Chest 140: 1643-1653
- 12. Dionne JC, Sullivan K, Mbuagbaw L, Takaoka A, Duan EH, Alhazzani W, Devlin JW, Duprey M, Moayyedi P, Armstrong D, Thabane L, Tsang JLY, Jaeschke R,

Hamielec C, Karachi T, Cartin-Ceba R, Muscedere J, Alshahrani MSS, Cook DJ, (2019) Diarrhoea: interventions, consequences and epidemiology in the intensive care unit (DICE-ICU): a protocol for a prospective multicentre cohort study. BMJ Open 9: e028237

- 13. Riddle MS, DuPont HL, Connor BA, (2016) ACG Clinical Guideline: Diagnosis, Treatment and Prevention of Acute Diarrheal Infections in Adults. The American Journal of Gastroenterology 111: 602-622
- 14. Weiss K, Louie T, Miller MA, Mullane K, Crook DW, Gorbach SL, (2015) Effects of proton pump inhibitors and histamine-2 receptor antagonists on response to fidaxomicin or vancomycin in patients with Clostridium difficile-associated diarrhoea. BMJ Open Gastroenterol 2: e000028
- 15. Novack L, Kogan S, Gimpelevich L, Howell M, Borer A, Kelly CP, Leffler DA, Novack V, (2014) Acid suppression therapy does not predispose to Clostridium difficile infection: the case of the potential bias. PLoS One 9: e110790
- 16. Cohen SH, Gerding DN, Johnson S, Kelly CP, Loo VG, McDonald LC, Pepin J, Wilcox MH, Society for Healthcare Epidemiology of America, Infectious Diseases Society of America, (2010) Clinical practice guidelines for Clostridium difficile infection in adults: 2010 update by the Society for Healthcare Epidemiology of America (SHEA) and the Infectious Diseases Society of America (IDSA). Infect Control Hosp Epidemiol 31: 431-455
- 17. Surawicz CM, Brandt LJ, Binion DG, Ananthakrishnan AN, Curry SR, Gilligan PH, McFarland LV, Mellow M, Zuckerbraun BS, (2013) Guidelines for diagnosis, treatment, and prevention of Clostridium difficile infections. Am J Gastroenterol 108: 478-498; quiz 499
- Debast SB, Bauer MP, Kuijper EJ, (2014) European Society of Clinical Microbiology and Infectious Diseases: Update of the Treatment Guidance Document for Clostridium difficile Infection. Clinical Microbiology and Infection 20: 1-26

CHAPTER 2:

Diarrhea: Interventions, Consequences and Epidemiology in the Intensive Care Unit: (DICE-ICU) – A protocol for a Prospective Multicenter Cohort Study (Published *BMJ Open* 2019;**9**:e028237)

Diarrhea: Interventions, Consequences and Epidemiology in the Intensive Care Unit:

(DICE-ICU)—A Protocol for a Prospective Multicenter Cohort Study

JC Dionne^{1,2}, K Sullivan¹, L Mbuagbaw^{2,3}, A Takaoka², EH Duan¹, W Alhazzani^{1,2}, JW Devlin⁴, MS Duprey⁴, P Moayyedi^{1,2,5}, D Armstrong^{1,5}, L Thabane^{2,3}, J Tsang¹, R Jaeschke¹, C Hamielec¹, T Karachi¹, R Cartin-Ceba⁶, J Muscedere⁷, M Alshahrani⁸ & DJ Cook^{1,2} for the DICE Investigators

1.Department of Medicine, McMaster University, Hamilton, Ontario, Canada 2. Department of Health Research Methods, Evidence, and Impact, McMaster University, Hamilton, Ontario, Canada

3. Biostatistics Unit, St Joseph's Healthcare, Hamilton, Ontario, Canada

4. School of Pharmacy, Northeastern University, Boston, Massachusetts, USA

5. Farncombe Family Digestive Health Research Institute, McMaster University, Hamilton, Ontario, Canada

6. Department of Critical Care, Mayo Clinic, Phoenix, Arizona, USA

- 7. Department of Critical Care Medicine, Queen's University, Kingston, Ontario, Canada
- 8. Department of Critical Care, University of Dammam, Dammam, Saudi Arabia

Corresponding Author: Dr. Joanna C. Dionne McMaster University 1280 Main Street West, 2V9 Hamilton, Ontario, Canada L8S 4K1 Email: Joanna.dionne@medportal.ca

Counts:

Abstract: 399 Main Text: 3624 References: 982 Tables: 1 Figures:0 Appendix: 1

Abstract

Introduction

Diarrhea is a frequent concern in the Intensive Care Unit (ICU) and is associated with prolonged mechanical ventilation, increased length of ICU stay, skin breakdown and renal dysfunction. However, its prevalence, etiology, and prognosis in the critically ill have been poorly studied. The primary objectives of this study are to determine the incidence, risk factors, and consequences of diarrhea in critically ill adults. The secondary objectives are to estimate the incidence of *Clostridium difficile*associated diarrhea (CDAD) in ICU patients and to validate the Bristol Stool Chart and Bliss Stool Classification System characterizing bowel movements in the ICU. Our primary outcome is diarrhea. Our secondary outcomes include: CDAD, ICU and hospital mortality, and ICU and hospital length of stay.

Methods and analysis

This international prospective cohort study will enroll patients over 10 weeks in 12 ICUs in Canada, the United States, Poland, and Saudi Arabia. We will include all patients 18 years of age and older who are admitted to the ICU for a least 24 hours and follow them daily until ICU discharge. Our primary outcome is the incidence of diarrhea based on the World Health Organization (WHO) definition, during the ICU stay. Our secondary outcomes include: CDAD, ICU and hospital mortality, and ICU and hospital length of stay. We will use logistic regression to identify factors associated with diarrhea (as defined using WHO criteria) and the kappa statistic to measure agreement on diarrhea rates between the WHO definition and the Bristol Stool Chart and Bliss Stool Classification System.

Ethics and dissemination

The protocol has been approved by the research ethics board of all participating centers. The DICE-ICU Study will generate evidence about diarrhea and its frequency, predisposing factors, and consequences, to inform critical care practice and future research.

Lay Summary

Diarrhea is a frequent clinical problem for hospitalized patients including those who are critically ill in the ICU. Diarrhea can cause complications such as skin damage, dehydration, and kidney problems. It is not clear how common diarrhea is in the ICU, the factors that cause it, or the best way for clinicians to assess it. The DICE-ICU study is an international prospective observational study to examine the frequency, risk factors and outcomes of diarrhea during critical illness.

Funding

DICE-ICU has received funding from Physician's Services Incorporated (PSI), the Canadian Association of Gastroenterology (CAG), Hamilton Regional Medical Associates (RMA) and the Canadian Institutes for Health Research (CIHR).

Article Summary

Strengths and Limitations of this Study

Strengths

- Large prospective, international, multicenter, cohort study of a mixed population of critically ill adults.
- Comprehensive evaluation of diarrhea incidence and its potential risk factors throughout the ICU stay.
- Bedside nurse characterization of all bowel movements with the WHO definition, Bristol Stool Chart and Bliss Stool Classification System to validate these scoring tools in critically ill adults.

Limitations

- Possible missing data to characterize some bowel movements.
- Possible reporting bias or observer bias influencing some data collection.

Introduction

The reported incidence of diarrhea among critically ill patients ranges from 2% to 95% [1, 2]. This wide range is due to the lack of a universally accepted definition in the intensive care unit (ICU). It is often difficult to differentiate true diarrhea (the passage of more than 3 liquid bowel movements per 24 hours) from a change in stool frequency or stool consistency (e.g., looser stools). There is also wide variation in what is considered 'a normal bowel habit' [3], ranging from 2-3 bowel movements per day to 3 bowel movements per week. Such 'normal variation' makes it challenging to define diarrhea and to identify what may be 'abnormal' in the ICU setting. The concept of what constitutes a normal bowel pattern in the ICU has not been well studied. The ideal definition of diarrhea in the ICU remains unclear.

The World Health Organization (WHO) defines diarrhea as the passage of 3 or more liquid stools per day[4]. While simple, and easily applied at the bedside, clinicians rarely refer to this definition in ICU practice. A criticism of the WHO definition is that quantification of stool is not necessarily an accurate indicator of colonic transit time. The most recognized stool evaluation instrument in hospitals is the Bristol Stool [5], comprising 7 categories with a graphical depictions and text descriptions for each category. A Bristol Score of 6 or 7 is classified as diarrhea [6]. The Bristol Stool Chart is a better predictor of whole-intestinal transit time than stool frequency [5]. The Bristol Stool Chart has subsequently been used to define diarrhea by the European Society for Clinical Microbiology and Infectious Disease, and has used the Bristol Stool Chart to define diarrhea for *Clostridioides difficile* infection [7].

The Bliss Stool Classification System is an alternative system initially developed to assess stool consistency in patients with fecal incontinence. The tool has 4 categories with depictions and descriptions and can be applied at the bedside, but with fewer categories; it has a good reliability when used by health care professionals, nursing students and volunteers [8]. Further reliability and validity testing has been performed [8, 9], though this instrument has not been as widely used in research. There are no studies that validate the Bristol Stool Chart or the Bliss Stool Classification System in the ICU setting for either clinical or research purposes.

Antibiotics, antifungal therapy, prokinetics and enteral nutrition may predispose to diarrhea in the critically ill [2]; however, the risk of diarrhea associated with these factors is unclear and poorly quantified due to the retrospective designs and small sample sizes of previous studies. Without strong evidence informing ICU clinicians of the possible etiologies of diarrhea, enteral nutrition is often considered the culprit, and feeds are discontinued [10, 11]. While the enteral route is the preferred method of nutrition delivery in the ICU [12], if diarrhea is misattributed to enteral nutrition, unnecessary feeding interruption may exacerbate caloric and protein deficits.

Studies on the epidemiology of diarrhea in critically ill patients are limited. These studies have explored issues of gastrointestinal failure (e.g., feeding intolerance, gastrointestinal hemorrhage, and ileus) [13], diarrhea in enterally fed critically ill patients [10], or risk factors of diarrhea [14, 15]. Research designs to

date have included database registry studies [15], case-control studies [14], and retrospective audits. Interest in diarrhea has become particularly relevant as enteral nutrition, often considered the cause of diarrhea in the ICU, is used earlier and more often than in the past. Furthermore, there is growing concern about *Clostridioides difficile* associated diarrhea (CDAD) in this setting.

We are conducting a prospective multicenter study with the following objectives: to determine the incidence and frequency of diarrhea, risk factors for diarrhea and consequences (ICU and hospital mortality, ICU and hospital length of stay) of diarrhea in critically ill adults and validate different stool classification systems. The primary outcomes are to determine the frequency and the incidence of diarrhea, defined using the WHO criteria and risk factors for diarrhea in this patient population. The secondary outcomes are to estimate the incidence of CDAD in ICU patients, validate the Bristol Stool Chart and Bliss Stool Classification System for characterizing bowel movements, ICU and hospital mortality, and ICU and hospital length of stay.

Methods and Analysis

Design

The DICE-ICU Study is a 10-week prospective cohort study of consecutively admitted critically ill patients, and will be conducted at 12 academic and community medical and surgical ICUs in Canada, the United States, Poland, and Saudi Arabia.

Participants

We will include all consecutive patients of 18 years of age or older admitted to the ICU for at least 24 hours, regardless of their mechanical ventilation status. There are no exclusion criteria except patients admitted to the ICU for < 24 hours. At centers with multiple ICUs, we will enroll patients in medical, surgical and mixed ICUs rather than specialized ICUs (e.g., cardiovascular surgery units). In each participating ICU, we will document several center-level variables including the number of ICU and hospital beds, population case-mix, unit design, university affiliation, and use of a 'bowel protocol' (an established order set of prescribed laxatives and/or motility agents with parameters that describe when to use these medications for patients who have not had a bowel movement)[16].

Patient and Public Involvement

Our protocol did not have a patient or patient family member engagement in its development of the research question, ascertainment of outcomes or methodology. Our population of interest is critically ill patients who are either mechanically ventilated, comatose or have altered level of consciousness due to their underlying critical illness condition or associated sedation. Such patient characteristics which typically persist for the majority of their ICU admission preclude meaningful real-time patient engagement as the study progresses. However, our ethics review board includes patient representatives who provided input to the design of the protocol and its implementation. Also, we will disseminate the results of the study to patients, families and citizens

through multimedia methods including pamphlets, social media and research boards in the ICU setting.

Enrolment

Daily, research coordinators will screen all newly admitted patients to each participating ICU who will also document the prior location (e.g., emergency department, operating room, medical or surgical ward), hospital and ICU admission dates. The research coordinators will collect baseline patient characteristics including age, sex and chronic comorbidities (pre-hospital), and Acute Physiology and Chronic Health Evaluation II (APACHE II) score and admitting diagnosis (at ICU admission). Conditions associated with an increased risk for diarrhea (e.g., gastrointestinal bleeding, history of short bowel syndrome, inflammatory bowel disease, history of *Clostridioides difficile* infection, and the presence of ileostomy or colostomy) or a decreased risk of diarrhea (e.g., opiates) (pre-ICU period) will also be documented.

Outcomes

We will perform prospective daily data collection until death in the ICU or ICU discharge. Daily, the bedside nurse will use the case report form refined during the DICE Pilot Study [17] to track all stools. The number and character of each stool will be documented daily using the WHO Definition and the Bristol Stool Chart and Bliss Stool Classification System. We will use these data to ascertain our primary outcome of the incidence and frequency of diarrhea in the study population over a 10 week

period in participating ICUs.

Research coordinators will collect data daily, completing a standardized, previously piloted and refined case report form (CRF)[17]. The CRF (Appendix 1) includes data on: life support utilization (mechanical ventilation, vasopressor usage, renal replacement therapy), laboratory values, physiotherapy, and clinical outcomes. Research coordinators will also document whether nutrition was administered, nutrition formulation, administrative route (enteral or parenteral), infusion rate, and any feeding interruption that may be risk factors for diarrhea (primary outcome). We will document nutritional targets as determined by the ICU dietitians and whether target rates are met. Research coordinators or bedside pharmacists will also track relevant medications (e.g., antibiotics, acid suppressants, antifungal agents, prokinetics, opioids, laxatives and hyperosmolar medications) that may be risk factors for diarrhea (primary outcome). Research coordinators will document the consequences of diarrhea such as electrolyte abnormalities, use of antidiarrheal agents, use of fecal management devices, and diagnostic test ordering (e.g., for *Clostridioides difficile*, malabsorption etc). Research coordinators will collect detailed data on all patients developing CDAD (secondary outcome). We will also document the length of ICU (secondary outcome) and hospital stay (secondary outcome and mortality (ICU and hospital) on all patients enrolled in the study (secondary outcomes).

Data Management

The research coordinator at each site will enter the data locally into a web-

based system (iDatafax, version 4.3.0, 2013) [18]. A Data Manager at the McMaster University DICE-ICU Methods Center will validate all data, ensuring that ambiguous, out of range or missing data are identified and addressed in a timely manner. We will make every attempt to resolve missing data by querying participating centers. If data remain missing, we will address this with the multiple imputation methods, based on the type and distribution of missing data.

Training of Sites

Research coordinators at each site will be oriented by the principal investigator to the data collection forms though site initiation visits in-person or by webinar, and standard operating procedures. At each site, bedside nurses will be oriented to the stool classification systems and trained by the research coordinator on how to record the patient's bowel movements on the case report forms at scheduled sessions and at the bedside. Throughout the study, the Methods Center Data Manager will also give suggestions and feedback to the site research coordinators on data collection to ensure protocol fidelity and uniformity across sites.

Central Adjudication

In duplicate, two independent adjudicators will review all possible cases of CDAD. Patients who have a possible *Clostridium difficile* infection will be adjudicated using the Infectious Disease Society of America criteria [19]. For all possible cases, the following will also be adjudicated: stool frequency,

complications (e.g., colectomy), treatments (antibiotics, surgery), and overall severity according to guidelines of the European Society of Clinical Microbiology and Infectious Diseases [20], Infectious Disease Society of America [19] and American College of Gastroenterology [21].

Sample size

Our sample size estimation is based on two approaches. The first is the standard rule of thumb approach which is based upon the independent factors under examination (in our 4-center DICE Pilot Study, there were 8 independent risk factors for diarrhea) and the number of events required for each degree of freedom which requires 20 events per factor. Using this approach, with 8 independent risk factors and 7 degrees of freedom, we would require 140 patients with diarrhea to examine these factors in a multivariate analysis [22]. Our second approach derives the sample size based upon our DICE Pilot Study with the primary objective of determining independent factors associated with diarrhea during critical illness. We have used the results of our DICE Pilot Study in which antibiotic exposure (main independent variable) was associated with diarrhea (adjusted odds ratio [OR] 2.15 95% CI 1.04-4.4) in the logistic regression to justify the sample size. Our sample size is computed for the research question that would require the largest sample size (which is the diarrhea risk factor analysis), inherently providing sufficient power to address the other objectives. In the DICE Pilot Study, we included the following variables: age, sex, APACHE II Score, and use of relevant drugs (e.g., motility modifiers, stool softeners). Among the 268 patients (80% of total population)

exposed to antimicrobials, 182 (67%) had diarrhea. Among the 67 patients not exposed to antimicrobials, only 19 (28%) had diarrhea (crude OR of 5.34; 95% CI 2.96 – 9.64). We have computed sample sizes for a range of plausible effect sizes (based on confidence intervals from the DICE Pilot Study). Approximately 1000 patients are required to detect an OR as small as 1.6 at level of significance α =0.05 and power (1- β)= 0.8, if 80% of the total population is exposed to antimicrobials (computations by G Power Version 3.1.9.2)[23]. Given the consideration of both approaches, this study will be adequately powered to answer our primary research questions and adequately explore risk factors for diarrhea. Given the observational design, the ultimate sample size will be determined by the number of patients admitted to the ICU in participating centers during the study period; we will target at least 1,000 critically ill adults.

Data Analysis

Analysis will include descriptive and inferential statistics. We present our detailed statistical plan for evaluation of the primary and secondary outcomes in Table 1. We will report baseline characteristics, that will be described using counts (percentage) for categorical variables and mean (standard deviation) or median (interquartile range) for continuous variables as appropriate. Our primary outcome will be defined by WHO criteria of \geq 3 liquid bowel movements/day, and additional definitions of diarrhea including a Bristol Stool Chart Score of 6 or 7, and Bliss Stool Classification System score of 4 will also be reported. The incidence of diarrhea will be computed as the number of new cases during ICU stay divided by the person-time at

risk. We will also report the frequency of diarrhea defined as the proportion (the number of patients who have developed diarrhea/the number of patients within the DICE-ICU cohort) of patients with new ICU-acquired diarrhea and will be computed for all 3 definitions.

The role of covariates (e.g., age, sex, APACHE II Score, drugs, prior gastrointestinal disease) in the occurrence of diarrhea will be analyzed with a logistic regression model that we developed during the DICE Pilot Study [17]. The consequences of diarrhea (time to ICU and time to hospital discharge) will be described and assessed using Cox regression; the influence of diarrhea on mortality will be assessed using logistic regression techniques. In these models, covariates will be entered as a block. Goodness-of-fit will be assessed by examining the residuals for model assumptions and using the Hosmer and Lemeshow goodness-of-fit test. Odds ratios (or hazard ratios), 95% confidence intervals and p-values will be reported.

For the validation of the Bristol Stool Chart and Bliss Stool Classification System we will use the WHO definition of diarrhea as the reference standard. We will compute agreement between the WHO definition of diarrhoea and the Bristol Stool Chart first; and then the Bliss Stool Classification System diarrhoea, using the Kappa statistic.

All analyses will be performed with SPSS software (version 22.0, 2013)[24].

Ethics and Dissemination

We have received local research ethics approval for the DICE-ICU Study in all participating centers. There are no safety concerns for enrolled patients. Information privacy will be addressed by de-identified data that is stored in password-protected computers in locked research offices at each center. There is a waiver of informed consent for this observational, non-interventional study in all centers except one that required written informed consent.

The results of this study will be disseminated by presenting the findings locally at each participating hospital, as well as nationally and internationally at critical care and gastroenterology conferences. Findings will be shared with interested national societies crafting guidelines in critical care. We will publish the results in a peer-review journal.

Discussion

On daily ICU rounds, diarrhea is discussed and addressed by a multidisciplinary team of clinicians - nurses, physicians, pharmacists and dietitians - and sometimes by concerned family members. Better understanding of the prevalence, characterization, risk factors and consequences of diarrhea will inform patient care for each of these professionals. Strategies initiated by each group could be implemented to prevent or treat diarrhea, in turn decreasing complications such as skin breakdown, electrolyte abnormalities, and nutritional deficiencies. For example, pharmacists may suggest changing medications; dietitians may modify feeding solutions; nurses may insert fecal management devices; families and bedside clinicians may increase the use of

protective materials and devices when entering the patient's room. Clinicians need to understand whether these interventions provide any benefit, cause any harm, and whether their cost is justified by the expected consequences associated with their use.

The burden of illness of diarrhea for patients appears to vary based on the definition used, highlighting the importance of making the definition explicit when citing incidence rates. The DICE-ICU Study will employ the 3 simplest measurement tools for diarrhea that are candidates for use in the busy ICU setting (the WHO definition, Bristol Stool Chart, and Bliss Stool Classification System). Clinicians perceive a high burden of illness and workload associated with diarrhea in the ICU [25]. Patients with diarrhea often have extensive work-ups to identify the underlying etiology of diarrhea. The European Society of Intensive Care Medicine (ESICM) Working Group on Abdominal Problems (2012) emphasizes that more research is required to identify the mechanism of diarrhea in critically ill patients, to identify different phenotypes of diarrhea, and thus, potential therapies[26].

A mechanism of interest for diarrhea includes alteration in the gut microbiota during critical illness. A study by lapichino and colleagues [27, 28] demonstrated in 15 critically ill patients who had not been exposed to antibiotics or steroids prior to ICU admission had a reduction in intestinal anaerobes with an increase in *Enterococcus* isolates. Interestingly, 12 of the 15 patients developed diarrhea and were also found to be negative for CDI [27]. A recent pilot prospective cohort study examined changes in fecal microbiota in 34 septic and non-septic critically ill patients in centers where systematic decontamination of the digestive tract (SDD) is used compared to 15 healthy controls [29]. The authors found low diversity of
species in the critically ill patient cohort compared to healthy controls including loss of *Faecalibacterium, Pseudobutryivibrio, Ruiminococcus, Subdoligranulum* [29]. There was also an increase of >75% of one genus in 4 of the 34 patients with *Enterococcus, Staphylcoccus, Escherichia* and *Shigella* in the critically ill patients which was not seen in the healthy controls cohort [29]. In this cohort there was no CDI infection; however, it is unclear if these patients had diarrhea during admission. These studies highlight the importance of future research into the microbiota during critical illness and how this may influence a patient's propensity to develop diarrhea.

While many reasons for diarrhea exist in the ICU, infectious etiologies are of particular concern. Although CDAD is a common concern, only a small percentage (11%) of patients with diarrhea are found to have CDAD [30]. The prevalence of CDAD in ICU patients is approximately 2% across a variety of ICUs, based on a recent systematic review of 16 retrospective and 6 prospective studies [30]. To date, there are only 4 prospective cohort studies focusing on ICU-acquired CDAD describing 92 patients [31-33] [34]. Patients with ICU-acquired CDAD appear to have an increased length of ICU and hospital stay compared to patients without CDAD [35]. However, this lack of high quality, observational data establishing the prevalence of ICU-acquired CDAD can lead to over-investigation, over-treatment, and over-attribution of diarrhea to this infection, potentially delaying the diagnosis of the true etiology of diarrhea. DICE-ICU will contribute to the growing knowledge of the prevalence of ICU-acquired CDAD. Data generated on the probability of various other etiologies of diarrhea in the DICE-ICU Study will offer probabilities associated with

each differential diagnosis, and may help to rationalize common, sometimes unnecessary resource-intensive investigations when seeking the root cause of diarrhea.

A more detailed understanding of diarrhea in the ICU will also help to refine approaches to care for patients with this problem. Future research may also illuminate whether diarrhea decreases physical contact with patients (e.g., pre-emptive isolation of patients associated with less frequent examination by physicians, and shorter visits by clinicians and families).

DICE-ICU has several strengths. It is a large prospective cohort study that encompasses both academic and community ICUs around the world. The study population will reflect a broad cohort of patients, enhancing the generalizability of the results. The sample size and enrolment of heterogenous patients will also allow for detailed examination of the incidence, risk factors and consequences of diarrhea, and will provide the first prospective study of incident cases of ICU acquired CDAD, examining the associated illness severity in this setting.

DICE-ICU has some potential limitations. Incomplete bedside documentation of bowel movements could introduce missing data; reporting bias may influence other data collection. Given the design, observer bias might influence some practices recorded as consequences of diarrhea. Although prospective cohort studies allow for identification of risk factors, there is the potential to identify spurious associations. Infants and children are excluded from this investigation.

The DICE-ICU Study will generate current, detailed multi-center clinical evidence on a common condition affecting many critically ill patients and influencing

different healthcare professionals in the ICU setting. As an international investigation, it will also be the largest prospective study to examine the frequency of, predisposing factors for, and consequences of diarrhea to inform critical care practice and future research.

Authors' contributions: The authors' roles are as follows

Conception and design: JC Dionne, D Cook, W Alhazzani, L Mbuagbaw

<u>Acquisition, analysis and interpretation of the data</u>: JC Dionne, D Cook, W Alhazzani, L Mbuagbaw, K Sullivan, J Devlin, M.Duprey, E. Duan, P. Moayyedi, D Armstrong, L Thabane, J Muscedere, J Tsang, R Jaeschke, A Takaoka, C Hamielec, T Karachi, R, Cartin-Ceba, M Alshahrani

Analysis of the data: JC Dionne, L Mbuagbaw, D Cook

<u>Drafting the manuscript</u>: JC Dionne, D Cook, L Mbuagbaw, L Thabane, E Duan, JW Devlin, MS Duprey, J Muscedere, J Tsang

<u>Critiquing the manuscript</u>: JC Dionne, D Cook, W Alhazzani, L Mbuagbaw, K Sullivan, J Devlin, M Duprey, E Duan, P Moayyedi, D Armstrong, L Thabane, J Muscedere J Tsang, R Jaeschke, A Takaoka, C Hamielec, T Karachi, R Cartin-Ceba, M Alshahrani

Final approval: All authors provided final approval of the manuscript

Funding statement: This work was supported by peer-review grants from the Hamilton Regional Medical Associates, Physicians Services Incorporated of Ontario, Canadian Association of Gastroenterology, and Hamilton Health Sciences Department of Medicine (Dr. J. Dionne) and the Canadian Institutes for Health Research (Dr. D Cook).

Competing interests' statement: The authors have no competing interests to declare.

Table 1: Statistical Analysis Plan

Research	Outcome	Covariates	Analytical approach
To determine incidence of	Primary Outcome	NA	The incidence of diarrhea will be computed as the
diarrhea and	WHO-defined diarrhea		number of new
consistency of	Different definitions of primary	-	stay divided by
bowel movements	outcome		the person-time at
during critical	Bristol Stool Chart-defined diarrhea		Descriptive
illness	Bliss Stool Classification System-		statistics
	defined diarrhea		patients with
To determine the			corresponding
frequency of			95% CI
diarrhea defined as			
the proportion of			
patients with new			
ICU-acquired			
diarrhea			
To determine risk	Primary Outcome	Age, Sex,	regression
factors associated	Dependent variable		- 3
with diarrhea	WHO-defined diarrhea	Score, drugs	
during critical		(motility	
illness		modifiers,	
		opiates, stool	
		softeners), prior	
		gastrointestinal	
		disease, center	
SECONDARY OUT	COMES Secondary Outcome	Covariates	A solution or solution
Research objective	Dependent variable	Covariates	Analytic approach
T. L. C. S. C. M. S.	Time to ICU discharge		
I o determine the	Time to hospital discharge	diarrhea. Age.	Cox regression
consequences of		Sex. APACHE II	
diarrhea		_Score	
	Mortality		Logistic
			regression
objective	Secondary Outcome	Covariates	Analytical approach
To determine the	IDSA-defined CDAD	NA	Descriptive
incidence of	ESCMID-defined CDAD		statistics
Clostridium	ACG-defined CDAD		(proportion of
difficile-associated			cases with
diarrhea (CDAD)			diarrhea during
during critical			siddy period)
illness			

To determine	Chance corrected agreement (Kappa	NA	Kappa statistic
agreement	score)		with 95% Cl
between WHO and Bristol and Bliss scores			

Abbreviations: Acute Physiology and Chronic Health Evaluation II (APACHE II); World Health Organization (WHO), *Clostridium-difficile* Associated Diarrhea (CDAD); Confidence Intervals (CI); European Society of Clinical Microbiology and Infectious Diseases (ESCMID); Infectious Disease Society of America (IDSA); American College of Gastroenterology (ACG); Not applicable(NA)

References:

- 1. Whelan, K., Judd, P.A., Taylor, M.A., Assessment of fecal output in patients receiving enteral tube feeding: a validation of a novel chart. European Journal of Clinical Nutrition, 2004. **58**: p. 1030-1037.
- 2. Ronan Thibault, S.G., Aurelie Cler, Nathalie Delieuvin, Claudia Paula Heidegger, Claude Pichard, *Diarrhoea in the ICU: respective contribution of feeding and antibiotics.* Critical Care, 2013. **17**(17): p. 1-8.
- 3. Schiller, L.R., et al., *Gastro 2013 APDW/WCOG Shanghai working party report: chronic diarrhea: definition, classification, diagnosis.* J Gastroenterol Hepatol, 2014. **29**(1): p. 6-25.
- 4. WHO. World Health Organization definition of diarrhea. 2016; Available from: Retrieved from URL <u>http://www.who.int/topics/diarrhoea/en/</u>
- 5. S.J. Lewis, K.W.H., *Stool form scale as a useful guide to intestinal transit time.* Scandinavian Journal of Gastroenterology, 1997. **32**: p. 920-924.
- 6. Mearin, F., et al., *Bowel Disorders*. Gastroenterology, 2016.
- 7. Caroff, D.A., et al., *The Bristol stool scale and its relationship to Clostridium difficile infection.* J Clin Microbiol, 2014. **52**(9): p. 3437-9.
- 8. Bliss, D., *Reliability of a stool consistency classification system.* Journal of WOCN, 2001. **28**(6): p. 305-313.
- 9. Bliss, D., Dhamani, K.A., Savik, K., Kirk, K., *Tool to classify stool consistency: Content validity and use by persons of diverse cultures.* Nursing and Health Sciences 2003. **5**: p. 115-121.
- 10. Jack, L., et al., *Diarrhoea risk factors in enterally tube fed critically ill patients: a retrospective audit.* Intensive Crit Care Nurs, 2010. **26**(6): p. 327-34.
- 11. Jack, L., et al., *Probiotics and diarrhoea management in enterally tube fed critically ill patients--what is the evidence?* Intensive Crit Care Nurs, 2010. **26**(6): p. 314-26.
- 12. McClave, S.A., et al., Guidelines for the Provision and Assessment of Nutrition Support Therapy in the Adult Critically III Patient: Society of Critical Care Medicine (SCCM) and American Society for Parenteral and Enteral Nutrition (A.S.P.E.N.). JPEN J Parenter Enteral Nutr, 2016. **40**(2): p. 159-211.
- 13. Reintam, A., et al., *Gastrointestinal failure in intensive care: a retrospective clinical study in three different intensive care units in Germany and Estonia.* BMC Gastroenterol, 2006. **6**: p. 19.
- 14. Macron, A.P., Antar Gamba, M., Carneiro Vianna, L.A., *Nosocomial diarrhea in the intensive care unit.* The Brazilian Journal of Infectious Diseases, 2006. **10**(6): p. 384-389.
- 15. Tirlapur, N., et al., *Diarrhoea in the critically ill is common, associated with poor outcome, and rarely due to Clostridium difficile.* Sci Rep, 2016. **6**: p. 24691.
- 16. Oczkowski, S.J.W., et al., *The Use of Bowel Protocols in Critically III Adult Patients: A Systematic Review and Meta-Analysis.* Crit Care Med, 2017. **45**(7): p. e718-e726.
- 17. Dionne, J.C., et al., *Tu1068 Diarrhea in the Intensive Care Unit: Epidemiology* (*The DICE Study*). Gastroenterology, 2016. **150**(4).

- 18. iDatafax idatafax Clinical Systems Incorporated, v., Seattle, WA: DF/NET Research INC, *iDatafax Clinical Systems Incorporated, version 4.3.0, Seattle, WA: DF/NET Research INC.* 2013.
- 19. McDonald, L.C., et al., *Clinical Practice Guidelines for Clostridium difficile* Infection in Adults and Children: 2017 Update by the Infectious Diseases Society of America (IDSA) and Society for Healthcare Epidemiology of America (SHEA). Clin Infect Dis, 2018. **66**(7): p. e1-e48.
- 20. Crobach, M.J., et al., European Society of Clinical Microbiology and Infectious Diseases: update of the diagnostic guidance document for Clostridium difficile infection. Clin Microbiol Infect, 2016. **22 Suppl 4**: p. S63-81.
- 21. Surawicz, C.M., et al., *Guidelines for diagnosis, treatment, and prevention of Clostridium difficile infections.* Am J Gastroenterol, 2013. **108**(4): p. 478-98; quiz 499.
- 22. Ogundimu, E.O., D.G. Altman, and G.S. Collins, Adequate sample size for developing prediction models is not simply related to events per variable. J Clin Epidemiol, 2016. **76**: p. 175-82.
- 23. Faul, F., Erdfelder, E., Lang, A.G., Buchner, A., *G*Power 3: A flexible statistical power analysis program for the social, behavioral, and biomedicalsciences.* Behavior Research Methods, 2007. **38**: p. 175-190.
- 24. Hosmer, D.W., Lemeshow, S., *pplied Logistic Regression. New York: Wiley IBM Corp. Released (2013).* 2013, IBM SPSS Statistics for Windows, Version 22.0: Armonk, NY: IBM Corp.
- 25. Heidegger, C.P., et al., *The burden of diarrhea in the intensive care unit (ICU-BD). A survey and observational study of the caregivers' opinions and workload.* Int J Nurs Stud, 2016. **59**: p. 163-8.
- 26. Reintam Blaser, A., et al., *Gastrointestinal function in intensive care patients: terminology, definitions and management. Recommendations of the ESICM Working Group on Abdominal Problems.* Intensive Care Med, 2012. **38**(3): p. 384-94.
- 27. Iapichino, G., et al., *Impact of antibiotics on the gut microbiota of critically ill patients.* J Med Microbiol, 2008. **57**(Pt 8): p. 1007-14.
- 28. Iapichino, G., J.M. Lankelma, and W. Joost Wiersinga, *Gut microbiota disruption in critically ill patients : Discussion on "Critically ill patients demonstrate large interpersonal variation of intestinal microbiota dysregulation: a pilot study".* Intensive Care Med, 2017. **43**(5): p. 718-719.
- 29. Lankelma, J.M., et al., *Critically ill patients demonstrate large interpersonal variation in intestinal microbiota dysregulation: a pilot study.* Intensive Care Med, 2017. **43**(1): p. 59-68.
- 30. Karanika, S., et al., *Prevalence and Clinical Outcomes of Clostridium difficile Infection in the Intensive Care Unit: A Systematic Review and Meta-Analysis.* Open Forum Infect Dis, 2016. **3**(1): p. ofv186.
- 31. Rotimi, V.O., Mokaddas, E.M., Jamal, W.Y., Verghese, T.L., el-Din, K., Junaid, T.A., *Hospital-acquired Clostridium difficile infection amongst ICU and burn patients in Kuwait.* Med Princ Pract, 2002. **11**(1).

- 32. Zahar, J.R., Schwebel, C., Adrie, C., Garouste-Orgeas, M., Francais, A., Vesin. A., Nguile-Makao, M., Laupland, K., Le-Monnier, A., Timsit, J.F., *Outcome of icu patients with clostridium difficile infection.* Critical Care, 2012. **16**.
- 33. Lumpkins, K., Bochicchio, G.V., Joshi, M., Gens, R., Bochicchio, K., Conway, A., Schaub, S., Scalea, T., *Clostridium difficile infection in critically injured trauma patients.* Surgical Infections, 2008. **9**(5): p. 497-501.
- 34. Nguyen, N.Q., et al., *Risk of Clostridium difficile diarrhoea in critically ill patients treated with erythromycin-based prokinetic therapy for feed intolerance.* Intensive Care Med, 2008. **34**(1): p. 169-73.
- 35. Dodek, P.M., et al., Length of stay and mortality due to Clostridium difficile infection acquired in the intensive care unit. Journal of Critical Care, 2013. **28**(4): p. 335-340.

	DICE Study055		Plate #010		Visit #000	
Pat II	ient 1		Patient Initials F L	Da Stud	te of y Day	(dd/mm/yyyy)
			BASELI	NE For	<u>m</u>	
		de	d/mm/yyyy			
1.	Study hospital admit date		201	6.	Height	cm inches
2.	Study ICU admit date			7.	Actual weight (ICU admission)	kg lbs
3.	Sex:	female	male			_
4.	Intubation date		201		N/A patient not intu	bated
5.	Date of birth					
8.	APACHE II Score (first 24 hours in st	tudy ICU):	Admission dia	gnosis c	:ode:	(if admitted from OR or PARR code should be 48-85)
	If "other" diagnosis	s code selected	,specify:			
9.	Location immedia	ately prior to <u>t</u>	<u>his</u> ICU admission (ch	eck ONE	box):	
	Emergency ro	om 🗌 ICU	(other hospital), adm da	ate:	Other ho:	spital admit date:
	Hospital ward	Eme	ergency (other hospital),	adm dat	te:dc	//mm/yyyy
	Operating room	m 🗌 Wan M 🗌 Nurs	d (other hospital), adm sing home, adm date:	date:		
	Other (specify):				
10.	Does the patient	have any of th	e following based on	chart rev	view only? (check A	LL that apply):
	Celiac disease	<u>.</u>			Gastroparesis	
	Irritable bowel	syndrome				atitis
		-,				 ium Difficile Infection
		Security Surgery	- E			
		im Dimcile integ	cuon Oseba diseesa autoasa "			
		vowel disease (cronn disease, uicerative	comis)	□ N	
		siomy				

	DICE Study055	Plate #020		Study Day
P	atient 1	Patient Initials	F L Date of Study Day FORM (page 1 of 4)	(dd/mm/yyyy)
1.	Advanced life support strategies	s received today		
	1. Invasive mechanical ventilation	n	🗌 No 📄 Yes, if ye	s specify: L ETT tracheostomy tube
	 Non-invasive mechanical ven (including CPAP/BiPAP for ar e.g., nocturnal) 	itilation ny duration,	🗌 No 📄 Yes, if ye	s specify: CPAP
	3. Inotropes or vasopressor infu	sions (check all)	🗌 No 🗌 Yes	
	🗌 dopamine 📃 n	orepinephrine	phenylephrine	milrinone
	🗌 dobutamine 🗌 e	pinephrine	vasopressin	midodrine
	4. Was dialysis performed today	/? 🗌 No	Yes, specify	
	intermittent (IHD) co	ntinuous (CRRT)	peritoneal other	(specify):
2.	Is this patient enrolled in the PR	OSPECT Pilot o	r RCT? No	Yes
3.	Laboratory results today (from A	AM Blood Work)	—	
eukocy	rtes	10 ⁹ /L Im ³ Sodium	(mmol/L or mEq/L)	Albumin g/L
lemogla	bbin g/L	Potassium	. (mmol/L or mEq/L)	Calcium mmo
latelets		Chloride	(mmol/L or mEq/L)	Magnesium . mmol/L
licarb	(mmol/L or mEg/L)	> Creatinine		//L Phosphate
4.	Did the patient receive any nutr	ition today?	No Yes, specify	
	Enteral Nutrition	rophic Feeds	Clear Fluids	
	Diet as tolerated	iefined as 10-20 mls/hr) "PN	Full Fluids	
5.	What activities were achieved t	oday (Check AL	L that apply)?	
	Chest physio 🗌 Up to ch	nair 🗌 Bike	Bed exercises A	nbulation 🗌 None
6.	Was a new antibiotic started to	day?	No Yes	
7.	Was there a change in antibioti	cs today?	No Yes	
8.	Were any of the following infec	tions suspected	today? 🗌 N/A, infectio	ns not specified.
	Respiratory Infection	No Y	es, was the culture positive	? 🗌 No 🔄 Yes, spec:
	Blood stream infection	No Y	es, was the culture positive	? 🗌 No 🗌 Yes, spec:
	Intra-abdominal infection		es, was the culture positive	? 🗌 No 🔲 Yes, spec:
	Urinary tract infection		es, was the culture positive	? 🗌 No 🗌 Yes, spec:
	C difficile associated diarrhea		es, was the culture positive	? 🗌 No 🔲 Yes, spec:
	Stool virology		es, was the culture positive	? No Yes spec
	Other infection (e.g., meningitis, sinusitis) Please specify:		es, was the culture positive	? No Yes, spec:
	Other infection (e.g., meningitis, sinusitis) Please specify:	□ No □ Y	es, was the culture positive	? 🗌 No 🗌 Yes, spec:

DICE Study055		Plate #021		ξ	Study Day
Patient 1		FL Patient Initials DAII Y DATA FOF	Date of Study Day	dd/mm	2 0 1
				L	
9. Did the patient reco	nve any of the	tollowing?	_	_	
1. H-2 receptor antagonist	If yes,	cimetidine (Tagame	et) famotidine	(Pepcid) other, s (Axid)	pecify:
	and specily.		n de la companya de la company		
N 2. Proton-pump	o Yes If yes, specify	Iansoprazole (Preva dexlansoprazole (D pantoprazole (e.g., P	acid) es exilant) on antoloc, Tecta) rad	omeprazole (Nexium) neprazole (Losec) beprazole (Pariet)	-
	and specify:		th		
N 3. Motility agent	o Yes Ifyes, s	pecify domperid (Motilium)	one me) Dose (M mg/24 hours	etoclopramide axeran) Dose	erythromycin
		other (so	ecify)	Dose	
	and spe		20	mg/24 ho	burs
4. Sorbitol/ N Hyperosmolar agents	o Yes] [] Ifyes,s	pecify Metformi	n Phenytoin	Magnesi	um
5. Laxative, N suppository or stool softener	o Yes ☐ ☐ Ifyes, _ specify	senna go duicolax gh	lytely lactulose	collace cit	ro-mag 🗌 peglyte
6. Enema	o Yes If yes, specify r and type	number received	Fleet	Soap suds	sses
7. Opiates	o Yes] [] Ifyes,pl	ease complete the Opi	ate Form		
8. Neuromuscular N blockers	o Yes ☐ ☐ Ifyes, ☐ specify	rocuronium [succinylcholine]	atracurium cisatracurium	mivacurium pancuronium	vecuronium 🗌
9. Probiotics	o Yes If yes, specify	Bio K	Other, specify:		
10. Chemotherapy N	o Yes	Cyclophosohamide	Capecitabine	Docetaxel	Paclitaxel
agents [(secreatory	specify				
diamhea)					
		Rinotecan		Oxalipatin	Other, specify:
11. Acetaminophen [o Yes ☐ ∐ lfyes, ☐ specify	Suspension	□РО	Both suspensio	n and PO

DICE Study05	55	Plate #022		Study Day
Patient 1		Patient Initials FL	Date of Study Day	(dd/mm/yyyy) 201
	DAILY	DATA FORM (page	<u>3 of 4)</u>	
10. Did the patient re	eceive any of the	following antibiotics	today (Please check ALL tha	t apply)?
1. Beta Lactams	No Yes If yes.	Penicillin G Penicillin V Ampicilin Ampicilin	Amoxicillin-Clavulanate Piperacillin/Tazobactum Oxacillin	Flucloxacillin Nafcillin Ticarcillin/Clavulanate
2. Cephalosporins	No Yes If yes.	Ampicilin-Sulbactam Amoxicilin Cefazolin Ceflexin (oral) Cefadroxil Cefuroxime Cefotiam Cefotiam Cefuroxime Cefepime	Dicloxacillin Cefaclor (oral) Cefepime Loracarbef Cefixime Cefoxitin Cefpodox Cefotaxime Cefdinir Ceftiaxone Proxetil (ceinacone Ceftibute)	Ceftobiprole Ceftaoline Ceftazidime/Avibactam Ceftolozane/Tazobactau oral) Not specified n (oral)
3. Carbapenems	No Yes If yes, specify	Imiperem Me	ropenem Ertapenem [Doripenem Not specified
4. Aminoglycosides	No Yes If yes,	Streptomycin Gentamicin Tobramycin	Netilmicin/Amikacin	
5. Quinolones	No Yes If yes.	Norfloxin Enoxacin Ofloxacin Ciprofloxacin	Levofloxacin Moxifloxacin Not specified	
6. Tetracyclines	No Yes If yes. specify	Tetracycline	Minocycline Not sp	ecified
7. Nitromidazoles	No Yes If yes, specify	Metronidazole (Flagy)	I) 🗭 🗌 IV 🗌 PO	
8. Macrolides	No Yes If yes.	Erythromycin Spiramycin Roxithromycin Clarythromycin	Azithromycin Not specified	
9. Lincosamides	No Yes If yes, specify	Clindamycin		

DICE Study05	55	Study Plate #023
Patient 1		(dd/mm/yyyy) Patient Date of 2 0 1 Initials E Study Day 2 0 1
	DAI	Y DATA FORM (page 4 of 4)
10. Did the patient re	ceive any o	the following antibiotics today (Please check ALL that apply)? CONTINUED
10. Azole Derivatives	No Yes If y	es, Miconazole Voriconazole Isavuconazonium crify Ketoconazole Posaconazole Not specified Fluconazole Amphotericin traconazole Clotrimazole
11. Echinocandins	No Yes Ify	es, Caspofungin Not specified cify Anidulafungin Micafungin
12. Glycopeptide	No Yes If y	es,Vancomycin →IVPOPR ×cifyDaptomycin
13. Monobactams	No Yes If sp	es, Aztreonam crífy
14. Antivirals 15. Other	No Yes No Yes I So Yes I So	Image: Sective

16. Last day of study daily data collection?

No
 Yes, patient died, was discharged to the ward, or study stopped at 70 days (submit Final Status Form)
 Yes, consent withdrawn for further data collection (submit a Final Status Form)

DICE Study055	■ ■ Plate #024	Study Day
Patient 1	Patient Date of Initials F L OPIATE & SEDATION FORM	(dd/mm/yyyy)

1. Did the patient receive any of the following Opiods?

	leive any of the following Opiot	15 :	Dose mg/24 hours
Morphine	Oral Bolus Infusion	Subcut. Other spec.	
Hydromorphone	Oral Bolus Infusion	Subcut. Cother spec.	
Percocet or Oxycodone	Oral Bolus Infusion	Subcut. Other spec.	
Propofol	Oral Bolus Infusion	Subcut. Cother spec.	
Midazolam	Oral Bolus Infusion	Subcut. Cther spec.	
Diazepam	Oral Bolus Infusion	Subcut. Conther spec.	
Lorazepam	Oral Bolus Infusion	Subcut. Conther spec.	
*Fentanyi	Oral Bolus Infusion	Subcut. Other spec.	
(or tylenol #1,2 or 3)	Oral Bolus Infusion	Subcut. Other spec.	
Demerol	Oral Bolus Infusion	Subcut. Spec.	_
Methadone	Oral Bolus Infusion	Subcut. Other spec.	_
*Dexemedetomidin	ne Oral Bolus Infusion	Subcut. Cother spec.	*mcg i 24 hou
Phenobarbitol	Oral Bolus Infusion	Subcut. Cother spec.	
Tramadol	Oral Bolus Infusion	Subcut. Cother spec.	_
Other (specify):	Oral Bolus Infusion	Subcut. Spec	_
Other (specify):	Oral Bolus Infusion	Subcut. Other spec.	_
Other (specify):	Oral Bolus Infusion	Subcut. Other spec.	_
Other (specify):	Oral Bolus Infusion	Subcut. Other spec.	
Other (specify):	Oral Bolus Infusion	Subcut. Spec.	

DICE Study055 Plate #030
(dd/mm/ywy)
Patient 1 Patient Date of Study Day 2 0 1
DIE TICIAN FORM
1. Please specify the diet the patient is receiving today? (check all that apply)
Diet as tolerated Unknown
2. What is the patients enteral nutrition target (target as determined by RD in total)?
3. What percentage of the patients nutritional target did they receive?
4. Were feeds interrupted? No Yes, specify
High residuals I lieus
Bleeding Unknown
5. Is the patient receiving TPN?
No Yes, specify Inadequate absorption Gastrointestinal fistula (short bowel syndrome)
Bowel obstruction
Other, specify:
Sepsis
and specify what formulation
Peripheral/central starter formula
and specify flow rate:
and specify Total 24 hour volume: L L L L L L
and specify lipid (intralipid)
6. Is the patient receiving free water? No Yes, specify how many mls in 24 hours

DICE Study055 Plate #031	Study Day
Patient 1 Patient Date of Initials F L DIETICIAN FORM	(dd/mm/yyyy) 201
7. Did the patient receive any enteral of oral nutrition today? No Yes, specific provides the provides of the patient receive any enteral of oral nutrition today? Image: performant of the	y: Ensure High Protein (1.0 kcal/mL) Ensure Plus Calories (1.5 kcal/ML) re) Oral (food) intake volume rot Oral (fluid) intake required
24h total ml of enteral nutrition delivered Image: Not applicable patient not receive enteral nutrition total enteral nutrition delivered ml/hr Image: Not applicable patient not receive enteral nutrition 8. What is the feeding tube insertion site today? (check ALL that apply) Image: Not applicable patient not receive enteral nutrition	ving us (specify:): 🗌 G tube
Postpyloric No feeding tube in situ	∐ GJ tube



STOOL CLASSIFICATION FORM





Diarrhea is defined as the passage of 3 or more liquid or loose stools per day

Hard and Formed	Soft but Formed	Loose & Unformed	Liquid
6 ⁰ 6 ())	త₿		000000
Having a hard or firm texture and retaining a definite shape like a banana, cigar or marbles	Retaining same general shape in the collection bag, does not spread all over the bottom of bag, or has a texture that appears like peanut butter	Lacking any shape of its own; spreads over the bottom of the collection bag; having a texture that appears like hot cereal	Like water

	DICE S	tudy055	■ ■ ■ Pi	ate #040		11	Study Day		
Pat I	iient D	1		Patient Initials	F L	Date of Study Day	(dd/mm/yyyy)		
	STOOL CLASSIFICATION FORM								
1.	Did the pa	itient have a	a bowel move	ment toda	ı y? [_	No 📋 Page not	Yes, please complete question 2-7		
2	Page not complete								
3.	Stool Clas	sification:					103		
	Bristol Bliss								
	Stool #1	Type 1-7	Score	smear sma l	mediur mediur	n 🗌			
	Stool #2			smear sma l	medium	n 🗌			
	Stool #3			smear sma l	medium Iarge	n 🗌			
	Stool #4			smear sma l	mediur	n 🗌			
	Stool #5			smear sma l	medium	n 🗌			
	Stool #6			smear sma l	medium	n 🗌			
	Stool #7			smear sma l	medium	n 🗌			
	Stool #8			smear sma l	medium Iarge	n 🗌 C 🗌 🗌 re S	Check if more than 8 stools to be ecorded for this study day (go to Additional Stool Classification Form)		
	Not applicable, too watery or continuous								
4.	. Were there any consequences of passing of stool today?								
	Feeds	held	Stool softe	ener held	Rectal ba	ig applied	Other, specify		
	Feeds	changed	Prokinetic	held	 Rectal tul	be inserted	—		
5.	Does the patient have any of the following in place?								
		al		-		y 🖵			
6.	Any other bowel hat	changes to its?	the patient's	care toda Yes, spec	y that you be	elieve cont	ne (mL) ributed to a change in the patient's		
7.	Did the pa loose stoo	tients bowe ols)?	el habits meet	the WHO Yes	Classificatio	on of Diarrh	ea today (3 or more liquid or		

DICE Stu	Idy055		Plate #041			Study Day
Patient	1		Patient [S S	Date of tudy Day	
	4	ADDITIO	NAL STOOL	<u> </u>	CATION	FORM
	Bristol	Bliss		Volumo		
	Type 1-/	Score	smear			
Stool #9			smal	large		
Stool #10			smear sma l	medium large		
Stool #11			smear	lame	H	
			3114			
Stool #12			smear		H	
			sma			
Stool #13			smear	medium		
			smal			
Stool #14			smear	medium		
5000 #14			smal	large		
04 - 1 <i>#45</i>			smear	medium		
Stool #15			smal	large		
			smear	medium		
Stool #16			smal	large		
Stool #17			smear	medium		
			smal			
Stool #18			smear			
			Siid			
Stool #19			smear sma l		H	
			smear	medium		
Stool #20			smal			
041 #04			smear	medium		
Stool #21			smal	large		
Stool #22			smear	medium		
VIVM TLL			smal	large		
Stool #23			smear	medium		
			smal	∐ large		Check if more stools are to be ecorded for this study day (go to Additional
Sta al #24			smear	medium		stool Classification Form)

DICE Study055	Plate #050			Study Day			
Patient 1	Patient Initials	F L	Date of Study Day	(dd/mm/yyyy)			
CLOSTRIDIUM DIFFICILE OUTCOME							

Please submit a copy of all positive or indeterminante culture reports AND supporting clinical documentation (i.e., physician notes, nursing notes, laboratory results, radiology reports, clinical notes relating to stool, colonoscopy reports and histology reports if available)

1. Clostridium difficile associated diarrhea? No Yes, specify:
≥ 3 episodes of unformed stools in ≤24 hours AND
Clostingium difficile toxin positive stool OR Colonscopic findings demonstrating pseudomembranous colitis
OR Histopathological findings of pseudomembranous colitis
2. Which test was this based upon? (Please check ALL that apply)
ELISA (enzyme-linked immunosorbent assay) Other, please specify:
PCR (polymerase chain reaction)
LAMP (loop-mediated isothermal amplification)
Cell Culture Cytotoxicity Assay
3. Clostridium Difficile Infection Severity (Clinical impression of Intensivist)
Mild Moderate Severe (e.g., toxic mega-colon)
4. Were there any consequences of the Clostridium difficile infection today?
Toxic megacolon Septic shock Other, specify
Bowel perforation Colectomy NONE

	DICE Study055 Plate #060	11	Study Day
Ρ	atient 1 Patient Initials F L ID FINAL STATUS	Date of Study Day (dd. Form	201 /mm/yyyy)
1.	Was the patient discharged from the ICU alive?	No Yes	
2.	Date of death or discharge from ICU (dd/mm/yyyy)		0 1
3.	Was the patient discharged from the hospital alive?	No Yes	Patient still in hospital at 1 year
4.	Date of death or discharge from hospital <u>or</u> if patient still hospitalized at 1 year, enter date 1 year from ICU discharge (dd/mm/yyyy)		1
5.	Was the patient transferred to another hospital?	☐ No ☐ Yes If yes, Term	was it to a Long 🗌 No Care facility? 📄 Yes

CHAPTER 3

Diarrhea During Critical Illness: A Multicenter Cohort Study (submitted to Intensive Care Medicine October 2021)

Diarrhea During Critical Illness: A Multicenter Cohort Study

Running Title: Diarrhea During Critical Illness

Joanna C. Dionne, MD, MSc, PhD(C)^{1,2}; Lawrence Mbuagbaw, MD, MPH, PhD^{2,3}; John W. Devlin, PharmD⁴; Matthew S. Duprey, PharmD PhD⁵; Rodrigo Cartin-Ceba, MD⁶; Jennifer Tsang, MD, PhD^{1,11}; Kristen Sullivan, MD¹; John Muscedere, MD⁷; Mohammed Alshahrani, MD⁸; Wojciech Szczeklik, MD ⁹; Paul Lysecki, MD¹⁰; Alyson Takaoka, MSc²; Brenda Reeve, MD¹²; Tracy Campbell, RD¹⁰; Karolina Borowska, MD⁹; Wojciech Serednicki⁹; Robert Cirone, MD¹³; Waleed Alhazzani, MD, MSc^{1,2}; Paul Moayyedi, MD^{1,2,14}; David Armstrong^{1,14}; Lehana Thabane, PhD^{2,3}; Roman Jaeschke, MD, MSc¹; Cindy Hamielec, MD¹; Tim Karachi, MD¹; Deborah J. Cook, MD, MSc^{1,2} for the DICE Investigators

1. Department of Medicine, McMaster University, Hamilton, Ontario, Canada

2. Department of Health Research Methods, Evidence, and Impact, McMaster

University, Hamilton, Ontario, Canada

3. Biostatistics Unit, St Joseph's Healthcare, Hamilton, Ontario, Canada

4. School of Pharmacy, Northeastern University, Boston, Massachusetts, USA

5. School of Public Health, Brown University, Providence, Rhode Island, USA

6. Department of Critical Care, Mayo Clinic, Phoenix, Arizona, USA

7. Department of Critical Care Medicine, Queen's University, Kingston, Ontario, Canada

8. Department of Critical Care, University of Dammam, Dammam, Saudi Arabia

9. Jagiellonian University Medical College, Krakow, Poland

10. Joseph Brant Hospital, Burlington, Ontario, Canada

11. Niagara Health System, St. Catharine's, Ontario, Canada

12. Brantford General Hospital, Brantford, Ontario, Canada

13. St. Joseph's Health Centre, Toronto, Ontario, Canada

14. Farncombe Family Digestive Health Research Institute, McMaster University,

Hamilton, Ontario, Canada

Corresponding Author: Dr. Joanna C. Dionne, Hamilton Health Sciences Juravinski Hospital and Cancer Centre, 711 Concession Street, A3-75, Hamilton, Ontario, Canada, L8V 1C3, email: <u>dionnejc@mcmaster.ca</u> ORCID https://orcid.org/0000-0002-9401-6868

Funding Sources: Peer-review grants from the Hamilton Regional Medical Associates,

Hamilton Health Sciences Department of Medicine, Physicians Services Incorporated of

Ontario, and the Canadian Association of Gastroenterology. Career Awards were from

Physicians Services Incorporated of Ontario (J Dionne) and the Canadian Institutes for

Health Research (D Cook).

Manuscript 2,998 Tables 5 Refences 32

Twitter: Diarrhea is common in ICU, rarely caused by C. difficile, has many

management consequences, but is not associated with mortality or increased length of stay.

Key words: Diarrhea, enteral nutrition, incidence, risk factors, critical illness

Take Home: In this study we demonstrated that diarrhea is common in the ICU, and rarely was *Clostridioides difficile* the cause. Diarrhea incidence varied based on the definition applied. Diarrhea was not independently associated with mortality or increased length of stay.

Abstract

Purpose: To study the incidence, risk factors, and outcomes of diarrhea during the ICU stay.

Methods: Prospective cohort of consecutive adults in the ICU for >24 hours during a 10-week period across 12 Intensive Care Units (ICUs) internationally.

Outcomes:1) incidence of diarrhea, 2) *Clostridioides difficile*-associated diarrhea (CDAD);3) ICU and hospital length-of-stay (LOS) and mortality in patients with diarrhea. We fit generalized linear models to evaluate the risk factors, management, morbidity and mortality associated with diarrhea.

Results: Among 1109 patients aged 61.4 (17.5) [mean (standard deviation)] years, 981(88.5%) were medical and 645 (58.2%) were mechanically ventilated. The incidence was 73.8% (818 patients, 73.8%, 95% confidence interval [CI] 71.1-76.6) using the WHO definition. Incidence varied across definitions (Bristol 53.5%, 95% CI 50.4-56.7; Bliss 37.7%,95% CI 34.9-40.4). Of 99 patients with diarrhea undergoing CDAD testing, 23 tested positive (2.1% incidence). Independent risk factors included enteral nutrition (RR 1.23, 95% CI 1.16-1.31, p<0.001), antibiotic days (RR 1.02, 95% CI 1.02-1.03, p<0.001), and suppositories (RR 1.14 95% CI 1.06-1.22, p<0.001). Opiates decreased diarrhea risk (RR 0.76, 95% CI 0.68-0.86, p<0.001). Diarrhea prompted management modifications (altered enteral nutrition or medications: RR 10.25, 95%CI 5.14-20.45, p<0.001) or other consequences (fecal management device or CDAD testing: RR 6.16, 95% CI 3.4-11.17, p<0.001). Diarrhea was not associated with increased ICU (RR

1.06, 95% CI 0.99-1.15, p=0.108) or hospital LOS (RR 1.06, 95%CI 0.94-1.19, p=0337), or hospital mortality (RR 0.78; 95% CI 0.58-1.03, p=0.081).

Conclusion: Diarrhea is common, has several risk factors, and prompts changes in patient care, but is not associated with increased morbidity or mortality.

Introduction

The reported incidence of diarrhea during critical illness ranges from 2%-95%[1, 2]. Differentiating diarrhea from changes in stool frequency, consistency, and volume that commonly occur during admission to the intensive care unit (ICU) is challenging[3]. Moreover, wide variation exists regarding what is considered a normal bowel habit in the general population[4], with definitions ranging from 3 bowel movements per week to 2-3 per day. Thus, there is no universal definition for what constitutes diarrhea in the ICU[5].

The World Health Organization (WHO) definition of diarrhea is the passage of 3 or more liquid stools per day[6], as adopted by the European Society of Intensive Care Medicine Abdominal Problem Working Group. Perhaps the most recognized stool evaluation instrument in hospitals is the Bristol Stool Chart [7], which is simple and easily applied at the bedside, comprised of descriptive text and a figure depicting each of the seven categories. The Bristol Stool Chart better predicts whole-intestinal transit time than stool frequency [7], and is used to define diarrhea associated with *Clostridioides difficile* by the European Society for Clinical Microbiology and Infectious Disease[8, 9]. The Bliss Stool Classification System has 4 categories with depictions and descriptions for each category. Despite reliability and validity when utilized by health care professionals[10, 11], this instrument is not widely used in research. Investigations in the ICU setting have employed the Bristol Stool Chart[12]; however, large studies validating these classification systems in critical illness are lacking.

Epidemiology of diarrhea in critically ill patients is limited in quality and

quantity. A recent systematic review identified 8 observational studies of diarrhea in this setting[5]. Studies have reported on diarrhea in enterally fed critically ill patients[13], diarrhea risk factors [14, 15] and manifestations of gastrointestinal failure (e.g., feeding intolerance and ileus)[16]. Designs included retrospective audits, registry analyses[14], case-control[16] and single-center studies[17]. Another recent systematic review included 12 prospective studies of diarrhea in the ICU [18]; from the final sample of 12,624 patients, the 1888 patients with diarrhea compared to those without had an associated increased ICU mortality (RR 1.43, 95% CI 1.03, 1.98), an increased length of stay in the ICU (MD 8.08 days, 95%CI 5.85,1032) and hospital (MD 9.67 days, 95%CI 2.17 to 17.16) [18].

The objectives of this study were to determine the incidence of diarrhea defined using the WHO criteria, including the incidence of *Clostridioides difficile* associated diarrhea (CDAD), to compare the incidence and definitions of diarrhea using the Bristol Stool Chart and Bliss Classification System, to identify diarrhea risk factors, and to describe the management modifications, consequences, and clinical outcomes associated with diarrhea.

Methods

Study Design and Population

The Diarrhea: Interventions, Consequences and Epidemiology in the Intensive Care Unit (DICE-ICU) Study is a prospective multicenter cohort study enrolling consecutive patients 18 years of age or older admitted to the ICU for \geq 24 hours. Patients were excluded if they were in ICU for < 24 hours; second and subsequent admissions were not considered to avoid non-independent observations. The design is reported elsewhere [19], including an internal pilot [20]. Participants were enrolled over a 10-week period in 12 academic and community medical-surgical ICUs in Canada (n=8), the United States (n=2), Poland (n=1), and Saudi Arabia (n=1). ICUs were enrolled serially, each determining its own 10-week study period from July 2014-August (internal pilot 2014-2015, main cohort 2016-2019). Patients were followed daily in the ICU until discharge, then hospital vital status and length of stay was documented, censored at 1 year. DICE ICU was approved by the research ethics board at each center with a waiver of informed consent except for 1 center which mandated a priori written consent. DICE is reported per STROBE guidelines [21].

Outcomes

The research team trained bedside nurses [19] to track the number and character of each stool daily. The reference standard and primary outcome was the WHO definition of at least 3 liquid bowel movements per day[6]; we also used the Bristol Stool Chart Score of 6 or 7 [7] and Bliss Stool Classification System score of 4 [10] as secondary diarrhea definitions.

Research staff collected baseline patient characteristics (i.e., age, sex, prehospital comorbidities, Acute Physiology and Chronic Health Evaluation II (APACHE II) score [22], admission diagnosis and pre-existing gastrointestinal conditions (i.e., inflammatory bowel disease, Celiac disease, short bowel syndrome, prior bowel resection, chronic pancreatitis, and gastroparesis, CDAD, ileostomy or colostomy). Research staff collected daily life support (i.e., invasive mechanical ventilation, vasopressors, renal replacement therapy), laboratory values, enteral nutrition (i.e., formulation, route, volume, and interruptions), medications known to influence the risk of diarrhea, and management modifications and consequence of diarrhea. CDAD testing was performed at the physician's discretion. Mortality and length of ICU and hospital stay were documented, censored at 1 year.

Data were validated by research staff and the principal investigator (JCD) at McMaster University's Methods Center.

Funding

This work was supported by peer-review grants from Hamilton Regional Medical Associates, McMaster University Department of Medicine, Physicians Services Incorporated of Ontario, and the Canadian Association of Gastroenterology which had no role in the design, analysis or interpretation of data or manuscript.

Statistical Analysis

The methodology and statistical analysis plan were published [19]. Briefly, our sample size was derived by two approaches: 1) the rule of thumb based on independent risk factors and number of events per degree of freedom and 2) the DICE internal pilot primary objectives. Baseline characteristics were analysed descriptively, reported as counts (percent), mean (standard deviation) or median (quartile 1, quartile 3) as appropriate. The incidence of diarrhea was the proportion of participants who developed diarrhea on day 1 or later in the ICU (WHO as the primary definition), and the Bristol and Bliss scores (secondary definitions). We also calculated the incidence rate (number of new cases of diarrhea divided by person-time at risk in the ICU). The prevalence of CDAD was calculated as the proportion of patients with CDAD upon ICU admission and the number of cases acquired in ICU. The incidence of CDAD was calculated as the proportion of patients with diarrhea testing positive subsequently during their ICU admission.

Using the kappa statistic, we calculated agreement among the WHO, Bliss and Bristol definitions.

For all adjusted analyses, we used generalized linear models. We specified a log link, a normal distribution (to estimated adjusted risk ratios) with clustered robust standard errors to account for potential clustering within centers. Goodness of fit was assessed using Akaike information criteria (AIC). Potential diarrhea risk factors (per the WHO definition[6])were determined based on previous studies identifying antibiotics, antifungals, suppository, prokinetics, CDAD, and enteral nutrition[2, 13, 14], further refined during the DICE pilot study[20]. The following were continuous and binary covariates: baseline factors (age, sex, APACHE II score center), and exposures in the ICU (enteral nutrition, and medications [opiates, motility agents, sorbitol, acid suppressants, total number of antibiotics and the number of days on antibiotics, and chemotherapy], as a block. We also analyzed differences in diarrhea risk factors for the Bristol Stool Chart and Bliss Stool Classification (Appendix Table 1A and 1B).

We fitted similar models to examine the management modifications and consequences of diarrhea adjusting for age, sex, and APACHE II score. Management modifications were any of: altered enteral nutrition (i.e., feeds held or decreased, formula changed), stool softener or prokinetic held, or anti-diarrheal agent administered. Management consequences were either fecal management device insertion or CDAD testing.

We also determined the association between diarrhea and ICU and hospital length of stay, and mortality, adjusting for age, sex, and APACHE II score. Adjusted risk ratios (RR), 95% confidence intervals (CI), p-values and AIC are reported. This deviates from the previously published statistical plan[19]. Odds ratios (OR) were the initial estimate of effect planned to be reported in DICE-ICU, however, following feedback from stakeholders, an adjusted risk ratio was implemented as it is more conservative estimate, and intuitive to clinicians at the bedside. Further, we accounted for potential clustering within centers by using clustered robust standard errors. Imputation methods were determined a priori in the case of significant missing data [19]. All analyses were performed using Stata (V. 16, 2019)[23].

Results

From June 2014-August 2019, 1114 patients were enrolled at 12 academic and community ICUs in Canada, the United States, Poland and Saudi Arabia (Figure 1), 1109 of whom were included in this study. The mean (standard deviation) age was 61.4 (17.5) years, APACHE II score was 18.8 (8.0), and 591 (53.2%) were mechanically ventilated at baseline (Table 1). Most patients were

medical (981, 88.5%). Diarrhea-related comorbidities at ICU admission included colectomy or ileostomy (2.4%), and inflammatory bowel disease (0.1%). Minimal data were missing; thus, imputation was not required. For main outcomes, patients with complete data were used.

Incidence of Diarrhea

Based on the WHO definition, 818 of 1109 patients developed diarrhea, for an incidence of 73.8% (95% CI 71.1-76.6); the median (quartile 1- quartile 3) time to diarrhea onset was 2 (1-4) days, for an incidence rate of 224.6/1000 person-days (95% CI 209.5-240.6). The incidence of diarrhea was 53.5% (95% CI 50.4-56.7) using the Bristol Stool Chart and 37.7% (95% CI 34.9-40.4) using the Bliss Stool Classification System. The incidence did not differ across centers (Appendix Table 4).

The prevalence of CDAD (Appendix Figure 1) in the ICU, including pre-ICU CDAD and ICU-acquired CDAD, was 85/1109 (7.7%). However, among 99 patients tested for CDAD, only 23 were positive (CDAD incidence in the ICU of 2.1%).

Comparison of the definitions of diarrhea: WHO, Bristol and Bliss

Compared with the WHO definition of diarrhea, agreement with a Bristol Stool Chart score of 6 or 7 was moderate (Kappa = 0.51, 95%Cl 0.46-0.55, p<0.001) and with a Bliss score of 4 was fair (Kappa = 0.31, 95%Cl 0.27-0.35, p<0.001). The pooled agreement across 3 definitions was fair (Kappa = 0.39, 95% Cl 0.36-0.42), p< 0.001). The WHO definition of diarrhea identifies more patients with diarrhea and is the definition used for this study.

Risk Factors for Diarrhea

Independent diarrhea risk factors (WHO definition) included enteral nutrition (RR 1.23, 95% CI 1.16-1.31, p<0.001), number of antibiotic days (RR 1.02, 95% CI 1.02-1.03, p<0.001) and suppository use (RR 1.14 95% CI 1.06-1.22, p<0.001) (Table 2). Opiates (RR 0.76, 95% CI 0.68-0.86, p<0.001) were associated with a decreased risk.

Risk factors for diarrhea using the Bristol Stool Chart definition were similar to the WHO definition; however, two additional risk factors were age (RR 1.00, 95%Cl 1.00-1.01, p=0.034) and total number of antibiotics (RR 1.05, 95%Cl 1.01-1.10, p=0.019). Considering the Bliss Stool Classification, diarrhea risk factors were similar to the WHO and Bristol definitions, with the addition of female sex (RR 1.11, 95%Cl 1.01-1.22, p=0.030) and acid suppressants (RR 1.66, 95%Cl 1.15-2.40, p=0.007) (Appendix Table 1A and 2B).

A post hoc analysis of enteral nutrition composition on the impact of diarrhea in this cohort, after adjustment for antibiotics and suppositories, demonstrated that high osmolarity EN (RR 1.14, 95%CI 1.08-1.20, p<0.001) and high fiber enteral nutrition (RR 1.11, 95%CI 1.11-1.17, p<0.001) were feeding compositional features associated with diarrhea (Appendix Table 2).

Management Modifications and Consequences of Diarrhea

The most frequent management modification prompted by diarrhea was holding a stool softener, and most frequent management consequence was ordering a CDI test (Table 3). After adjusting for age, sex, center and APACHE II score, diarrhea was associated with at least at least one management modification: discontinuing stool softener or prokinetic (RR 10.25, 95%CI 5.14-20.45, p<0.001) and fecal management devices (rectal bag applied or rectal tube inserted) or *C. difficile* testing (RR 6.16, 95% CI 3.4-11.17, p<0.001).

Clinical Consequences of Diarrhea

Patients with diarrhea (WHO definition) stayed in the ICU a median of 6.5 days (IQR 4.0,12.0) in contrast to those without diarrhea who stayed 3.0 days (IQR 2.0,4.0). Patients with diarrhea stayed in hospital 15 days (IQR 8.0,31.0) compared to those without who stayed 7.0 days (IQR 3.0,14.0), p<0.001). However, adjusting for age, sex, APACHE II score, gastrointestinal diagnoses, diabetes and center, patients with diarrhea had a similar ICU LOS (RR 1.06, 95% CI 0.99-1.15, p=0.108) and hospital LOS (RR 1.06, 95% CI 0.94-1.19, p=0.337) to others (Appendix Table 3A). Diarrhea was not independently associated with hospital mortality (RR 0.78; 95% CI 0.58-1.03, p=0.081). The association of diarrhea with duration of ICU and hospital stay and hospital mortality was similar for the Bristol and Bliss definitions (Appendix Table 3B and 3C).

Discussion

In this international multicenter prospective cohort study of 1109 critically ill patients, diarrhea was common, and the incidence varied based on the definition. Independent modifiable risk factors for diarrhea included enteral nutrition, suppository use, and number of antibiotic days, while opioid use was associated with a lower risk of diarrhea. These risk factors were consistent across definitions.
Adjusted analyses found that diarrhea was not associated with hospital mortality or longer stays in the ICU and hospital.

Variation in bowel habit definitions[4], and inattention to usual bowel habits before critical illness make it challenging to identify what may be abnormal for a critically ill patient. A systematic review of constipation, diarrhea and the use of bowel protocols in the ICU identified 8 cohort studies examining diarrhea [5]. Most studies were retrospective or single-center, and used the definition applied in this study of 3 or more liquid bowel movements per day. DICE-ICU is the largest prospective multicenter study conducted with the primary focus on diarrhea in this setting; it also serves as an initial study examining the differences in stool assessment metrics and definitions. A previous single-center prospective study of 1300 critically ill patients examining a wide range of conditions contributing to gastrointestinal dysfunction (i.e. vomiting, diarrhea, bowel dilation, and gastric residuals)[17], documented only 14% as experiencing diarrhea [17], and found that having more than 2 gastrointestinal symptoms was associated with increased mortality and a longer length of ICU stay[17]. Our study focused on diarrhea specifically, rather than gastrointestinal dysfunction more generally, using 3 definitions, and analyzed risk factors and outcomes in the ICU setting.

We documented fair agreement across all diarrhea definitions applied. The WHO and the Bristol Stool Chart demonstrated moderate agreement. While the WHO definition was associated with the highest incidence of diarrhea, analyses yielded several consistent risk factors across diarrhea definitions. The attributable morbidity and mortality of diarrhea across definitions was similar. Ensuring consistent nomenclature

in practice will improve interprofessional recognition of diarrhea at the bedside, and help to advance research in this field, including the testing effective interventions to prevent and treat diarrhea.

Our findings quantify and highlight the importance of antibiotic appropriateness and minimizing the number of antibiotic days for patients in the ICU. We showed that every additional day of antibiotic exposure is associated with a 10% increased risk of diarrhea per day, after adjusting for multiple antibiotics. Antibiotic stewardship programs may help to tailor antibiotic therapy and prevent indiscriminate prescribing; whether this reduces the burden of diarrhea remains to be evaluated [24, 25]. Our results are consistent with a prior study suggesting that antibiotics, suppositories, and enteral nutrition predispose critically ill patients to diarrhea [2]; our study helps to quantify the associated the risk.

While the enteral route is the preferred method of nutrition delivery in the ICU[26] , it is often considered a cause of diarrhea, prompting discontinuation[13], which in turn may interrupt nutritional support. Preliminary data have shown an association between high protein feeds and diarrhea compared to other types of enteral nutrition[27]. In our study, enteral nutrition was associated with the development of diarrhea. Post-hoc analysis of nutritional composition suggested that high osmolality feeds or high fiber feeds was associated with diarrhea, rather than general exposure to enteral nutrition. Further research is needed on the association between diarrhea, different feeding formulae and feeding schedules (e.g., continuous or intermittent bolus).

This study documented several interventions that are initiated in response to diarrhea including altering or holding enteral nutrition, changing medications, investigating an infectious etiology, and rectal appliance management. These interventions have implications for patients and the health care system. If feeds are held frequently, this may exacerbate caloric and protein deficits. Frequent CDAD testing, although congruent with recent guidelines suggesting heightened awareness of this infection[28], incur laboratory and other costs related to contact isolation precautions for patients and clinicians until results are available.

We found that patients who experienced diarrhea had a similar length of stay in the ICU and in hospital compared to other patients. Patients with diarrhea did not have an increased risk of death. Previous studies have yielded conflicting results regarding the association of diarrhea with increased mortality. In a recent systematic review of prospective studies of diarrhea in the ICU, an association between mortality and ICU and hospital length of stay was found; however, included studies had relatively small sample sizes, were at moderate risk of bias and the overall certainty of evidence was low[18]. Reasons for worse outcomes in patients with diarrhea seen in some studies may reflect changes in gut perfusion or altered gut microbiota during critical illness [29]. Translational research has shown reduced microbiome diversity in respiratory and gastrointestinal samples correlates with higher disease severity and adverse outcomes [30-33].

Limitations of our study include lack of mechanistic data to help explain the relationship between diarrhea and clinical outcomes. We cannot exclude the possibility of observer bias influencing patient management in response to diarrhea,

or unmeasured confounders affecting analyses. Although our incidence of CDAD of 2.1% in this cohort is consistent with other ICU studies, CDAD testing was at the discretion of the ICU physician, which may lead to an underestimate of the incidence. We did not classify the appropriateness of antibiotics or analyze broad spectrum antibiotics in this study. Strengths of this study include the large sample size and heterogenous population allowing for detailed examination of risk factors and outcomes. Our internal pilot study refined the study methods and calculation of the sample size for multivariable regression. We published our methods and analysis plan in a peer review journal[19] enhancing the transparency of this report. We enrolled consecutive, critically ill patients in both academic and community ICUs with international representation, enhancing the generalizability of the findings. Based on additional stakeholder input, we have presented our results as risk ratios instead of odds ratios (per protocol) to facilitate interpretability.

Our study may serve as a foundation for further work in refining a definition for diarrhea that is easily applied at the bedside. A universal validated definition of diarrhea in this population could be useful for interprofessional practice, to inform translational and clinical research on enteric infectious diseases, malabsorption, and gastrointestinal dysfunction. Future investigations should examine whether addressing modifiable risk factors may prevent diarrhea and impact favorably on patient-important outcomes. Additional studies on gastrointestinal dysbiosis in critical illness may yield information on propensity to develop diarrhea and its attributable morbidity and mortality. Economic analyses would quantify the resources associated with diarrhea, which lead to bedside interventions by nurses, dieticians and

pharmacists, diagnostic tests, and increased use of consumables such as gowns and other personal protective equipment.

In conclusion, diarrhea is common among critically ill patients, and the incidence varies based on the definition employed. Modifiable diarrhea risk factors include enteral nutrition and duration of antibiotic exposure. Further studies are needed to evaluate whether modifying these factors reduces the incidence of diarrhea, and to determine the impact on healthcare costs.

Acknowledgments:

We appreciate the data management assistance of Suzanne Duchesne and Nicole Zytaruk for this study. We also thank the research teams at each participating institution.

Declarations:

Authors' contributions: The authors' roles are as follows

Conception and design: JC Dionne, DJ Cook, W Alhazzani, L Mbuagbaw

Acquisition, analysis and interpretation of the data: JC Dionne, DJ Cook, L Mbuagbaw, W Alhazzani, K Sullivan, JW Devlin, M Duprey, P Moayyedi, D Armstrong, L Thabane, J Muscedere, J Tsang, R Cirone, R Jaeschke, A Takaoka, C Hamielec, T Karachi, R Cartin-Ceba, M Alshahrani, B Reeve, W Szczeklik, K Borowska, W Serednicki, P Lysecki, T Campbell

Analysis of the data: JC Dionne, L Mbuagbaw, D Cook

<u>Drafting the manuscript</u>: JC Dionne, D Cook, L Mbuagbaw, L Thabane, JW Devlin, MS Duprey, J Muscedere, J Tsang, A Takaoka

<u>Critiquing the manuscript</u>: JW Devlin, K Sullivan, M Duprey, P Moayyedi, D Armstrong, L Thabane, J Muscedere, J Tsang, R Jaeschke, A Takaoka, C Hamielec, T Karachi, R Cartin-Ceba, M Alshahrani, W Alhazzani, B Reeve, W Szczeklik, K Borowska, W Serednicki, P Lysecki, T Campbell Final approval: All authors provided final approval of the manuscript

Funding statement: This work was supported by peer-review grants from the Hamilton Regional Medical Associates, Hamilton Health Sciences Department of Medicine, Physicians Services Incorporated of Ontario, and the Canadian Association of Gastroenterology. Career Awards were from Physicians Services Incorporated of Ontario (J Dionne) and the Canadian Institutes for Health Research (D Cook).

Competing interests' statement: The authors have no competing interests to declare.

Data availability: Upon request

Code availability: Not applicable

Participating Centers:

Niagara Health System: J Tsang (Lead), Dimitra (Gina) Fleming, Susan O'Farrell, Brittany Young, Allison Brown, Helen Su, Robin Owen, Kathryn Lalonde, Kathleen Willis Joseph Brant Hospital: P Lysecki (Lead), T Campbell St. Joseph's Healthcare Hamilton: D Cook (Lead), K Sullivan, A Takaoka St. Joseph's Hospital Toronto: R Cirone (Lead), K Kavikondala Hamilton Health Sciences (General Site): C Hamielec (Lead), K Sullivan Hamilton Health Sciences (Juravinski Site): T Karachi (Lead), K Sullivan Northeastern University, Boston: J Devlin (Lead), M Duprey Mayo Clinic, Phoenix: R Cartin-Ceba (Lead), H Raza Kingston General Hospital: J Muscedere (Lead), M Hunt, I Georgescu University of Dammam, A Alshahrani (Lead), LP Asonto Brantford General Hospital, B Reeve (Lead), W Dechert Jagellonian University Medical School, W Szczeklik (Lead), K Borowska

Table 1: Baseline Characteristics

Characteristics	Total Cohort	Diarrhea	No Diarrhea
	(n=1109)	(n=818)	(n=291)
Sex: Female: n (%)	489 (44.1)	372 (45.5)	117 (40.2)
Age: Mean (SD)	61.4 (17.5)	61.8 (17.2)	60.3 (18.3)
APACHE II Score: Mean (SD) ^{8 missing}	18.8 (7.98)	19.1 (7.8)	17.8 (8.5)
Type of Patient, n (%) ^{7 missing}			
Medical	981 (88.5)	731 (89.9)	250 (86.5)
Surgical	59 (5.3)	38 (4.7)	21 (7.3)
Trauma	62 (5.6)	44 (5.4)	18 (6.2)
Admitting Diagnosis, n (%) ^{7 missing}	· · ·		
Cardiovascular	140 (12.6)	95 (11.7)	45 (15.6)
Respiratory	272 (24.5)	214 (26.3)	58 (20.1)
Gastrointestinal	161 (14.5)	119 (14.6)	42 (14.5)
Neurologic	137 (12.4)	98 (12.1)	39 (13.5)
Sepsis	117 (10.6)	95 (11.7)	22 (7.6)
Trauma	62 (5.6)	44 (5.4)	18 (6.2)
Metabolic	86 (7.8)	55 (6.8)	31 (10.7)
Hematologic	10 (0.9)	9 (1.1)	1 (0.3)
Renal	30 (2.7)	24 (3)	6 (2.1)
Gynecologic	2 (0.2)	2 (0.2)	0 (0)
Orthopaedic Surgery	14 (1.3)	8 (1)	6 (2.1)
Cardiovascular Surgery	6 (0.5)	2 (0.2)	4 (1.4)
Other Medical	28 (2.5)	22 (2.7)	6 (2.1)
Other Surgical	37 (3.3)	26 (3.2)	11 (3.8)
Location Prior to ICU: ^{9 missing}			
Emergency room	451 (40.7)	327 (40)	124 (42.6)
Hospital Ward	266 (24)	221 (27)	45 (15.5)
OR/Recovery Room	221 (19.9)	129 (15.8)	92 (31.6)
Other	46 (4.1)	42 (5.1)	4 (1.4)
ICU (Other hospital)	45 (4.1)	38 (4.6)	7 (2.4)
Emergency (other hospital)	58 (5.2)	42 (5.1)	16 (5.5)
Ward (Other hospital)	21 (1.9)	18 (2.2)	3 (1)
Relevant Comorbid Conditions:			
Celiac disease ^{1 missing}	2 (0.1)	1 (0.1)	1 (0.3)
Irritable bowel	6 (0.5)	5 (0.6)	1 (0.3)
Diabetes	318 (28.7)	229 (28)	89 (30.6)
Prior bowel resection surgery	30 (2.7)	25 (3.1)	5 (1.7)
Inflammatory bowel disease (Crohn	24 (2.2)	22 (2.7)	2 (0.7)
disease, ulcerative colitis)			
Colectomy/Ileostomy	27 (2.4)	21 (2.6)	6 (2.1)
Chronic Pancreatitis	10 (0.9)	8 (1)	2 (0.7)
Current Clostridium Difficile Infection	11 (.1)	9 (1.1)	2 (0.7)

Gastroparesis	7 (0.6)	6 (0.7)	1 (0.3)
Study Day 1			
Invasive mechanical ventilation ^{4 missing}	591 (53.2)	455 (55.9)	136 (46.7)
Inotropes or vasopressors 6 missing	405 (36.5)	304 (37.3)	101 (34.9)
Dialysis/renal replacement ^{6 missing}	79 (7.1)	59 (7.3)	20 (6.9)

Legend for Table 1: In this table we present baseline characteristics of 1,109 critically ill patients. SD=standard deviation. APACHE=Acute Physiology and Chronic Health Evaluation. Surgical patients were defined according to Canadian Critical Care Trials group definition.

	WHO Incidence: 73.8% (95% CI 71.1-76.6)			
Model	Multivariable model (full)		Multivariable model (reduced)	
Covariates	Adjusted RR (95%Cl)	P - value	Adjusted RR (95%CI)	P -value
Sex				
	0.94 (0.89 –1.00)	0.066		
Age				
	1.00 (1.00 –1.00)	0.101		
APACHE II				
score	1.00 (0.99 –1.00)	0.173		
Opiates			0.76 (0.68-0.86)	<0.001
	0.76 (0.67 - 0.86)	<0.001		
Chemothera				
ру	1.05 (0.91 - 1.20)	0.509		
Antibiotics			1.03 (1.00-1.06)	0.097
(total #)	1.03 (1.00 - 1.05)	0.030		
Antibiotic			1.02 (1.02-1.03)	<0.001
days	1.02 (1.01 - 1.03)	<0.001		
Motility				
Agent	1.04 (0.98 - 1.10)	0.200		
Sorbitol				
	1.06 (0.96 - 1.17)	0.225		
Suppository		<0.001	1.14 (1.06-1.22)	<0.001
	1.13 (1.06 - 1.19)			
Enteral		<0.001	1.23 (1.16-1.31)	<0.001
Nutrition	1.23 (1.16 - 1.31)			
Acid				
suppressant	1 08 (0 94 - 1 23)	0.201		
Gastrointesti	1.00 (0.94 - 1.23)	0.234		
nal				
comorbiditie				
S**	0.98 (0.93 - 1.04)	0.507		
AIC	0.982		0.982	

Legend for Table 2: In this table we present independent risk factors for diarrhea (WHO definition) using a generalized linear model, adjusting for age, sex, APACHE II Score, opiates, chemotherapy, number of antibiotics, antibiotic days, motility agent,

sorbitol, suppository, enteral nutrition, acid suppressants, gastrointestinal comorbidities and center. Total number of antibiotics reflects the number of unique antibiotics that a patient received. RR= Risk Ratio.CI=confidence interval. APACHE=Acute Physiology and Chronic Health Evaluation. Celiac disease, Prior bowel resection surgery, Inflammatory bowel disease (Crohn disease, ulcerative colitis), Colectomy/Ileostomy, Chronic Pancreatitis, Gastroparesis, Diabetes. AIC= Akaike Information Criterion.

Table 3: Management	t modifications and	consequences	of diarrhea
---------------------	---------------------	--------------	-------------

Management modification and consequences	Patients with diarrhea** (n=818)	No diarrhea (n=291)	RR (95%CI)	P- value
Any management modification, n			10.25 (5.14-	<0.001
(%)	166 (20.29)	5 (1.7)	20.45)	
Stool Softener held, n (%)	118 (14.4)	4 (1.4)		
Feeds held, n (%)	52 (6.4)	0 (0)		
Feeds changed, n (%)	42 (5.1)	0 (0)		
Prokinetic held, n (%)	21 (2.6)	1 (0.3)		
Any management consequence, n			6.16 (3.4-11.17)	<0.001
(%)	171 (20.9)	10 (3.4)		
Clostridioides difficile associated				
diarrhea test, n (%)	94 (11.5)	6 (2.1)		
Other consequence, n (%)	63 (7.7)	4 (1.4)		
Rectal tube inserted, n (%)	37 (4.5)	4 (1.4)		
Rectal bag applied, n (%)	17 (2.1)	2 (0.7)		

Legend for Table 3: In this table we present the management modifications and consequences of diarrhea (WHO definition) on individual management consequences. RR=Risk Ratio. *Adjusted for age, sex, APACHE II score, center.

Variable	Death		ICU stay		Hospital stay	
	RR (95% CI)	Р	RR (95% CI)	Р	RR (95% CI)	Р
Diarrhaa	0.78 (0.58 -		1.06 (0.99 -		1.06 (0.94 -	
Diarriea	1.03)	0.081	1.15)	0.108	1.19)	0.337
Sov	1.01 (0.82 -		1.00(0.96 -		0.99 (0.93 -	
Sex	1.24)	0.932	1.05)	0.854	1.05)	0.755
Ago	1.02 (1.01 -		1.00 (1.00 –		1 (0.99 –	
Age	1.04)	0.002	1.00)	0.231	1.00)	0.004
APACHE	1.04 (1.02 -		0.99 (0.98 -		0.98 (0.98 -	
II score	1.06)	<0.001	0.99)	<0.001	0.99)	<0.001

Table 4:	ICU and Hos	pital Length of	f Stay and	Mortality
----------	-------------	-----------------	------------	-----------

Legend for Table 4: In this table we present the association of diarrhea (WHO) on any management consequences, ICU and hospital length of stay, and mortality. The models were adjusted for age, sex, APACHE II score, center. RR=Risk Ratios, adjusted. CI=confidence interval. APACHE=Acute Physiology and Chronic Health Evaluation.

Figure 1: Flow Diagram for DICE-ICU Study



Legend for Figure 1: Flow diagram of patients enrolled in the DICE-ICU Study.

References

- Whelan K, Judd PA, Taylor MA, (2004) Assessment of fecal output in patients receiving enteral tube feeding: validation of a novel chart. Eur J Clin Nutr 58: 1030-1037
- 2. Thibault R, Cler A, Delieuvin N, Heidegger, CP, Pichard C, (2013) Diarrhoea in the ICU: respective contribution of feeding and antibiotics. Critical Care 17: 1-8
- 3. Reintam Blaser A, Malbrain ML, Starkopf J, Fruhwald S, Jakob SM, De Waele J, Braun JP, Poeze M, Spies C, (2012) Gastrointestinal function in intensive care patients: terminology, definitions and management. Recommendations of the ESICM Working Group on Abdominal Problems. Intensive Care Med 38: 384-394
- 4. Schiller LR, Pardi DS, Spiller R, Semrad CE, Surawicz CM, Giannella RA, Krejs GJ, Farthing MJ, Sellin JH, (2014) Gastro 2013 APDW/WCOG Shanghai working party report: chronic diarrhea: definition, classification, diagnosis. J Gastroenterol Hepatol 29: 6-25
- 5. Hay T, Bellomo R, Rechnitzer T, See E, Ali Abdelhamid Y, Deane AM, (2019) Constipation, diarrhea, and prophylactic laxative bowel regimens in the critically ill: A systematic review and meta-analysis. Journal of Critical Care
- 6. Organization WH, (2005) The treatment of diarrhoea: A manual for physicians and other senior health workers: 1-43
- 7. Lewis SJ, Heaton KW, (1997) Stool form scale as a useful guide to intestinal transit time. Scandinavian Journal of Gastroenterology 32: 920-924
- 8. Crobach MJ, Planche T, Eckert C, Barbut F, Terveer EM, Dekkers OM, Wilcox MH, Kuijper EJ, (2016) European Society of Clinical Microbiology and Infectious Diseases: update of the diagnostic guidance document for Clostridium difficile infection. Clin Microbiol Infect 22 Suppl 4: S63-81
- 9. Caroff DA, Edelstein PH, Hamilton K, Pegues DA, Program CDCPE, (2014) The Bristol stool scale and its relationship to Clostridium difficile infection. J Clin Microbiol 52: 3437-3439
- 10. Bliss D, (2001) Reliability of a stool consistency classification system. Journal of WOCN 28: 305-313
- 11. Zimmaro Bliss D, Dhamani KA, Savik K, Kirk K, (2003) Tool to classify stool consistency: Content validity and use by person of diverse cultures. Nursing and Health Sciences: 115-121
- 12. Bishop S, Young H, Goldsmith D, Buldock D, Chin M, Bellomo R, (2010) Bowel motions in critically ill patients:a pilot observational study. Critical Care Resuscitation 12: 182-185
- 13. Jack L, Coyer F, Courtney M, Venkatesh B, (2010) Diarrhoea risk factors in enterally tube fed critically ill patients: a retrospective audit. Intensive Crit Care Nurs 26: 327-334
- 14. Tirlapur N, Puthucheary ZA, Cooper JA, Sanders J, Coen PG, Moonesinghe SR, Wilson AP, Mythen MG, Montgomery HE, (2016) Diarrhoea in the critically ill is common, associated with poor outcome, and rarely due to Clostridium difficile. Sci Rep 6: 24691
- 15. Marcon AP, Amaral L, Carneiro, V (2006) Nosocomial diarrhea in the intensive care unit. The Brizilian Journal of Infectious Disease 10: 384-389

- Reintam A, Parm P, Redlich U, Tooding LM, Starkopf J, Kohler F, Spies C, Kern H, (2006) Gastrointestinal failure in intensive care: a retrospective clinical study in three different intensive care units in Germany and Estonia. BMC Gastroenterol 6: 19
- 17. Reintam A, Parm P, Kitus R, Kern H, Starkopf J, (2009) Gastrointestinal symptoms in intensive care patients. Acta Anaesthesiol Scand 53: 318-324
- 18. Taito S, Kawai Y, Liu K, Ariie T, Tsujimoto Y, Banno M, Kataoka Y, (2019) Diarrhea and patient outcomes in the intensive care unit: Systematic review and meta-analysis. J Crit Care 53: 142-148
- Dionne JC, Sullivan K, Mbuagbaw L, Takaoka A, Duan EH, Alhazzani W, Devlin JW, Duprey M, Moayyedi P, Armstrong D, Thabane L, Tsang JLY, Jaeschke R, Hamielec C, Karachi T, Cartin-Ceba R, Muscedere J, Alshahrani MSS, Cook DJ, (2019) Diarrhoea: interventions, consequences and epidemiology in the intensive care unit (DICE-ICU): a protocol for a prospective multicentre cohort study. BMJ Open 9: e028237
- 20. Dionne JC, Sullivan K, Mbuagbaw L, Al-Hazzani W, Karachi T, Hamielec C, Tsang J, Takaoka A, Cook DJ, (2016) Tu1068 Diarrhea in the Intensive Care Unit: Epidemiology (The DICE Study). Gastroenterology 150
- 21. von Elm E, Altman DG, Egger M, Pocock SJ, Gotzsche PC, Vandenbroucke JP, Initiative S, (2008) The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. J Clin Epidemiol 61: 344-349
- 22. Knaus WA, Draper EA, Wagner DP, Zimmerman JE, (1985) APACHE II: a severity of disease classification system. Crit Care Med 13: 818-829
- 23. StataCorp., (2019) Statistical Software: Release 16
- 24. Lindsay PJ, Rohailla S, Taggart LR, Lightfoot D, Havey T, Daneman N, Lowe C, Muller MP, (2019) Antimicrobial Stewardship and Intensive Care Unit Mortality: A Systematic Review. Clin Infect Dis 68: 748-756
- 25. Pickens CI, Wunderink RG, (2019) Principles and Practice of Antibiotic Stewardship in the ICU. Chest 156: 163-171
- 26. McClave SA, Taylor BE, Martindale RG, Warren MM, Johnson DR, Braunschweig C, McCarthy MS, Davanos E, Rice TW, Cresci GA, Gervasio JM, Sacks GS, Roberts PR, Compher C, Society of Critical Care M, American Society for P, Enteral N, (2016) Guidelines for the Provision and Assessment of Nutrition Support Therapy in the Adult Critically III Patient: Society of Critical Care Medicine (SCCM) and American Society for Parenteral and Enteral Nutrition (A.S.P.E.N.). JPEN J Parenter Enteral Nutr 40: 159-211
- 27. Dionne JC, Campbell T, Janisse N, Zytaruk N, Saunders L, Heels-Ansdell D, Francois L, Johnstone J, Cook DJ, (2020) Effect of fiber, osmolarity, and protein content of enteral nutrition on the development of diarrhea in critical illness. Gastroenterology 158
- McDonald LC, Gerding DN, Johnson S, Bakken JS, Carroll KC, Coffin SE, Dubberke ER, Garey KW, Gould CV, Kelly C, Loo V, Shaklee Sammons J, Sandora TJ, Wilcox MH, (2018) Clinical Practice Guidelines for Clostridium difficile Infection in Adults and Children: 2017 Update by the Infectious Diseases

Society of America (IDSA) and Society for Healthcare Epidemiology of America (SHEA). Clin Infect Dis 66: e1-e48

- 29. McDonald D, Ackermann G, Khailova L, Baird C, Heyland D, Kozar R, Lemieux M, Derenski K, King J, Vis-Kampen C, Knight R, Wischmeyer PE, (2016) Extreme Dysbiosis of the Microbiome in Critical Illness. mSphere 1
- 30. Lankelma JM, van Vught LA, Belzer C, Schultz MJ, van der Poll T, de Vos WM, Wiersinga WJ, (2017) Critically ill patients demonstrate large interpersonal variation in intestinal microbiota dysregulation: a pilot study. Intensive Care Med 43: 59-68
- 31. Lamarche D, Johnstone J, Zytaruk N, Clarke F, Hand L, Loukov D, Szamosi JC, Rossi L, Schenck LP, Verschoor CP, McDonald E, Meade MO, Marshall JC, Bowdish DME, Karachi T, Heels-Ansdell D, Cook DJ, Surette MG, Investigators P, Canadian Critical Care Trials G, Canadian Critical Care Translational Biology G, (2018) Microbial dysbiosis and mortality during mechanical ventilation: a prospective observational study. Respir Res 19: 245
- 32. Iapichino G, Callegari ML, Marzorati S, Cigada M, Corbella D, Ferrari S, Morelli L, (2008) Impact of antibiotics on the gut microbiota of critically ill patients. J Med Microbiol 57: 1007-1014
- 33. Iapichino G, Lankelma JM, Joost Wiersinga W, (2017) Gut microbiota disruption in critically ill patients : Discussion on "Critically ill patients demonstrate large interpersonal variation of intestinal microbiota dysregulation: a pilot study". Intensive Care Med 43: 718-719

Appendix

Table 1: Diarrhea risk factors based on the Bristol Stool Chart and Bliss Classification

Model	Multivariable Mode	I Bristol	Multivariable Model Bristol	
Covariates	Adjusted OR (95% CI)	P -value	Adjusted RR (95%Cl)	P -value
Sex	1.13 (1.00 - 1.29)	0.053	1.03 (0.98 - 1.08)	0.268
Age	1.01 (1.00 - 1.02)	0.062	1.00 (1.00 - 1.01)	0.034
APACHE II score	1.02 (0.99 - 1.05)	0.214	1.00 (1.00 - 1.01)	0.276
Opiates	0.58 (0.3 - 1.13)	0.111	0.84 (0.65 - 1.08)	0.167
Chemotherapy	1.24 (0.56 - 2.72)	0.598	1.06 (0.86 - 1.32)	0.567
Antibiotics (total #)	1.24 (0.95 - 1.62)	0.115	1.05 (1.01 - 1.1)	0.019
Antibiotic days	1.19 (1.03 - 1.37)	0.016	1.03 (1.02 - 1.04)	<0.001
Motility Agent	0.9 (0.68 - 1.18)	0.432	0.96 (0.89 - 1.04)	0.347
Sorbitol	1.71 (1.00 - 2.94)	0.051	1.2 (0.98 - 1.46)	0.078
Suppository	1.37 (0.98 - 1.92)	0.065	1.14 (1.02 - 1.27)	0.023
Enteral Nutrition	1.75 (1.19 - 2.57)	0.004	1.30 (1.13 - 1.5)	<0.001
Acid Suppressants	1.39 (0.76 - 2.54)	0.285	1.16 (0.88 - 1.52)	0.287
Gastrointestinal comorbidities**	1.49 (0.89 - 2.48)	0.130	1.11 (0.93 - 1.33)	0.236
AIC	1.202		1.290	

Table 1A: Multivari	able models for Diarrhea	(Bristol Stool Chart definition) (RR)
	Bristol incidence: 53 5%	(95% CI 50 4-56 4)

*Models adjusted for centre; ** Celiac disease, Prior bowel resection surgery, Inflammatory bowel disease (Crohn disease, ulcerative colitis), Colectomy/Ileostomy, Chronic Pancreatitis, Gastroparesis. AIC= Akaike Information Criterion.

	Bliss incidence: 37.7% (95% Cl 34.8-40.6)				
Model	Multivariate Model	Bliss	Multivariate Model Bliss		
Covariates	Adjusted OR (95% CI)	P -value	Adjusted RR (95%Cl)	P -value	
Sex	1.35 (1.01 - 1.8)	0.043	1.11 (1.01 - 1.22)	0.030	
Age	1.01 (1.00 - 1.02)	0.141	1.00(1.00 - 1.01)	0.050	
APACHE II score	1.02 (0.99 - 1.04)	0.178	1.00 (0.99 - 1.01)	0.429	
Opiates	0.93 (0.59 - 1.46)	0.753	1.02 (0.74 - 1.42)	0.886	
Chemotherapy	1.49 (0.64 - 3.45)	0.351	1.25 (0.96 - 1.62)	0.101	
Antibiotics (total			1.06 (1.01 - 1.12)	0.019	
#)	1.15 (0.92 - 1.44)	0.218			
Antibiotic days	1.11 (1.01 - 1.22)	0.023	1.02 (1.01 - 1.03)	<0.001	
Motility Agent	1.13 (0.84 - 1.5)	0.425	1.04 (0.91 - 1.18)	0.576	
Sorbitol	1.64 (0.9 - 2.97)	0.105	1.32 (0.95 - 1.83)	0.097	
Suppository	1.29 (0.9 - 1.85)	0.168	1.14 (0.95 - 1.35)	0.150	
Enteral Nutrition	2.2 (1.5 - 3.24)	<0.001	1.67 (1.38 - 2.01)	<0.001	
Acid		<0.001	1.66 (1.15 - 2.4)	0.007	
Suppressants	2.17 (1.44 - 3.28)				
Gastrointestinal			1.12 (0.93 - 1.35)	0.235	
comorbidities**	1.51 (0.92 - 2.48)	0.102			
AIC	1.206		1.267		

Table 1B: Multivariable models for Diarrhea (Bliss Stool Classification System)(RR)

*Models adjusted for centre; ** Celiac disease, Prior bowel resection surgery, Inflammatory bowel disease (Crohn disease, ulcerative colitis), Colectomy/Ileostomy, Chronic Pancreatitis, Gastroparesis. AIC= Akaike Information Criterion.

Figure 1: CDAD DICE-ICU Incidence and Prevalence



	WHO Incidence: 73.8% (95% CI 71.1-76.6)		
Model	Multivariable model (full)*		
Covariates	Adjusted RR (95%CI)	P -value	
Sex	0.95 (0.90 - 1.01)	0.083	
Age	1.00 (1.00 – 1.00)	0.630	
APACHE II score	1.00 (1.00 – 1.00)	0.786	
Opiates	0.78 (0.69 - 0.89)	<0.001	
Chemotherapy	1.00 (0.87 - 1.15)	0.994	
Antibiotics (total #)	1.03 (1.01 - 1.06)	0.003	
Antibiotic days	1.03 (1.02 - 1.04)	<0.001	
Motility Agent	1.04 (0.99 - 1.11)	0.142	
Sorbitol	1.04 (0.97 - 1.12)	0.289	
Suppository	1.13 (1.06 - 1.20)	<0.001	
High protein	0.97 (0.93 - 1.02)	0.308	
High osmolarity	1.14 (1.08 - 1.20)	<0.001	
High fibre	1.11 (1.05 - 1.17)	<0.001	
Acid Suppressants	1.08 (0.95 - 1.22)	0.258	
Gastrointestinal comorbidities**	1.12 (1.03 - 1.22)	0.010	
AIC	0.981		

Table 2: Enteral Nutrition Components Associated with Diarrhea (WHO Definition)

*Model adjusted for centre; ** Celiac disease, Prior bowel resection surgery, Inflammatory bowel disease (Crohn disease, ulcerative colitis), Colectomy/Ileostomy, Chronic Pancreatitis, Gastroparesis. AIC= Akaike Information Criterion.

Table 3: Consequences of Diarrhea

Variable	Death		ICU stay		Hospital stay	
	RR (95% CI)	Ρ	RR (95% CI)	Ρ	RR (95% CI)	Р
Diarrikaa	0.78 (0.58 -		1.06 (0.99 -		1.06 (0.94 -	
Diarriea	1.03)	0.081	1.15)	0.108	1.19)	0.337
Sex	1.01 (0.82 -		1.00(0.96 -		0.99 (0.93 -	
	1.24)	0.932	1.05)	0.854	1.05)	0.755
Ago	1.02 (1.01 -		1.00 (1.00 –		1.00 (0.99 –	
Age	1.04)	0.002	1.00)	0.231	1.00)	0.004
APACHE II	1.04 (1.02 -		0.99 (0.98 -		0.98 (0.98 -	
score	1.06)	<0.001	0.99)	<0.001	0.99)	<0.001
AIC*	1.018		0.724		1.032	

Table 3A: Consequences of Diarrhea (WHO Definition)

AIC= Akaike Information Criterion.

Table 3B: Consequences of Diarrhea (Bristol Stool Chart)

Variable	Death		ICU stay		Hospital stay	
	RR (95% CI)	Р	RR (95% CI)	Р	RR (95% CI)	Р
Diarrhoa	0.91 (0.67 -		1.04 (0.96 -		1.03 (0.92 -	
Diarmea	1.24)	0.567	1.13)	0.379	1.14)	0.637
Cov	1.02 (0.83 -		1.00 (0.96 -		0.99 (0.93 -	
Sex	1.26)	0.849	1.05)	0.997	1.05)	0.678
Age	1.02 (1.01 -		1.00 (1.00 –		1.00 (0.99 –	
	1.04)	0.002	1.00)	0.230	1.00)	0.005
APACHE II	1.04 (1.02 -		0.98 (0.98 -		0.98 (0.98 -	
score	1.06)	<0.001	0.99)	<0.001	0.99)	<0.001
AIC*	1.024		0.726		1.034	

AIC= Akaike Information Criterion.

Table 3C: Consed	nuences of Diarrhea	(Rliss Stool	Classification	System)
	fuences or Diarriea	(DII35 31001	Classification	Systemy

Variable	Death		ICU stay		Hospital stay	
	RR (95% CI)	Ρ	RR (95% CI)	Ρ	RR (95% CI)	Ρ
Diarrhea	1.10 (0.82 -		1.03 (0.95 -		1.00 (0.90 -	
	1.49)	0.518	1.11)	0.466	1.10)	0.939
Sex	1.02 (0.82 -		1.00 (0.96 -		0.99 (0.93 -	
	1.26)	0.888	1.05)	0.985	1.05)	0.686
Age	1.02 (1.01 -		1.00 (1.00 –		1.00 (0.99 –	
	1.04)	0.001	1.00)	0.244	1.00)	0.005

APACHE II	1.04 (1.02 -		0.98 (0.98 -		0.98 (0.98 -	
score	1.06)	<0.001	0.99)	<0.001	0.99)	<0.001
AIC*	1.024		0.727		1.034	

AIC= Akaike Information Criterion.

Table 4: Center various in diarrhea incidence

Proportion with Diarrhea

centre		ES (95% CI)
SJHH	·	0.78 (0.68, 0.86)
HGH		0.84 (0.72, 0.92)
JH	_ _	0.76 (0.66, 0.85)
Niagara	_ 	0.67 (0.56, 0.77)
Brantford GH	2	0.76 (0.69, 0.83)
JBH		· 0.94 (0.83, 0.99)
TUMC		0.67 (0.55, 0.77)
Mayo Arizona		0.54 (0.39, 0.69)
KGH		0.78 (0.71, 0.83)
UDamman		0.84 (0.72, 0.92)
JUMS		0.70 (0.60, 0.79)
SJHC TO	_ _	0.53 (0.41, 0.65)
Total	+	0.74 (0.71, 0.76)

0 .1 .2 .3 .4 .5 .6 .7 .8 .9 1

CHAPTER 4

Content Analysis of Bowel Management Protocols For the Management of Constipation in Adult Critically III Patients (Published in Journal of Critical Care 58(2020) 98-104)

Content Analysis of Bowel Management Protocols For The Management of Constipation in Adult Critically III Patients

Joanna C. Dionne MD, MSc PhD (C) FRCP(C) ^{1,2}, Jennie Johnstone MD PhD FRCP(C) ^{3,4},Orla Smith RN PhD^{5,6}, Louise Rose RN PhD ^{7,8}, Simon Oczkowski MD MHSc MSc FRCP (C)^{1,2}, Yaseen Arabi MD FCCP FCCM^{9,10,11}, Erick H. Duan MD FRCP(C)^{1,2}, François Lauzier MD MSc FRCP (C)^{12,13,14}, Waleed Alhazzani MD MSc FRCP(C)^{1,2}, Norine Alam MD¹⁵, Nicole Zytaruk RN², Josie Campisi MSc¹², Deborah J Cook MD MSc FRCP(C)^{1,2}

1 Department of Medicine, McMaster University, 2. Department of Health Research Methods, Evidence and Impact, McMaster University, Hamilton, Ontario, Canada 3. Department of Laboratory Medicine and Pathobiology, University of Toronto, Toronto, Ontario, Canada, 4, Public Health Ontario, Toronto, Ontario, Canada, 5. Heart and Vascular Program and Critical Care Department, St. Michael's Hospital, Toronto, Ontario, Canada, 6. Li Ka Shing Knowledge Institute, St. Michael's Hospital, Toronto, Ontario, Canada, 7. Department of Critical Care Medicine, Sunnybrook Health Sciences Centre, Toronto, Canada, 8. Florence Nightingale Faculty of Nursing, Midwifery and Palliative Care, King's College London, London, UK, 9. College of Medicine, King Saud Bin Abdulaziz University for Health Sciences, Riyadh, Kingdom of Saudi Arabia, 10. King Abdullah International Medical Research Center, 11. Intensive Care Department, Ministry of National Guard Health Affairs, Riyadh, Kingdom of Saudi Arabia, Riyadh, Kingdom of Saudi Arabia, 12. CHU de Québec - Université Laval Research Center, Population Health and Optimal Health Practices Research Unit (Trauma – Emergency – Critical Care Medicine), Université Laval, Québec City, Québec, Canada. 13. Division of Critical Care Medicine. Department of Anesthesiology and Critical Care Medicine. Université Laval, Québec City, Québec, Canada. 14. Department of Medicine, Université Laval, Québec City, Québec, Canada. 15. McGill University Health Centre Research Institute, Department of Critical Care, Montreal General Hospital, Montréal, Québec, Canada,

Correspondence:

DJ Cook, Departments of Medicine & Health Research Methods, Evidence and Impact, McMaster University Health Sciences Center, Room 2C11, 1200 Main Street West, Hamilton, Ontario, Canada, L8N 3Z5; debcook@mcmaster.ca

Funding:

This study was peer-review funded by the Hamilton Academy of Health Sciences Organization, the Physicians Services Incorporated of Ontario, the Canadian Frailty Network, and the Canadian Institutes for Health Research.

Key Words: gastrointestinal, diarrhea, bowels, critical care

Conflict of Interest: None declared

Reprints: Not available

Abstract:

Objective: Alterations in bowel habits are common in the intensive care unit (ICU), and bowel protocols are gaining acceptance. Our objective was to characterize the content of bowel protocols in a cross-sectional analysis of ICUs.

Design: We engaged 44 adult ICUs to perform a content analysis of available bowel protocols, addressing 1) initiation criteria, 2) medications incorporated, 3) medication escalation, 4) discontinuation criteria, 5) stool assessment methods and 6) bowel protocol contraindications.

Setting: ICUs in Canada, the United States and Saudi Arabia

Patients: Adult patients admitted to medical and surgical ICUs in academic or community centers participating in a probiotics trial.

Interventions: None.

Measurement and Main Results: Bowel protocols were operant in 33 of 44 ICUs (79.5%). The most common medications were senna (81.0%) and bisacodyl (75.6%). Less common agents were sodium phosphate (45.9%) glycerin (43.2%), docusate sodium (43.2%), polyethylene glycol 3350 (37.8%), lactulose (29.7%), sodium citrate (16.2%), milk of magnesia (13.5%) and mineral oil (16.2%). Bowel protocols were activated by nurses (62.8%) based on initiation criteria; including no bowel movement for 24-96 hours (35.1%); opioid use (18.9%); "at risk for constipation" (13.5%); stool on digital rectal exam (10.8%); feeding initiation (10.8%); and ICU admission (8.1%). Criteria for laxative escalation included time from last bowel movement (59.4%), opioid use (18.9%), and no stool on digital rectal exam (10.8%), while 15 (40.5%) bowel protocols included diarrhea as a discontinuation criterion.

Conclusions: Bowel protocols have variable initiation, escalation, and discontinuation criteria incorporating different classes of laxatives, reflecting unclear evidence about optimal bowel management strategies in the ICU.

Word Count: 2657 Abstract Word Count: 256 Tables: 2 Figures: 2 Supplemental Appendix Table: 1

Introduction

Alterations in bowel habits is common in the intensive care unit (ICU). The literature regarding gastrointestinal dysfunction in the ICU is growing, however the evidence is not as robust compared to other research disciplines in critical care such as sepsis, acute respiratory distress syndrome and cardiopulmonary research. In the areas of gastrointestinal dysfunction, the interest in optimizing the treatment of constipation, diarrhea and prophylactic laxative bowel management is growing however, more research is needed. A previously conducted survey identified clinicians working the ICU felt dissatisfied to very dissatisfied with the bowel management programs at their centers and felt this is a neglected area of critical care (Knowles, 2010).

Bowel management protocols have been used to treat and prevent constipation as well as prophylactically to treat non-defecation (Hay 2019). The challenge remains in the balance of treating or preventing constipation can result in the development of diarrhea. In observational studies, constipation has been associated with increased length of hospital stay, delirium, feeding intolerance as well as increased duration of mechanical ventilation and longer length of hospital stay [3]. In Contrast, diarrhea in critically ill patients can be associated with feeding intolerance or modification of enteral nutrition, electrolyte disturbances, renal failure and skin breakdown [4]. Diarrhea can result in unnecessary testing for *Clostridioides difficile*, isolation and other interventions (e.g., insertion of fecal management devices). (Dionne 2016; Hay 2019).

Hay and colleagues (2019) conducted a meta-analysis and systematic review exploring the epidemiology of constipation and diarrhea in the ICU and the impact of prophylactic laxative bowel regimens on patient outcomes. Diarrhea and constipation

found to be common with variability and definitions applied (Hay 2019). Prophylactic bowel regimens were found to reduce constipation, may increase the risk of diarrhea, but did not impact patient important outcomes (Hay 2019). The review found similar studies to the previously published systematic review by Oczkowski et al (2017) exploring the impact of bowel protocols on feeding intolerance, constipation, and duration of mechanical ventilation reported variable results. The use of bowel protocols likely reduces the risk of constipation (relative risk [RR] 0.50, 95% confidence interval (Cl) 0.25 to 1.01), although having little impact on duration of mechanical ventilation (mean difference 0.01 days, 95% Cl -2.67 to 2.69) or feeding tolerance (RR 0.94, 95% Cl 0.62 to 1.42) [1]. A key message of these two systematic reviews is the general dearth of research on bowel protocols in critically ill patients with respect to patient important outcomes and different mechanisms of laxative use (Hay 2019; Oczkowski, 2017). The gap in the current body of research is what bowel protocol strategies are being implemented at the bedside in ICUs.

The primary objective of this study was to characterize the content of bowel protocols for the prevention of constipation implemented in ICUs in centers that participated in a randomized trial examining the effects of probiotics for critically ill patients to prevent ICU-acquired infections [Probiotics: Prevention of Severe Pneumonia and Endotracheal Colonization Trial (PROSPECT) <u>NCT02462590</u>]. Our secondary objective was to characterize: 1) initiation criteria; 2) medications incorporated; 3) approaches to protocol escalation; 4) discontinuation criteria; 5) stool assessment methods; and 6) contraindications.

Materials and Methods

This observational study was nested within the PROSPECT trial that allocated patients to receive *Lactobacillus rhamnosus* GG (Culturelle, Locin Industries Ltd) or identical placebo, administered enterally twice daily for up to 60 days while in the ICU [5]. The primary outcome of the trial was ventilator-associated pneumonia and secondary outcomes included other nosocomial infections and diarrhea. The management of diarrhea and constipation was at the discretion of the treating team. The existence of a bowel protocol was not a criterion for a center's participation in the trial.

We asked the 44 PROSPECT participating centers in Canada, the United States, Saudi Arabia, (as listed in supplement Table 1) to complete a site status form indicating whether they had a bowel protocol; if they did, our research project manager requested a copy. Bowel protocols were defined as any document that was explicitly initiated to promote bowel movements in critically ill patients. They could be stand alone documents or incorporated into the admission order set as a supplemental treatment plan. Protocols that pertained only to the care of fecal management devices were excluded.

We used content analysis and descriptive statistics (frequency and percentages) to characterize the content of existing bowel protocols. Content analysis is a research method using qualitative and/or quantitative analysis of documents to describe a phenomenon of interest [6], as used commonly in the fields of nursing, psychiatry, geriatrics, and public health [6]. The domains of interest for the classification schema were established by discussion between two investigators (JCD and DJC) and informed

by review of the bowel protocols collated; thereafter, content was abstracted by one investigator (JCD). We did not contact participating centers to document the year that the bowel protocol was developed, the frequency and focus of any updates, or the professional background of the developers. We analyzed the bowel protocols for: 1) initiation criteria, 2) medications incorporated, 3) medication escalation approaches, 4) discontinuation criteria, 5) stool assessment methods and 6) contraindications to bowel protocols.

Ethics

The PROSPECT trial received ethics approval at the Hamilton Integrated Research Ethics Board and all participating centers.

Results

All 44 ICUs responded regarding the presence or absence of a bowel protocol (100% response rate); 33 (79.5%) had at least one bowel protocol, accounting for a total of 37 bowel protocols. Of the 33 ICUs that had bowel protocols, 7 were community hospitals and 26 were academic hospitals. Of these 37 protocols, 22 protocols (59.4%) were stand-alone documents, and the remaining 15 protocols (40.5%) were embedded into ICU admission orders. Six ICUs across 5 hospitals had separate protocols for spinal cord injury patients, 4 ICUs had protocols for patients receiving opioids, and 3 sites had different protocols for surgical, cardiac and medical ICUs. Two protocols were excluded from 2 centers because they only focused on the use of fecal management devices. The 11 hospitals that did not have a specific bowel protocol were academic ICUs in Canada and the United States.

Figure 1 summarizes the laxatives used in the 37 bowel protocols. Stimulant and osmotic drug classes were the most commonly utilized in the protocols reviewed. The most commonly used stimulant laxatives included senna (n=30, 81.0%) and bisacodyl (n=28, 75.6%). Other laxatives were sodium phosphate (n=17, 45.9%), glycerin (n=16, 43.2%), docusate sodium (n=16, 43.2%), polyethylene glycol (PEG)-3350 (n=14, 37.8%), lactulose (n=11, 29.7%) and sodium citrate (n=6, 16.2%). Mineral oil (n=3, 8.1%) and milk of magnesia (n=5, 13.5%) were less often incorporated. Route of administration of laxatives included oral, suppository and enema delivery. Of the 10 different laxatives identified in the protocols, the dosage range varied widely. None of the bowel protocols included recommendations for use of prokinetics or provided explicit guidance in terms of daily maintenance therapy to prevent constipation (e.g., daily dosing of PEG-3350 to prevent constipation). Other than enteral nutrition being an initiation criterion, none of the bowel protocols recommended the addition of fiber to feeds to treat constipation or incorporated strategies to minimizing opioids. Only one bowel protocol indicated the need for an abdominal x-ray if no bowel movement occurred after use of laxative therapy.

The decision to activate a bowel protocol was typically driven by the bedside nurse (n=23, 62.8%) based on initiation criteria outlined in the protocol ordered by a physician. These criteria included: 1) no bowel movement for 24-96 hours (n=13, 35.1%; 1 protocol had a 24-hour criterion,7 protocols used 48 hours, 3 protocols used at least 72 hours and 2 protocols had greater than 96 hours as a criterion); 2) opioid use (n=7, 18.9%); 3) "at risk for constipation" without definition (n=5, 13.5;%); 4) stool on digital rectal exam performed by nurses (n=4, 10.8;%); 5) enteral nutrition initiation (n=4, 10.8;%); 5) enteral nutrition (n=4, 10.8;%); 5) enteral nutrition (n=4, 10.8;

10.8%); and 6) ICU admission (n=3, 8.1%). The remaining protocols had no explicit initiation criteria (Figure 2).

Criteria for laxative escalation included time from last bowel movement (n=22, 59.4%); opioid use (n=7, 18.9%); or no stool on digital rectal exam (n=6, 16.2%). Escalation strategies included increasing the frequency of a laxative, combination therapy with other medications, and/or the addition of enema therapy or additional oral therapy. The remaining protocols did not specify when to change drug dose or class. Fifteen (40.5%) bowel protocols included diarrhea as a discontinuation criterion (Table 1). The protocols that did not have an initiation criterion, could still be escalated by the bedside nurse as there were set orders for laxatives on an "as-needed basis". Only 2 protocols included the use of the Bristol Stool chart and a site-specific definition. While this analysis was not focused on the use of or fecal management devices, rectal tubes were incorporated into a possible definition of diarrhea (e.g., >500mls of stool via rectal tube) at one center.

Contraindications to the use of bowel protocols (Table 2) were present in 15 (40.5%) protocols and included: 1) renal disease (n=8, 21.6%; specifically use of sodium phosphate enemas); 2) major abdominal surgery or bowel obstruction (n=8, 21.6%); 3) neutropenia or bone marrow transplant (n=5, 13.5%); 4) nausea, vomiting or undiagnosed abdominal pain (n=4, 10.8%); 5) platelet count < 50 x 10⁹/L. (n=2, 5.2%); and 6) fecal impaction (n=1, 2.7%).

Discussion

In this study, most ICUs participating in the PROSPECT trial reported having a bowel protocol at their institution. These protocols were highly variable in their constitution with differing criteria for initiation, medications used, contraindications to ongoing usage as well as their escalation. Very few bowel protocols had discontinuation criteria, which is important to minimize the risk of iatrogenic diarrhea. Only two protocols were guided by a stool assessment chart, which can objectively and consistently characterize and track bowel movements [7]. The lack of explicit tracking of bowel movements in response to use of a bowel protocol may relate to the fact that no specific stool chart had been validated in the critical care setting until just recently [8]. Although, not previously validated, the Bristol Stool Chart has been used in the ICU research setting for the study of diarrhea in 44 patients (Bishop, 2010).

The findings of this study demonstrate that time of the last bowel movement was most commonly used as a bowel protocol initiation or escalation criterion. However, protocols lacked maintenance therapy (e.g., daily dosing of laxative therapy to maintain regularity of bowel movements). Timely initiation of bowel protocols is challenging in critically ill adults due to lack of evidence about patients' pre-existing "normal" bowel habits, and generally unclear optimal or "personalized" initiation criteria. Bowel protocols for the prevention and treatment of constipation, prophylactic laxatives in non defecation, prevention of opioid induced constipation, will likely require different initiation criteria laxative classes, doses and will have different outcomes. For example, a prophylactic bowel protocol may decrease constipation in this population when initiated early. In an observational study [9], early initiation of prophylactic PEG-4000 was

associated with less gastrointestinal paralysis compared to its initiation following 4 days without a bowel movement.

The bowel protocols analyzed in this study incorporated a variety of medications - most commonly stimulant laxatives such as senna. Surprisingly, many protocols did not include medications studied in previous randomized control trials of bowel protocols such as lactulose and PEG [1]. Different regimens of such medications include: 1) lactulose administered every 8 hours compared to usual care [10]; 2) lactulose administered every 12 hours compared to usual care [10]; 3) lactulose and PEG administered every 8 hours versus placebo (including rescue therapy with enemas and intravenous neostigmine if patients enrolled in the treatment arm had not had a bowel movement for 7 days after being treated with PEG) [11]; and 4) lactulose administered 4 times daily compared to placebo [12]. In the context of limited evidence, the development of future ICU bowel protocols should include lactulose and PEG, as these drugs appear to be the primary laxatives employed.

A recent single centre pilot cluster cross-over RCT (Hay 2019) that study three regimens: 1) docusate sodium with senna tables BID when feeding was initiated. If no bowel movement by day 5 an osmotic laxative was added (movicol sachet) administered daily, 2) No laxatives administered, until day 3 after feeding initiation at which point docusate sodium, senna and 20 mls of lactulose were administer twice a day and 3) no laxative administration until after 6 days of feeding, when patient was administered docusate sodium, senna and 20 mls of lactulose twice a day (Hay 2019). The primary outcome was insertion of a rectal tube for diarrhea. Secondary and tertiary outcomes included diarrhea, ileus, obstruction, skin breakdown, duration of mechanical

ventilation, ICU length of stay, and 30-day mortality (Hay 2019). A total of 570 patients were enrolled across all regimens. A total of 53/570 (9.3%) patients required a rectal tube, 78/570 (13.7%) developed diarrhea, 13/570 (2.3%) developed ileus or obstruction, *Clostridioides Difficile* occurred in 1 patient. However, 18/570 (3.2%) developed local pressure ulcers. In terms of terirary outcomes, duration of mechanical ventilation was 81.4 hours (39.0-156.0), ICU length of stay 4.9 days (3.0-8.0) and 30-day mortality 108/570 (18.9%) (Hay 2019). The authors found that withholding laxatives until day 6 of feeding did not result in an increase rate of complications (Hay 2019). More research is needed to examine which regimens and dosages are the most effective for incorporating into protocols, as well as which initiation and escalation criteria are most appropriate. In the bowel protocols we examined, concomitant feeding formulae were not well described. Further research is required to identify the impact of enteral nutrition and the addition of fiber to feeds to prevent constipation and diarrhea, so as to inform synergistic strategies to manage gut motility.

It is important to highlight that respect to prophylactic laxative bowel protocols, that there is no evidence that non-defecation needs to be prevented. It is important that future studies not only examined the reason for the bowel protocols (i.e. opioid use, constipation, prophylactic) but should address which regimens and medication should be used, incorporate stool assessment charts, and incorporate nutrition alterations (i.e. addition of fiber, change in rate and location of feeding administration. It is also vital that future research assesses how these protocols are implemented and if these protocols improve patient outcomes compared to a control (Hay, 2019; Hay 2019).
Strengths of this study include the identification of all contemporary ICU-specific bowel protocols and their detailed characterization using document analysis. We characterized 6 dimensions of 37 bowel protocols in centers caring for heterogeneous critically ill patients. These protocols represent tools available for diverse practice settings including both academic and community centers participating in the PROSPECT trial from different countries.

Our study has several limitations. We did not critically appraise the evidence behind each component of these protocols, which is in general limited, nor did we collect stool management methods encoded in electronic medical records or bowel protocol utilization profiles. We did not survey institutions which did not have a bowel protocol to inquire about why not. We did not examine how these bowel protocols were developed, implemented at the bedside or measure compliance. Another limitation is the confounding that was inherent into the treatment goals for the bowel protocol implementation including prevention of constipation, treatment of constipation and opioid use and its effect on gut motility. Finally, selection bias may have been introduced by the convenience sample of centers participating in the PROSPECT trial, although these hospitals reflect a diverse group of ICUs.

Future bowel protocols should incorporate explicit definitions of constipation and diarrhea in this patient population, in addition to the utilization of stool tracking tools.

Conclusion

In summary, we found that bowel protocols have variable and sometimes unspecified initiation criteria, unclear escalation and discontinuation criteria. Uncertainty about optimal bowel management strategies is reflected in these protocols, underscoring how more research is needed in this often-neglected domain of practice. Further investigations focusing on the class and dose of effective strategies for critically ill patients is needed.





Legend for Figure 1: Medications included in all bowel protocols are shown in this table. Protocols included a combination of medications, such that percentages sum to more than 100%.





Legend for Figure 2: The criteria cited for initiating bowel protocols are presented here. Some bowel protocols had more than one criterion for initiation. The bowel protocols did not define what "at risk for constipation" meant.

Criteria for Medication Escalation	N (%)
Time from last bowel movement	22 (59.4)
Opioid use	7 (18.9)
No stool on digital rectal exam	6 (16.2)
No criteria described	6 (16.2)
Criteria for Protocol Discontinuation	N (%)
Diarrhea	15 (40.5)

Table 1: Medication Escalation and Bowel Protocol Discontinuation Criteria

Legend for Table 1: The criteria incorporated for escalation of medication are presented here. Some bowel protocols had more than one criterion for escalation of therapy, such that percentages add up to more than 100%. Less than half of the bowel protocols had an explicit discontinuation criterion.

Table 2: Contraindications to Bowel Protocols

Contraindications	N (%)
Renal disease	8 (21.6)
Major abdominal surgery/bowel obstruction	8 (21.6)
Neutropenia (or bone marrow transplant)	5 (13.5)
Nausea and vomiting, undiagnosed abdominal pain	4 (10.8)
Platelet count < 50,000 x 10 ⁹ /L	2 (5.2)
Fecal impaction	1 (2.7)

Legend for Table 2: Contraindications to bowel protocols are presented here. Some bowel protocols had more than one contraindication criterion.

Acknowledgements

We are grateful to the patients and families who participated in this study, and the Research Coordinators and Site Investigators. iHealth provided the blinded probiotic study product for the trial, within which this study was nested. None of these individuals played a role in the design, conduct, analysis, interpretation or write-up of this study. JC Dionne holds a Physicians Services Incorporated Research Trainee Award and a Canadian Association of Gastroenterology PhD Studentship Award. S Oczkowski holds an internal Early Career Investigator Award from the Department of Medicine, McMaster University. F Lauzier is a recipient of a Research Career Award from the Fonds de la recherche du Québec-Santé. D Cook holds a Chair of the Canadian Institutes of Health Research. Thank you to Dr Liz Wilcox and Dr David Williamson from the Canadian Critical Care Trials group for their diligent review and formative feedback on the manuscript.We also wish to thank the anonomyus peer reviewers from the Journal of Critical Care for the excellent and thoughtful review.

Competing Interests

None of the authors disclose any competing interests.

PROSPECT Methods Center: Nicole Zytaruk (Project Coordinator), Lois Saunders, Shelley Anderson, Mary Copland, Megan Davis, Alyson Takaoka, France Clarke, Neala Hoad, Kristine Wachmann, Diane Heels-Ansdell, (McMaster University, Hamilton, Ontario), Jennie Johnstone (University of Toronto)

PROSPECT Steering Committee: Dr. Deborah Cook, Dr. Jennie Johnstone, Dr. Maureen Meade, Dr. Francois Lauzier, Dr. John Marshall, Dr. Lehana Thabane

PROSPECT Collaborators

Canada:

St Joseph's Healthcare, Hamilton, ON

Dr Deborah Cook (Lead); Dr Erick Duan, Dr Mark Soth, (Co-Investigators); France Clarke, Mary Copland; Neala Hoad, Marnie Jakab, Melissa Shears, Alyson Takaoka, Nicole Zytaruk.

Hamilton General Hospital, Hamilton Health Science Center, Hamilton, ON

Dr Maureen Meade (Lead); Dr Emilie Belley-Cote (Co-Investigator) Lori Hand, Dr Harjot Jagdey, Lisa Klotz, Alexandra Sabev, Nevena Savja.

Juravinski Hospital Hamilton Health Science Centre, Hamilton, ON

Dr Timothy Karachi (Lead), Dr Bram Rochwerg (Co-Investigator); Mashari Alghuroba, Alia Khaled, Lauren Locco, Tina Millen, Ryan Vaisler.

St Michael's Hospital, Toronto, ON

Dr John Marshall (Lead); Dr Jan Friedrich (Co-Investigator); Jennifer Hodder, Imrana Khalid, Julie Lee, Yoon Lee, Pragma Roy, Kurtis Salway, Gyan Sandhu, Marlene Santos, Orla Smith, Melissa Wang.

St Paul's Hospital, Vancouver BC

Dr Peter Dodek (Lead); Dr Najib Ayas (Co-Investigator); Maria Agda, Victoria Alcuaz, Betty Jean Ashley, Kelsey Brewer, Janice Palmer.

Mount Sinai Hospital, Toronto, ON

Dr Geeta Mehta (Lead), Dr Stephen Lapinsky, Dr Laveena Munshi; Maedean Brown, Brittany Giacomino, Marnie Jakab, Alan Kraguljac, Sumesh Shah, Erik Tamberg, Laura Vergeer.

Hospital de l'Enfant-Jesus, Quebec City, QC

Dr Francois Lauzier (Lead); Dr Alexis Tourgeon (Co-Investigator); Danny Barriault, David Bellemare, Anick Boivin, Sarah-Judith Breton, Marjorie Daigle, Charles Delisle-Thibeault, Panagiota Giannakouros, Stephanie Grenier, Gabrielle Guilbault, Anne Laberge, Caroline Leger, Catherine Ouellet, Marie-Claude Tremblay.

Ottawa Civic Hospital, Ottawa, ON

Dr Lauralyn McIntyre (Lead) Dr Joe Pagilarello (Co-Investigator); Pierre Cardinal, Gianni D'Egidio, Shane English, Mike Hartwick, Jonathon Hooper, Gwynne Jones, John Kim, Dal Kubelik, Kwadwo Kyeremanteng, Hilary Meggison, Sherissa Microys, Dave Neiliovitz, Guiseppe Pagliarello, Rakesh Patel, Jo Po, Peter Reardon, Erin Rosenberg, Aimee Sarti, Andrew Seely (Co-Investigators); Shelley Acres, Brigette Gomes, Heather Langlois, Liane Leclair, Sydney Miezitis, Kaitlyn Montroy, Rebecca Porteous, Shawna Reddie, Irene Watpool.

Ottawa General Hospital, Ottawa, ON

Dr Lauralyn McIntyre (Lead), Dr Pierre Cardinal, Dr Gianni D'Egidio, Dr Shane English, Dr Mike Hartwick, Dr Jonathon Hooper, Dr Gwynne Jones, Dr John Kim, Dr Dal Kubelik, Dr Kwadwo Kyeremanteng, Dr Hilary Meggison, Dr Sherissa Microys, Dr Dave Neiliovitz, Dr Guiseppe Pagliarello, Dr Rakesh Patel, Dr Jo Po, Dr Peter Reardon, Dr Erin Rosenberg, Dr Aimee Sarti, Dr Andrew Seely (Co-Investigators); Shelley Acres, Heather Langlois, Liane Leclair, Sydney Miezitis, Kaitlyn Montroy, Rebecca Porteous, Shawna Reddie, Amanda Van Beinum, Allyshia Van Tol, Irene Watpool.

Vancouver General Hospital, Vancouver, BC

Dr Bill Henderson (Lead), Dr Donald Griesdale, Dr Mypinder Sekhon (Co-Investigators); Denise Foster, Suzie Logie.

University Health Network, Toronto Western Hospital, Toronto, ON

Dr Margaret Herridge (Lead), Drs Alberto Goffi, Dr Eyal Golan, Dr John Granton, Dr Victoria McCredie, Dr Elizabeth Wilcox (Co- Investigators); Jaimie Archer, Daniel Chen,

Paulina Farias, Brooke Fraser, Cheryl Geen-Smith, Barbara Kosky, Andrea Matte, Christina Pugliese, Priscila Robles, Lia Stenyk, Cristian Urrea, Karolina Walczak.

Vancouver Island Health Authority, Vancouver, BC

Dr Gordon Wood (Lead), Dr Daniel Ovakim (Co-Investigator); Fiona Auld, Gayle Camey, Ralph Fleming, Jennifer Good, Mandeep Manhas.

Toronto General Hospital, Toronto, ON

Dr Margaret Herridge (Lead), Dr Eyal Golan, Dr John Granton, (Co-Investigators); Jaimie Archer, Daniel Chen, Brooke Fraser, Cheryl Geen-Smith, Andrea Matte, Priscilia Robles, Cristian Urrea.

Sacre Coeur Hospital, Montreal, QC

Dr Emmanuel Charbonney (Lead) Dr Yoan Lamarche, Dr Soazig Leguillan, Dr Karim Serri, Dr Colin Verdant, Dr Yanick Beaulieu, Dr Patrick Bellemare, Dr Philippe Bernard, Dr Marc Giasson, Dr Véronique Brunette, Dr Alexandros Cavayas, Dr Emilie Levesque, (Co-Investigators); Halina Labikova, Julia Lainer Palacios, Marie-Eve Langlois, Virginie Williams.

Sherbrooke Hospital, Sherbrooke, QC

Dr Francois Lamontagne (Lead), Dr Fred D'Argon (Co-Investigator); Virginie Bolduc, Elaine Carbonneau, Helene Fournier, Marjolaire Frairot, Joannie Marchand, Marie-Helen Masse, Nicole Poitras, Elsa Vasse.

McGill University Health Centre (formerly Royal Victoria Hospital), Montreal, QC

Dr Arnie Kristof (Lead), Dr Peter Goldberg, Dr Roupen Hatzakorzian, Dr Sheldon Magder, Dr Jason Shahin, Dr Salman Qureshi (Co-Investigators); Josie Campisi.

Montreal General Hospital, Montreal, QC

Dr Kosar Khwaja (Lead), Dr Dan Deckelbaum, Dr Jeremy Grushka, Dr Ash Gursahaney, Dr David Hornstein, Dr Dev Jayaraman, Dr Tarek Razek, Dr Robert Salasidis, Dr Patrizia Zanelli (Co-Investigators); Norine Alam, Laura Garcia, Kosar Khwaja.

Center Hospital University Montreal (CHUM) – Notre Dame, Montreal, QC

Dr Martin Girard (Lead); Dr Pierre Aslanian (Site Co-Investigator); Fatna Benettaib, Dounia Boumahni, Casey Bourdeau Caporuscio, Ali Ghamraoui, Martine Lebrasseur, Maria Madrid Trinidad, Nicole Poitras, Romain Rigal.

Centre Hospital University Montreal (CHUM) – Saint Luc, Montreal, QC

Dr Martin Girard (Lead); Dr Pierre Aslanian (Co-Investigator); Fatna Benettaib, Casey Bourdeau Caporuscio, Dounia Boumahni, Ali Ghamraoui, Martine Lebrasseur, Maria Madrid Trinidad, Nicole Poitras, Romain Rigal.

Centre Hospital University Montreal (CHUM) - Hotel Dieu, Montreal, QC

Dr Martin Girard (Lead); Dr Pierre Aslanian (Site Co-Investigator); Fatna Benettaib, Dounia Boumahni, Ali Ghamraoui, Martine Lebrasseur, Lancelot Legendre Courville, Nicole Poitras, Romain Rigal, Maya Salame.

Royal Alexandra, Edmonton, AB

Dr Jim Kutsiogiannis (Lead), Dr Raiyan Chowdhury, Dr Jon Davidow, Dr Curt Johnston, Dr Richard Johnston, Dr Kim Macala, Dr Sam Marcushamer, Dr Darren Markland, Dr Doug Matheson, Dr Damian Paton-Gay, Dr David Zygun (Co-Investigators); Nadine Grant, Tayne Hewer, Pat Thompson.

Nova Scotia Health Authority QEII, Halifax, NS

Dr Osama Loubani (Lead), Dr Robert Green, Dr Rick Hall, (Co-Investigators); Diana Gillis, Lisa Julien, Laura Lee Magennis, Tamara Mitterer.

Kingston General, Kingston, ON

Dr John Muscedere (Lead), Dr Gord Boyd, Dr David Maslove (Co-Investigators); Tracy Boyd, Susan Fleury, Ilinca Georgescu, Miranda Hunt, Danielle Muscedere, Nicole O'Callaghan.

St Boniface Hospital and Health Science Centre, Winnipeg, MB

Dr Ryan Zarychanski (Lead), Dr Marcus Blouw, Dr Kendiss Olafson, Dr Bojan Paunovic, Dr Heather Smith (Co-Investigators); Oliver Gutieror; Justin Lys, Nicole Marten, Sherri Lynn Wingfield.

London Health Sciences Centre- Victoria Hospital, London, ON

Dr Ian Ball (Lead); Tracey Bentall, Eileen Campbell, Susie Imerovski, Athena Ovsenek, Rebecca Rondinelli.

Brantford General Hospital, Brantford, ON

Dr Brenda Reeve (Lead), Dr Yousery Koubaesh (Co-Investigator); Karen Bento, Sharlene Davies, Megan Davis, Will Dechert, Krista Gallo, Barbara Longo, Courtney Mullen, Terry Norman, Elysia Skrzypek, Laurenne Wierenga.

Health Sciences Centre, Winnipeg, MB

Dr Ryan Zarychanski (Lead), Dr Bojan Paunovic (Co-Investigator); Justin Lys, Nicole Marten.

St Joseph's Hospital, Toronto, ON

Dr Rob Cirone (Lead), Dr Jennie Johnstone (Co-Investigator); Kanthi Kavikondala, Axelle Pellerin, Laura Tomat.

Royal Columbia Hospital, New Westminster, BC

Dr Steve Reynolds (Lead); Suzette Willems.

William Osler Hospital, McKenzie Health, Brampton Civic Hospital Brampton, ON

Dr Sebastien Trop (Lead), Dr Alexandra Binnie, Dr Ronald Heslegrave, (Co-Investigator); Kim Sharman, Zaynab Panchbhaya.

London Health Sciences Centre – University Hospital, London, ON

Dr Dave Nagpal (Lead), Dr Ian Ball (Co-Investigators); Tracey Bentall, Eileen Campbell, Susie Imerovski, Rebecca Rondinelli, Jessica Sturt-Smith.

University of Alberta, Edmonton, AB

Dr Wendy Sligl (Lead), Dr Sean Bagshaw (Co-Investigator); Nadia Baig, Lorena McCoshen.

Foothills Hospital, University of Calgary, Calgary, AB

Dr Tom Stelfox (Lead), Dr Luc Berthiaume, Dr Philippe Couillard, Dr Jonathan Gaudet, Dr Ken Parhar (Co-Investigators); Joshua Booth, Cassidy Codan, Stacy Ruddell.

Peter Lougheed, University of Calgary, Calgary, AB

Dr Dan Niven (Lead), Dr Luc Berthiaume, Dr Philippe Couillard, Dr Chip Doing, Dr Jonathan Gaudet, Dr Ken Parhar (Co-Investigators); Joshua Booth, Cassidy Codan, Gina Fleming, Katie Ross, Stacy Ruddell.

Niagara Health, St Catharines, ON

Dr Jennifer Ly Tsang (Lead), Dr Erick Duan (Co-Investigator); Mercedes Carmargo, Gina Dimitra, Gina Fleming, Bev Hoekstra, Katie Ross, Moises Vasquez.

Maisonneuve Rosemont, Montreal, QC

Dr Francois Marquis (Lead); Pauline Dul, Johanne Harvey, Danae Tassy, Veronique Tran.

Grand River Hospital, Kitchener, ON

Dr Paul Hosek (Lead), Dr Bill Plaxton (Co-Investigator); Catherine Armstrong, Rhonda Barber, Janelle Ellis, Melissa Gabnouri, Emilie Gordon, Rebecca Haegens, Brooklynn Hillis, Rebecca Jesso, Jenn McLaren, Elliot McMillan, Mariska Pelkmans, Matthew Rekman, Sylvia Sinkovitis, Michelle White.

Sunnybrook Health Sciences Centre, Toronto, ON

Dr Neill Adhikari (Lead), Dr Andre Amaral, Dr Andre Carlos, Dr Brian Cuthbertson, Dr Rob Fowler, Dr Damon Scales (Co-Investigators); Navjot Kaur, Nicole Marinoff, Adic Perez, Jane Wang.

Joseph Brant Hospital, Burlington, ON

Dr Paul Lysecki (Lead); Tracy Campbell.

CHU de Quebec - Universite Laval, Hotel Dieu, Quebec, QC

Dr Francois Lauzier (Lead), Dr Maude St-Onge, Dr Alexis Tourgeon, (Co-Investigators); Anick Boivin, Charles Delisle-Thibeault, Marie-Claude Tremblay.

Institut Universitaire de Cardiologie et de Pneumologie de Quebec, Quebec, QC

Dr Francois Lellouche (Lead), Dr Ying Tung Sia, Dr Mathieu Simon (Co-Investigators); Pierre-Alexandre Bouchard, Patricia Lizotte.

l'Hotel-Dieu de Levis, Levis, QC

Dr Patrick Archambault (Lead), Dr Jean-Francois Bellemare, Dr Simon Bordeleau, Dr Christine Drouin, Dr Benloit Duhaime, Dr Ann Laberge, Dr Philippe Lachance (Co-Investigators); Melanie Constantin, Estel Deblois, Maude Dionne, Lise Lavoie, Isabelle Michel, Alexandre Pepin, Sandrine Poulin.

United States

Mayo Clinic, Rochester, MN

Dr Rodrigo Cartin-Ceba (Lead), Dr Richard Oeckler (Co-Investigator); Brenda Anderson, Lavonne Liedl, Laurie Meade, Sueanne Weist.

St John's Mercy Medical Center, St Louis, MO

Dr Robert Taylor (Lead); Margaret Cytron, Kim Fowler, Katie Krause, Jackie O'Brien, Marianne Tow.

Saudi Arabia

King Abdulaziz Medical Center, Riyadh, Saudi Arabia

Dr Yaseen Arabi (Lead), Dr Abdulaziz Aldawood, Dr Shmeylan Al Harbi, Dr Haytham Tlayjah, (Co-Investigators), Dr alaaEldien Ghanem, Dr Ahmad Hassanein (ICU Physicians); Hala Abdullah, Nadiah Alanizy, Njoud Al Bogami, Ahmad Deeb, Felwa Bin Humaid.

References

- 1. Oczkowski, S.J.W., et al., *The Use of Bowel Protocols in Critically III Adult Patients: A Systematic Review and Meta-Analysis.* Crit Care Med, 2017. **45**(7): p. e718-e726.
- 2. Schiller, L.R., et al., *Gastro 2013 APDW/WCOG Shanghai working party report: chronic diarrhea: definition, classification, diagnosis.* J Gastroenterol Hepatol, 2014. **29**(1): p. 6-25.
- 3. Tirlapur, N., et al., *Diarrhoea in the critically ill is common, associated with poor outcome, and rarely due to Clostridium difficile.* Sci Rep, 2016. **6**: p. 24691.
- 4. Dionne, J.C., et al., *Tu1068 Diarrhea in the Intensive Care Unit: Epidemiology* (*The DICE Study*). Gastroenterology, 2016. **150**(4).
- 5. Johnstone, J., et al., *Probiotics: Prevention of Severe Pneumonia and Endotracheal Colonization Trial-PROSPECT: protocol for a feasibility randomized pilot trial.* Pilot Feasibility Stud, 2015. **1**: p. 19.
- 6. Elo, S. and H. Kyngas, *The qualitative content analysis process.* J Adv Nurs, 2008. **62**(1): p. 107-15.
- 7. Mearin, F., et al., Bowel Disorders. Gastroenterology, 2016.
- 8. Dionne, J.C., Mbuagbaw, L., Moayyedi, P., Armstrong, D., Thabane, L., Alhazzani, W., Jaeschke, J., Cook, D.J., Validation of the Bristol Stool Chart and Bliss Classification System in the Intensive Care Unit (ICU). in Digestive Diseases Week. 2019: San Diego, California
- 9. Begona Guardiola, J.A.L.-P., Jordi Ibanez, Joan M. Raurich, *Prophylaxis Versus Treatment Use of Laxative for Paralysis of Lower Gastrointestinal Tract in Critically III Patients.* Journal of Clinical Gastroenterology, 2016. **50**(2): p. e13e18.
- 10. Masri, Y., J. Abubaker, and R. Ahmed, *Prophylactic use of laxative for constipation in critically ill patients.* Ann Thorac Med, 2010. **5**(4): p. 228-31.
- 11. van der Spoel, J.I., et al., *Laxation of critically ill patients with lactulose or polyethylene glycol: a two-center randomized, double-blind, placebo-controlled trial.* Crit Care Med, 2007. **35**(12): p. 2726-31.
- 12. Zvonicek, V., Sevick, P., Votava, M., Ondrovcik, P., Zabranska, S., Sas, I., Kraus, R., Sramek, V., *The effect of lactulose in prevention of ventilator associated pneumonia (VAP).* Critical Care, 2001. **12**: p. p39.
- 13. Katelaris, P., et al., Comparison of the effectiveness of polyethylene glycol with and without electrolytes in constipation: a systematic review and network metaanalysis. BMC Gastroenterol, 2016. **16**: p. 42.

Hospital	Type of ICU	Community or Academic	Numb er of beds	Bowel Protoco I
Canada				
St. Joseph's Healthcare Hamilton, Hamilton, ON	Medical, Surgical	Academic	21	Yes
Hamilton General Hospital, Hamilton, ON	Medical, Surgical, Trauma, Neurosurgery	Academic	31	Yes
Juravinski Hospital, Hamilton, ON	Medical, Surgical	Academic	23	Yes
St. Michael's Hospital, Toronto, ON	Medical, Surgical, Trauma, Neurosurgery	Academic	24	Yes
Mount Sinai Hospital, Toronto, ON	Medical, Surgical	Academic	16	No
The Ottawa Hospital, Civic Campus Ottawa, ON	Medical, Surgical, Trauma, Neurosurgery	Academic	26	Yes
The Ottawa Hospital, General Campus, Ottawa, ON	Medical, Surgical, Trauma	Academic	25	Yes
University Health Network, Toronto Western Hospital, Toronto, ON	Medical, Surgical, Trauma, Neurosurgery	Academic	24	No
University Health Network, Toronto General Hospital, Toronto, ON	Medical, Surgical	Academic	28	Yes
Kingston General Hospital, Kingston, ON	Medical, Surgical, Trauma, Neurosurgery	Academic	33	Yes
London Health Science Centre, Victoria Hospital, London, ON	Medical, Surgical, Trauma, Neurosurgery	Academic	29	Yes
London Health Science Centre, University Hospital, London, ON	Medical, Surgical	Academic	19	Yes
Brantford General Hospital, Brantford,	Medical, Surgical	Community	15	Yes

ON

Supplement-Table 1: PROSPECT Centers With and Without Bowel Protocols

Hospital	Type of ICU	Community or Academic	Numb er of beds	Bowel Protoco I
St. Joseph's Health Center, Toronto, ON	Medical, Surgical	Community	20	Yes
William Osler, Brampton Civic Hospital, Brampton, ON	Medical, Surgical	Community	44	Yes
Niagara Health, St Catharine's Hospital, St. Catharine's, ON	Medical, Surgical	Community	14	Yes
Grand River Hospital, Kitchener, ON	Medical, Surgical	Community	20	Yes
Sunnybrook Hospital, Toronto, ON	Medical, Surgical, Trauma, Neurosurgery	Academic	34	Yes
Joseph Brant Hospital, Burlington, ON	Medical, Surgical	Community	20	Yes
Hôpital de l'Enfant-Jésus, Québec City, QC	Medical, Surgical, Trauma, Neurosurgery	Academic	29	Yes
Hôpital du Sacré-Cœur de Montréal, QC	Medical, Surgical, Trauma, Neurosurgery	Academic	36	Yes
Centre Hospitalier Universitaire de Sherbrooke (CHUS), Sherbrooke, QC	Medical, Surgical, Trauma, Neurosurgery	Academic	44	No
Hôpital Royal Victoria, Montreal, QC	Medical, Surgical	Academic	30	No
Montreal General Hospital, Montreal, QC	Medical, Surgical, Trauma	Academic	22	No
Hôpital Notre Dame, Montréal, QC	Medical, Surgical, Neurosurgery	Academic	16	No
Hôpital Saint-Luc, Montreal, QC	Medical, Surgical	Academic	14	No
The Centre hospitalier de l'Université de Montréal (N-CHUM), Montreal, QC	Medical, Surgical, Neurosurgery	Academic	60	No
Hôpital Maisonneuve-Rosemont, Montréal, QC	Medical, Surgical	Academic	16	Yes

Hospital	Type of ICU	Type of ICU Community or Academic		Bowel Protoco I
Institut universitaire de cardiologie et de pneumologie de Québec - Université Laval, Québec City, QC	Medical, Surgical	Academic	35	Yes
Hôtel-Dieu de Lévis Hôpital, Québec City, QC	Medical, Surgical	Community	18	Yes
St Paul's Hospital, Vancouver, BC	Medical, Surgical	Academic	19	Yes
Vancouver General Hospital, Vancouver, BC	Medical, Surgical, Trauma, Neurosurgery	Academic	34	Yes
Vancouver Island Health Authority, Victoria, BC	Medical, Surgical, Trauma	Academic	20	Yes
Royal Columbia Hospital, New Westminster, BC	Medical, Surgical	Academic	16	Yes
University of Alberta Hospital, Edmonton, AB	Medical, Surgical, Trauma	Academic	32	No
Foothills Medical Center, Calgary, AB	Medical, Surgical, Trauma	Academic	28	Yes
Peter Lougheed Hospital, Calgary, AB	Medical, Surgical	Academic	22	Yes
Royal Alexandra Hospital, Edmonton, AB	Medical, Surgical, Trauma	Academic	27	No
St. Boniface Hospital, Winnipeg, MB	Medical, Surgical	Academic	13	Yes
Health Science Centre Winnipeg, Winnipeg, MB	Medical, Surgical, Trauma, Neurosurgery	Academic	27	Yes
QEII Health Science Center, Halifax, NS	Medical, Surgical, Trauma, Neurosurgery	Academic	21	Yes
United States of America				
Mayo Clinic, Rochester, MN	Medical, Surgical	Academic	24	No
Mercy Medical Center, St. Louis, MO	Medical, Surgical, Trauma	Academic	54	Yes
Saudi Arabia				

Hospital	Type of ICU	Community or Academic	Numb er of beds	Bowel Protoco I
King Abdulaziz Medical City, Riyadh, SA	Medical, Surgical, Trauma	Academic	60	Yes

Legend for Table 1 (supplement): Listed here are the characteristics of participating centers, including type of ICU, number of ICU beds, type of ICU, setting and presence or absence of bowel protocols.

Chapter 5 Clostridioides Difficile Infection in Mechanically Ventilated Critically III Patients: A Nested Cohort Study

Clostridioides difficle infection in Mechanically Ventilated Critically III Patients: A Nested Cohort Study

JC Dionne, E Duan, J Johnstone, D Heels-Ansdell, F Lauzier, Y Arabi, J Marshall, W Sligl, B Rochwerg, N Adhikari, D Williamson, D Niven, S Reynolds, P Dodek, N Zytaruk, D Cook For the PROSPECT Investigators and the Canadian Critical Care Trials Group

Abstract

Introduction: *Clostridioides difficile* infection (CDI) is associated with significant morbidity and mortality in the intensive care unit (ICU). Our objective was to determine the frequency, timing, severity, predictors, impact and treatment of CDI among critically ill patients enrolled in a randomized trial.

Methods: We performed a nested cohort study of CDI within an international randomized trial comparing the probiotic *Lactobacillus rhamnosus GG* to placebo in mechanically ventilated patients to evaluate pneumonia and other ICU-acquired infections including CDI (PROSPECT, NCT02462590). To identify patients with possible CDI, we collected daily data regarding diarrhea during the ICU admission and all *C. difficile* testing during hospital admission. We conducted blinded duplicate adjudication of all patients with possible CDI, using standardized definitions to diagnose CDI, determine timing (pre-ICU, in ICU, post-ICU), rate severity (comparing the American College of Gastroenterology, Society for Healthcare Epidemiology of America (SHEA)-Infectious Diseases Society of America (IDSA), and the European Society of Clinical Microbiology and Infections Disease criteria), analyze risk factors, describe treatments, and assess hospital mortality and length of stay.

Results: Of 2650 mechanically ventilated patients enrolled in PROSPECT, 86 were diagnosed with CDI over 90,833 hospital-days, generating a rate of 9.5 per 10,000 hospital-days and a CDI prevalence of 3.2% (95% confidence interval [CI] 2.6% to 4.0%). Nine (0.3%) patients were identified with CDI pre-ICU, and 57 (2.2%) developed CDI in the ICU, with a rate of 1.3 per 1000 ICU-days. An additional 20 (0.8%) patients developed CDI post-ICU. Hospital-based CDI relapse or reoccurrence occurred in 8 of 86 (9.3%) patients. The majority of infections were mild to moderate in severity consistently scored across all three severity systems; 58 (68.2%) having mild to moderate disease based on SHEA, 65 (76.5%) based on ACG and 71 (83.5%) based on the ESCMID. A minority of patients had severe CDI: 18 (21.2%, SHEA), 8 (9.4%, ACG), and 14 (16.5%, ESCMID). There were only 9 (10.6%) and 12 (14.1%) patients with severe, complicated CDI based on SHEA and ACG scores, respectively agreement among severity scores was fair to moderate (range of 2-way kappas 0.47 to 0.64). Complications of CDI included septic shock (23, 26.7%), end-organ failure (14, 16.3%), toxic megacolon (1, 1.2%) requiring a colectomy. Among patients with and without CDI, crude hospital mortality was 27.3% versus 28.0%, respectively; however, CDI was not independently associated with hospital mortality (HR 0.97; 95% CI 0.58-1.63, p=0.916) after adjusting for the APACHE II score. Patients with CDI stayed in hospital a mean of 42 days (IQR 22,77) compared to those without who stayed 22 days (IQR 12,40), p<0.001). Adjusting for APACHE II score, randomized group, medical/surgical/trauma admitting diagnosis, and center in linear regression, this difference remained significant (p<0.001). Most patients received one or more of metronidazole (66, 76.7%), primarily

administered orally (48, 55.8%), and oral vancomycin (56, 65.1%). No patients received fidaxomicin. Fecal transplant was performed in 3 (3.5%) patients. No independent risk factor for CDI was identified.

Conclusion: Among mechanically ventilated patients enrolled in a probiotics trial, CDI was relatively uncommon and the severity was mild to moderate. We found no independent risk factors for CDI. CDI was not associated with an increased risk of hospital mortality but was associated with a significantly longer length of stay in hospital.

Introduction

Clostridioides difficile infection (CDI) is associated with significant morbidity and mortality [1] in hospitalized patients, with 462,100 cases annually in the United States. [2]. Community-acquired recurrence rates may be as high as 31,300 cases annually, and healthcare-associated rates are estimated at 38,500 cases annually[2]. The burden of CDI on the healthcare system is significant, costing approximately \$8,911 USD to \$30,049 USD per hospitalized patient [3] with an estimated annual cost of 1.1 to 3.2 billion USD per year [4]. A Canadian provincial study in Alberta found that the mean attributable cost per case was \$18,386 (CAD 2018; USD \$14,190; 95% CI, \$14,312-\$22,460; USD \$11,046-\$17,334) [5].

With its impact on patients and the healthcare system, CDI has been a focus of health research in areas such as epidemiology, therapeutics, and infection control. However, most CDI research consists of retrospective cohort, cross-sectional studies, and few studies have specifically examined the impact of CDI in the Intensive Care Unit (ICU). One retrospective matched cohort demonstrated that ICU-acquired CDI was

associated with increased length of ICU and hospital stay but not increased mortality [1]. Conflicting results have been generated in prior studies about CDI-associated mortality rates of 5.5% to 6.9% in hospitalized patients[4, 6-11].

The objectives of this study were to 1) analyze the incidence and prevalence of CDI in the ICU, 2) describe the timing of CDI (pre-ICU, in ICU, and post-ICU), 3) assess the severity of CDI using 3 scoring systems and understand the agreement among scores, 4) analyze CDI risk factors, 5) examine the hospital mortality and length of stay of patients with CDI compared to without CDI and 6) document CDI treatments used.

Methods

We performed a nested cohort study within a randomized controlled trial in 2,650 mechanically ventilated patients compared the probiotic *Lactobacillus rhamnosus* GG to placebo on the primary outcome of ventilator-associated pneumonia (PROSPECT, Probiotics to prevent Severe Pneumonia and Endotracheal Colonization Trial, NCT02462590)[12]. Other outcomes were other ICU-acquired infections including CDI, antimicrobial use, and diarrhea. Enrolled patients were at least 18 years old and expected to be mechanically ventilated for at least 72 hours. Patients were excluded if immunocompromized (HIV with a CD4 count less than 200 cells/µL, chronic immunosuppressive medications, chemotherapy in the last 3 months, prior organ or hematological transplant, or absolute neutrophil count less than 500 cells/µL); if they carried increased risk of endovascular infection; had severe acute pancreatitis; a percutaneous enteral feeding tube, were unable to receive enteral medication; had plans for palliation; and those previously enrolled in this trial or a related trial.

Daily all bowel movements were counted and characterized using the Bristol Stool Chart[13, 14]. For each enrolled patient, research coordinators recorded baseline data and daily clinical data (e.g., *Clostridioides difficile* testing, culture results, infections, treatments, diarrhea defined as Bristol Type 6 or 7), length of stay and mortality. For all patients who had a suspected or confirmed CDI, we collected additional information from the medical chart, and a CDI case report form (Appendix) was sent to the PROSPECT Methods Center. In this study, we used standardized definitions to define CDI as follows: laboratory confirmation of *C. difficile* together with three or more episodes of diarrhea within a 24-hour period[15].

Two independent adjudicators conducted blinded adjudication of patients with suspected CDI to categorize the timing of infection (pre-ICU, in ICU, post-ICU). We defined pre-ICU CDI if the diagnosis was made <72 hours of ICU admission or earlier. CDI was defined as being acquired in ICU if it was diagnosed 72 hours after ICU admission, and up to 72 hours after ICU discharge. We defined post ICU CDI if the diagnosis was made 72 hours after ICU discharge[15].

Adjudication was also used to assess whether it was a new infection, relapse, or recurrence, and rate the severity of infection using three severity scales (American College of Gastroenterology[16], Society for Healthcare Epidemiology of America (SHEA)-Infectious Diseases Society of America (IDSA) [17], and the European Society of Clinical Microbiology and Infections Disease criteria[18]) (Box 1). We resolved any disagreement using a third adjudicator.

Analysis

We calculated measures of central tendency and dispersion. To identify new cases of CDI acquired only in ICU, we calculated incidence as the total number of expressed as both a percentage of the entire cohort and as a rate per 1000 ICU days. We defined period prevalence during the entire study by the number of patients with CDI (pre-ICU, ICU, post-ICU) expressed as a percentage and as a rate per 10,000 hospital-days.

We assessed the agreement between the three scoring systems using kappa statistics in pairwise comparison.

We examined risk factors for CDI identified in the ICU (excluding patients who had a) CDI diagnosed pre-ICU, b) CDI diagnosed in ICU prior to their third day in the PROSPECT study, or c) CDI diagnosed after discharge from ICU) using Cox regression, considering baseline illness severity as measured by APACHE II score as well as 3 time dependent exposures within the previous 3 days (vasopressors, antibiotics and stress ulcer prophylaxis with either histamine-2-receptor antagonists (H2RA) or proton pump inhibitors (PPI)), reporting adjusted hazard ratios (HR) with corresponding 95% confidence intervals (CI) and p-values.

We examined the association between CDI identified after ICU admission (excluding patients who had CDI diagnosed pre-ICU) and hospital mortality by crude comparison, and using Cox regression, unadjusted and adjusted for the APACHE II score, reporting hazard ratios (HR) with corresponding 95% confidence intervals (CI) and p-values. We conducted a sensitivity analysis excluding prevalent CDI in the first 2 days.

We examined the association between CDI identified in the ICU (excluding patients who had CDI diagnosed pre-ICU) and the logarithm of hospital length of stay by crude comparison, reported using a t-test on the log-transformed variable. We also used linear regression, reporting results unadjusted and adjusted for APACHE II score, randomized group, medical/surgical trauma admitting diagnosis and center, showing median differences with corresponding interquartile ranges and p-values. We conducted a sensitivity analysis on survivors only.

We report treatment received as number and percentages. We conducted a secondary analyses of risk factors for CDI including APACHE II Score, histamine-2 receptor agonists (H2RA or proton pump inhibitors, antibiotic exposure, or vasopressor use, by using a Cox proportional hazards regression. All analyses were performed using SAS (version 9.4, 2013)[19].

Results

Among 2650 patients enrolled in this study, characteristics are shown in Table 1, including number of BMs and diarrhea, and the timing of CDI diagnosis. Overall, 86 patients were diagnosed with CDI at any point during the study, which comprised 90,883 hospital-days. This corresponds to a rate of 9.5 per 10,000 hospital-days and a study prevalence of 3.2% (95% CI 2.6% to 4.0%). A total of 9 (0.3%) patients developed CDI prior to ICU admission and 57 (2.2%) developed CDI during 43,927 ICU-days, corresponding to a rate of 1.3 per 1000 ICU-days. In contrast, 20 (0.8%), patients were found to have CDI following ICU discharge. A relapse or recurrence of CDI was found in 8 of 86 patients (9.3%) patients. Complications of CDI included septic shock (23,

26.7%), end organ failure (14, 16.3%), and toxic megacolon that required colectomy (1, 1.2%).

The most common diagnostic test used to help identify CDI was nucleic acid amplification test (NAAT)-based assays; polymerase chain reaction (PCR) (50, 58.1%) or loop mediated isothermal amplifications (LAMP) (23, 26.7%), with one patient having 2 diagnostic tests. Only a minority used exclusively enzyme linked immunosorbent assay (ELISA) testing (12, 14.0%) or other tests (2, 2.3%).

The majority of infections were mild to moderate in severity consistently scored across all three severity systems (Table 2), with 58 (68.2%) having mild to moderate disease based on SHEA, 65 (76.5%) based on ACG and 71 (83.5%) based on the ESCMID. A minority of patients had severe CDI: 18 (21.2%, SHEA), 8 (9.4%, ACG), and 14 (16.5%, ESCMID). There were only 9 (10.6%) and 12 (14.1%) patients with severe, complicated CDI based on SHEA and ACG scores, respectively.

Agreement between the ACG and SHEA scores (Table 3) (Kappa 0.63, 95% CI 0.46 to 0.80) and the ESCMID compared to ACG (Kappa 0.64, 95% CI 0.43 to 0.84) was moderate. The agreement between the ESCMID and the SHEA score was fair (Kappa 0.47, 95% CI 0.27 to 0.67). The hospital length of stay and hospital mortality of patients with CDI across all severity scores is displayed in Table 4.

Associations with CDI documented in the Cox regression included APACHE II score (HR 0.83, 95% CI 0.55-1.25), p=0.372), receipt of stress ulcer prophylaxis (HR 1.28 (0.39, 4.21) p=0.686), antibiotics (1.24 (0.50, 3.06), p=0.647) and vasopressors or inotropes (HR 1.09 (0.57, 2.11), p=0.790) (Table 5). None of these were statistically significant[11].

Among patients with and without CDI, crude hospital mortality was 27.3% versus 28.0%, respectively; however, CDI was not independently associated with hospital mortality (HR 0.97; 95% CI 0.58, 1.63, p=0.916) after adjusting for the APACHE II score (Table 6). The sensitivity analysis excluding prevalent CDI in the first 2 days yielded a HR of 1.15 (95% CI 0.63, 2.11).

Patients with CDI stayed in hospital a mean of 42 days (IQR 22,77) compared to those without who stayed 22 days (IQR 12,40), p<0.001). Adjusting for APACHE II score, randomized group, medical/surgical/trauma admitting diagnosis, and center in linear regression, this difference remained significant (p<0.001). Results were similar in a sensitivity analysis including only survivors (p<0.001). (Table 7).

Treatments received included metronidazole (66, 76.7%), most often orally (48, 55.8%), oral vancomycin (56, 65.1%). No patients received fidaxomicin. Fecal transplant was performed in 3 (3.5%) patients.

Discussion

This secondary analysis of critically ill patients enrolled in the PROSPECT randomized trial demonstrated that most critically ill patients develop this infection while they are in the ICU, but some infections are comorbidities before critical illness, and some infections are only evident as in the recovery phase after ICU discharge. CDI acquired in ICU was mild to moderate in severity according to 3 pre-existing criteria, with the most common complication being septic shock. The most common treatment was metronidazole, reflective of guideline recommendations at the time of trial conduct [17]. The agreement between the severity ratings was fair to moderate. The incidence of CDI we documented of 2.2% is in keeping with other studies reporting CDI incidence in critically ill patients [20]. Our study showed a 3.2% prevalence of CDI in this cohort that was higher than some other studies[20] however in one retrospective cohort study, an incidence of ICU-acquired CDI has been reported as high as 3.2 cases per 1,000 patient days[21]. Karanika and colleagues [20] conducted a systematic review of 80,835 ICU patients and found a CDI prevalence of 2%; [20]. In comparison, another systematic review of CDI in hospitalized patients showed that the incidence across 13 studies ranged from 2.8 to 15.8 cases per 10,000 patient days [22]. Capturing the true incidence of CDI is challenging in clinical studies due to variability in definitions and testing strategies.

Other studies have sought to determine the incidence, prevalence, risk factors and outcomes for CDI in the hospital [8, 9, 11] and in the ICU [1, 21] prolonged hospitalization[17], chemotherapy, enteral nutrition[17, 23] and proton pump inhibitors (PPI). However, data are conflicting as recent data do not show that exposure to PPIs increases the risk of CDI, and the prior signal may be as a result of confounding[17, 24, 25]. These CDI risk factors are particularly relevant to critically ill patients, who are at increased risk of many ICU-acquired infections, including CDI.

We documented fair agreement across all CDI severity scoring systems. Few studies have compared the severity of CDI across scoring systems. One study examined the differences in severity scores in classifying disease severity in patients with CDI using the IDSA and ACG scoring systems in their ability to predict colectomy and mortality[26]. A retrospective cohort study of 894 patients with CDI found that the ACG score labelled more patients as severe and complicated compared to the IDSA

definition [26], possibly because the former includes many non-specific illness severity parameters[16] Although the IDSA [17] severity metric includes a WBC count, ileus or megacolon criterion, it also includes acute kidney injury which is non-specific and common during critical illness. Ensuring consistent nomenclature in practice will improve interprofessional recognition of CDI severity. Given the complexities of critical illness, an ICU-specific metric for CDI severity in this population may be warranted to better risk stratify these patients.

We found that patients with CDI had a similar hospital mortality compared to other patients. Prior research yielded conflicting results regarding the association of CDI with mortality. The hospital mortality observed in this critically ill population for those with CDI was 27.3%. One systematic review [20] of CDI in ICU patients showed a hospital mortality of 32%, compared to 24% in patients without CDI [20]. Manthey and colleagues conducted a retrospective cohort study of 144 patients with CDI found that 28 day mortality in critically ill patients with CDI to be 27.3% compared to 9.0% in non-CDI patients[[27]. The difference seen in our study may be attributed to the larger sample size of our study, prospective design, and potentially sicker patients at baseline given trial enrollment criteria including dependence on mechanical ventilation[12, 28] .

Other studies have examined the impact of CDI on length of stay compared to patients without CDI in critically ill patients. In one review [20], critically ill patients with CDI had longer length of stay in ICU and hospital compared to those whose without CDI [20], underscoring the economic impact for the health care system. We also found a significant association between CDI and hospital length of stay which cannot be interpreted as causation in this observational study. This could mean that CDI

increases the hospital length of stay, but could also reflect how being in hospital increases the risk of CDI. Another explanation is that the risk factors for a longer hospital length of stay also increase the risk of developing CDI. From an economic perspective, patients with CDI require isolation in single rooms and consumables for infection control purposes; the increased length of stay for patients with CDI is a further indication of associated healthcare costs.

The treatment for CDI reflects guideline recommendations for the treatment of CDI from 2013 to 2019 during the years of patient enrolment. Antimicrobial guidelines for first CDI occurrence in 2010 recommended metronidazole as first line for mild to moderate CDI [17], then oral vancomycin in 2017 [29], and in 2021, fidaxomicin [2].

The strengths of our study include the large sample size, prospective enrolment of critically ill patients, documentation of the number and characteristics of all bowel movements, rigorous methodology with blinded adjudication of the CDI microbiologic data and outcomes, and application of three validated scores to assess CDI severity. Patients had heterogeneous diagnoses and were enrolled in 3 countries, thereby enhancing the generalizability of our findings. However, this study has limitations. We do not have data on the prevalence of NAP1/B1/027 hypervirulent strains. Findings in this study for CDI antimicrobial treatment reflect these changes in practice over time. Although characteristics of patients with and without CDI are demonstrated, the relatively low CDI event rate means that other unmeasured confounders could have affected the multivariate analyses of CDI risk factors and analysis of attributable mortality and length of stay. This cohort was determined by participants in a randomized trial of probiotics, focused on patients expected to be mechanically

ventilated for at least 72 hours; therefore, our results do not represent critically ill patients with a short stay in the ICU.

Conclusion

We found that CDI was relatively uncommon in this population of mechanically ventilated critically ill patients. When acquired in the ICU, CDI severity was mild to moderate and most patients received metronidazole, reflecting guideline recommendations at the time. No independent risk factors were identified. Future research of CDI in the critically ill should focus on developing an ICU-specific metric to characterize CDI severity.

Acknowledgements

We thank the patients and families contributing to this study and the bedside nurses and physicians who supported this work. We are grateful for the work of the Research Coordinators and Investigators of the PROSPECT Trial and the Canadian Critical Care Trials Group. We appreciate the PROSPECT Methods Center staff for their expertise and data management, including Lois Saunders, Mary Copland, BA, Shelley Anderson-White, BA, Alyson Takaoka, MSc, France Clarke, RRT, Lori Hand, RRT, Megan Davis, BSc, Melissa Shears, MD and Kristine Wachmann, RN.

References

- 1. Dodek PM, Norena M, Ayas NT, Romney M, Wong H, (2013) Length of stay and mortality due to Clostridium difficile infection acquired in the intensive care unit. Journal of Critical Care 28: 335-340
- Johnson S, Lavergne V, Skinner AM, Gonzales-Luna AJ, Garey KW, Kelly CP, Wilcox MH, (2021) Clinical Practice Guideline by the Infectious Diseases Society of America (IDSA) and Society for Healthcare Epidemiology of America (SHEA): 2021 Focused Update Guidelines on Management of Clostridioides difficile Infection in Adults. Clin Infect Dis 73: e1029-e1044
- 3. Gupta A, Ananthakrishnan AN, (2021) Economic burden and cost-effectiveness of therapies for Clostridiodes difficile infection: a narrative review. Therap Adv Gastroenterol 14: 17562848211018654
- 4. Bobo LD, Dubberke ER, Kollef M, (2011) Clostridium difficile in the ICU: the struggle continues. Chest 140: 1643-1653
- Leal JR, Conly J, Weaver R, Wick J, Henderson EA, Manns B, Ronksley P, (2019) Attributable costs and length of stay of hospital-acquired Clostridioides difficile: A population-based matched cohort study in Alberta, Canada. Infect Control Hosp Epidemiol 40: 1135-1143
- 6. Oake N, Taljaard M, van Walraven C, Wilson KM, Roth V, Forester A, (2010) The Effect of Hospital-Acquired Clostridium difficile Infection on In-Hospital Mortality. ARCH INTERN ME 170: 1804-1810
- 7. Forster AJ, Taljaard M, Oake N, Wilson K, Roth V, Van Walraven C, (2012) The effect of hospital-acquired infection with Clostridium difficile on length of stay in hospital. Canadian Medical Association Journal 184: 37-42
- 8. Dubberke ER, Butler AM, Reske KA, Agniel D, Olsen MA, D'Angelo G, McDonald LC, Fraser VJ, (2008) Attributable outcomes of endemic Clostridium difficileassociated disease in nonsurgical patients. Emerg Infect Dis 14: 1031-1038
- Loo V, Poirier L, Miller M, Oughton M, Libman M, Michaud S, Bourgault A, Nguyen T, Frenette C, Kelly M, Vibien A, Brassard P, Fenn S, Dewar K, Hudson T, Horn R, René P, Monczak Y, Dascal A, (2005) A Predominantly Clonal Multi-Institutional Outbreak of Clostridium difficile–Associated Diarrhea with High Morbidity and Mortality. N Engl J Med 353: 2442-2449
- 10. Pepin J, Valiquette L, Gagnon S, Routhier S, Brazeau I, (2007) Outcomes of Clostridium difficile-associated disease treated with metronidazole or vancomycin before and after the emergence of NAP1/027. Am J Gastroenterol 102: 2781-2788
- 11. Zilberberg M, Shorr A, Kollef MH, (2008) Increase in Adult Clostridium difficile– related Hospitalizations and Case-Fatality Rate, United States, 2000–2005. Emerging Infectious Diseases 14: 929-931
- 12. Johnstone J, Meade M, Lauzier F, Marshall J, Duan E, Dionne J, Arabi YM, Heels-Ansdell D, Thabane L, Lamarche D, Surette M, Zytaruk N, Mehta S, Dodek P, McIntyre L, English S, Rochwerg B, Karachi T, Henderson W, Wood G, Ovakim D, Herridge M, Granton J, Wilcox ME, Goffi A, Stelfox HT, Niven D, Muscedere J, Lamontagne F, D'Aragon F, St-Arnaud C, Ball I, Nagpal D, Girard M, Aslanian P, Charbonney E, Williamson D, Sligl W, Friedrich J, Adhikari NK,

Marquis F, Archambault P, Khwaja K, Kristof A, Kutsogiannis J, Zarychanski R, Paunovic B, Reeve B, Lellouche F, Hosek P, Tsang J, Binnie A, Trop S, Loubani O, Hall R, Cirone R, Reynolds S, Lysecki P, Golan E, Cartin-Ceba R, Taylor R, Cook D, Prevention of Severe P, Endotracheal Colonization Trial I, the Canadian Critical Care Trials G, (2021) Effect of Probiotics on Incident Ventilator-Associated Pneumonia in Critically III Patients: A Randomized Clinical Trial. JAMA 326: 1024-1033

- 13. Caroff DA, Edelstein PH, Hamilton K, Pegues DA, Program CDCPE, (2014) The Bristol stool scale and its relationship to Clostridium difficile infection. J Clin Microbiol 52: 3437-3439
- 14. Lewis SJ, Heaton KW, (1997) Stool form scale as a useful guide to intestinal transit time. Scandinavian Journal of Gastroenterology 32: 920-924
- 15. Ontario Agency for Health Protection and Promotion PIDAC (2013) Testing, Surveillance and Management of Clostridium difficile. In: Editor (ed)^(eds) Book Testing, Surveillance and Management of Clostridium difficile., City, pp.
- 16. Surawicz CM, Brandt LJ, Binion DG, Ananthakrishnan AN, Curry SR, Gilligan PH, McFarland LV, Mellow M, Zuckerbraun BS, (2013) Guidelines for diagnosis, treatment, and prevention of Clostridium difficile infections. Am J Gastroenterol 108: 478-498; quiz 499
- Cohen SH, Gerding DN, Johnson S, Kelly CP, Loo VG, McDonald LC, Pepin J, Wilcox MH, Society for Healthcare Epidemiology of A, Infectious Diseases Society of A, (2010) Clinical practice guidelines for Clostridium difficile infection in adults: 2010 update by the society for healthcare epidemiology of America (SHEA) and the infectious diseases society of America (IDSA). Infect Control Hosp Epidemiol 31: 431-455
- Debast SB, Bauer MP, Kuijper EJ, (2014) European Society of Clinical Microbiology and Infectious Diseases: Update of the Treatment Guidance Document for Clostridium difficile Infection. Clinical Microbiology and Infection 20: 1-26
- 19. SAS/ACCESS SII (2013) SAS/ACCESS. In: Editor (ed)^(eds) Book SAS/ACCESS. SAS Institute Inc, City, pp.
- 20. Karanika S, Paudel S, Zervou FN, Grigoras C, Zacharioudakis IM, Mylonakis E, (2016) Prevalence and Clinical Outcomes of Clostridium difficile Infection in the Intensive Care Unit: A Systematic Review and Meta-Analysis. Open Forum Infectious Diseases 3: ofv186
- 21. Lawrence SJ, Puzniak LA, Shadel BN, Gillespie KN, Kollef MH, Mundy LM, (2007) Clostridium difficile in the Intensive Care Unit: Epidemiology, Costs, and Colonization Pressure. Infection Control & Hospital Epidemiology 28: 123-130
- 22. Marra AR, Perencevich EN, Nelson RE, Samore M, Khader K, Chiang HY, Chorazy ML, Herwaldt LA, Diekema DJ, Kuxhausen MF, Blevins A, Ward MA, McDanel JS, Nair R, Balkenende E, Schweizer ML, (2020) Incidence and Outcomes Associated With Clostridium difficile Infections: A Systematic Review and Meta-analysis. JAMA Netw Open 3: e1917597
- 23. Riddle DJ, Dubberke ER, (2009) Clostridium difficile infection in the intensive care unit. Infect Dis Clin North Am 23: 727-743

- 24. Weiss K, Louie T, Miller MA, Mullane K, Crook DW, Gorbach SL, (2015) Effects of proton pump inhibitors and histamine-2 receptor antagonists on response to fidaxomicin or vancomycin in patients with Clostridium difficile-associated diarrhoea. BMJ Open Gastroenterol 2: e000028
- 25. Novack L, Kogan S, Gimpelevich L, Howell M, Borer A, Kelly CP, Leffler DA, Novack V, (2014) Acid suppression therapy does not predispose to Clostridium difficile infection: the case of the potential bias. PLoS One 9: e110790
- 26. Khan A, Elashery A, Kapadia S, Chandra S, (2014) Performance of severity of illness classification for Clostridium difficile infection to predict need-for-colectomy or inpatient death. Journal of Community Hospital Internal Medicine Perspectives 4: 24711
- 27. Manthey CF, Dranova D, Christner M, Drolz A, Kluge S, Lohse AW, Fuhrmann V, (2019) Initial therapy affects duration of diarrhoea in critically ill patients with Clostridioides difficile infection (CDI). Critical Care 23
- 28. Johnstone J, Meade M, Marshall J, Heyland DK, Surette MG, Bowdish DM, Lauzier F, Thebane L, Cook DJ, Investigators P, the Canadian Critical Care Trials G, (2015) Probiotics: Prevention of Severe Pneumonia and Endotracheal Colonization Trial-PROSPECT: protocol for a feasibility randomized pilot trial. Pilot Feasibility Stud 1: 19
- McDonald LC, Gerding DN, Johnson S, Bakken JS, Carroll KC, Coffin SE, Dubberke ER, Garey KW, Gould CV, Kelly C, Loo V, Shaklee Sammons J, Sandora TJ, Wilcox MH, (2018) Clinical Practice Guidelines for Clostridium difficile Infection in Adults and Children: 2017 Update by the Infectious Diseases Society of America (IDSA) and Society for Healthcare Epidemiology of America (SHEA). Clin Infect Dis 66: e1-e48

Box 1: Definitions of CDI Severity

	SHEA[17]	ACG[16]	ESCMID[18]
Mild to Moderate CDI	WBC < 15x109/L AND Creatinine <1.5 x premorbid level	Diarrhea PLUS signs and symptoms not meeting the criteria for sever or complicated	A clinical picture compatible with CDI and microbiological evidence of free toxins and the presence of CDI in stool without reasonable evidence of another cause of diarrhea OR Pseudomembranous colitis as diagnosed
			during endoscopy, after colectomy or autopsy
Severe	WBC > 15x10 ⁹ /L Creatinine > 1.5 X Premorbid level	Serum Albumin <30g/L PLUS one of the following: WBC > 15x10 ⁹ /L Abdominal tenderness	An episode of CDI with one or more specific signs and symptoms of severe colitis or complicated course of disease with significant systemic toxin effects and shock, resulting in the need for ICU admission, colectomy or death
Severe with complications	Hypotension Shock Ileus Megacolon	Any of the following attributable to CDI: Hypotension with vasopressors Fever > 38.5° C Ileus or significant abdominal distention Mental status changes WBC > $35x10^{9}$ /L or $<2x10^{9}$ /I	

Lactate> 2.2 mmol/L End-organ failure (ex. New	
mechanical ventilation or dialysis	

Legend for Box: These are the definitions from the clinical practice guidelines cited above.

Table 1: Patient Characteristics

Characteristics	Patients without	Patients with CDI	Total
	CDI	n=86	n=2650
	n=2564		
Age in years, mean (SD)	59.8 (16.5)	60.7 (15.4)	59.8 (16.5)
APACHE II, mean (SD)	22.0 (7.8)	23.0 (7.1)	22.0 (7.8)
Clinical Frailty Score†			
mean (SD)	3.4 (1.6)	3.9 (1.5)	3.4 (1.6)
≥5. number (%)	450/2116 (21.3)	22/66 (33.3)	472/2182
-, (,			(21.6)
Females, n (%)	1018 (39.7)	45 (52.3)	1063 (40.1)
Type of Patient, number (%)			
Medical	1954 (76.2)	73 (84.9)	2027 (76.5)
Surgical	265 (10.3)	5 (5.8)	270 (10.2)
Trauma	345 (13.5)	8 (9.3)	353 (13.3)
Admitting Diagnosis, number			
(%)	888 (34.6)	29 (33.7)	917 (34.6)
Respiratory	456 (17.8)	13 (15.1)	469 (17.7)
Neurologic	355 (13.8)	9 (10.5)	364 (13.7)
Trauma	307 (12.0)	19 (22.1)	326 (12.3)
Sepsis	243 (9.5)	5 (5.8)	248 (9.4)
Cardiovascular	103 (4.0)	1 (1.2)	104 (3.9)
Gastrointestinal	88 (3.4)	6 (7.0)	94 (3.5)
Metabolic	28 (1.1)	0 (0.0)	28 (1.1)
Renal	11 (0.4)	0 (0.0)	11 (0.4)
Cardiovascular surgery	5 (0.2)	0 (0.0)	5 (0.2)
Hematologic	3 (0.1)	0 (0.0)	3 (0.1)
Orthopedic	2 (0.1)	0 (0.0)	2 (0.1)
Gynecologic	39 (1.5)	4 (4.7)	43 (1.6)
Other Medical	36 (1.4)	0 (0.0)	36 (1.4)
Other Surgical			
Study Day 1			
Inotropes or vasopressors	1567 (61.1)	54 (62.8)	1621 (61.2)
Renal replacement therapy	202 (7.9)	13 (15.1)	215 (8.1)
H2RA or PPI	2320 (90.5)	83 (96.5)	2403 (90.7)
Antibiotics	2113 (82.4)	73 (84.9)	2186 (82.5)
Baseline Antibiotics			
On day of randomization or	2218 (86.5)	75 (87.2)	2293 (86.5)
within 2 days prior			()
On day of randomization	1106 (43.1)	39 (45.3)	1145 (43.2)
AND day prior AND day prior	()	- (/	
to that (used in our subgroup			
analysis)			

Legend for Table 1: In this table we present baseline characteristics of 2,650 critically ill patients. SD=standard deviation. APACHE=Acute Physiology and Chronic Health Evaluation II Score. † n=2182. Pre-Hospital Admission Clinical Frailty Scale Form 3B became mandatory for patients randomized on or after May 2016.
Table 2: Severity of CDI infection

	SHEA/IDSA	ACG	ESCMID
Mild/moderate	58 (68.2)	65 (76.5)	71 (83.5)
Severe	18 (21.2)	8 (9.4)	14 (16.5)
Severe with	9 (10.6)	12 (14.1)	N/A
complications			

Legend for Table 2: Number (%) of the 85 patients with CDI infection. Above reflects the highest severity per patient.

Table 3: Agreement between CDI Severity Scores

			SHEA/IDSA	
		Mild/moderate	Severe	Severe with Complications
	Mild/moderate	55	10	0
100	Severe	1	7	0
ACG	Severe with	2	1	9
	complications			

Kappa = 0.63 (0.46, 0.80)

Weighted kappa = 0.76 (0.60, 0.92)

		SHEA/IDSA		
			Mild/moderate	Severe with or without Complications
ESCMID	Mild/moderate	56		15
ESCIVID	Severe	2		12

Kappa = 0.47 (0.27, 0.67)

		ACG		
		Mild/moderate	Severe with or without Complications	
ESCMID	Mild/moderate	63	8	
ESCIMID	Severe	2	12	

Kappa = 0.64 (0.43, 0.84)

Legend for Table 3: In this series of tables, we present the CDI severity classification according to different severity metrics, and agreement statistics between severity scores.

Table 4: Relationship Between CDI Severity Scores and Hospital Length of Stay and Mortality

CDI Severity Score	Number of	Hospital LOS	Hospital Mortality
	Patients	(%)	(%)
SHEA			
Mild/moderate	58	8 (13.8)	17 (29.3)
Severe	18	4 (22.2)	4 (22.2)
Severe with	9	4 (44.4)	5 (55.6)
complications			
ACG			
Mild/moderate	65	9 (13.8)	17 (26.2)
Severe	8	3 (37.5)	3 (37.5)
Severe with	12	4 (33.3)	6 (50.0)
complications			
ESCMID			
Mild/moderate	71	12 (16.9)	20 (28.2)
Severe	14	4 (28.6)	6 (42.9)

Legend for Table 4: In this table we present hospital length of stay and hospital mortality rates for patients with CDI across different severity ratings across three severity scores.

Table 5 Risk Factors for CDI

	Hazard Ratio (95% CI)	P-value
Baseline characteristics		
APACHE II (10-point increase)	0.83 (0.55, 1.25)	0.372
Time-dependent factors (in the preceding 3		
days)		
H ₂ RA and/or PPI	1.28 (0.39, 4.21)	0.686
Antibiotics	1.24 (0.50, 3.06)	0.647
Vasopressors or inotropes	1.09 (0.57, 2.11)	0.790

Legend for Table 5: In this table we present the Cox proportional hazards regression analysis results with incident *CDI* diagnosed in ICU as the outcome (study day 3 or later per main trial definition). This model includes n=2624 patients with 40 events. Patients with prevalent *C diff* infection (n=26) are excluded from this analysis. Because there are so few events, neither randomized group nor stratification by medical/surgical/trauma admitting diagnosis or center.

Table 6: Association of CDI and Mortality

	Patients without CDI n=2564	Patients With CDI n=77	Total n=2641	Adjusted Hazard Ratio (95% CI)	P- value
Death in hospital, N patients (%)	718 (28.0)	21 (27.3)	739 (28.0)	0.97 (0.58, 1.63)	0.916
Sensitivity Analysis: Excluding patients with prevalent <i>CDI</i>	Patients without CDI n=2564	Patients with CDI n=60	Total n=2624	Adjusted Hazard Ratio (95% CI)	P- value
Death in hospital, N patients (%)	718 (28.0)	16 (26.7)	734 (28.0)	1.15 (0.63, 2.11)	0.645

Legend for Table 6: In this table, we present the Cox regression analysis results with hospital mortality as the outcome. Patients with *C diff* infection diagnosed prior to ICU admission (n=9) are excluded from this analysis. This model includes n=2641 patients with 739 events. In this time-to-event analysis, the time starts on day of ICU admission. This analysis is adjusted for APACHE II score and randomized group, stratified by medical/surgical/trauma and center. C diff infection is entered as a time-dependent variable.

	Patients without CDI n=2564	Patients With CDI n=77	Total n=2641	P-value – unadjusted analysis*	P-value – from linear regression†
All Patients					
Duration of hospital stay in				<0.001	<0.001
days	22 (12-	42 (22-	22 (13-		
median (Q1-Q3)	40)	77)	41)		
total range	1-630	7-334	1-630		
Sensitivity Analysis: Only patients discharged alive from hospital	Patients without CDI n=1846	Patients With CDI n=56	Total n=1902	P-value – unadjusted analysis	P-value – from linear regression
Duration of hospital stay in				<0.001	<0.001
days	25 (14-	45 (29-	25 (15-		
median (Q1-Q3)	44)	84.5)	46)		
total range	2-630	8-288	2-630		

Table 7: Association of CDI on Length of Stay

Legend for Table 7: In this table, we present the association between CDI identified in the ICU (excluding patients who had CDI diagnosed pre-ICU) and the logarithm of hospital length of stay by crude comparison (* t-test performed on the log-transformed variable). We also show this association analyzed by linear regression presenting results unadjusted and adjusted for APACHE II score, randomized group, medical/surgical trauma admitting diagnosis and center, reporting unadjusted and adjusted mean differences with corresponding 95% confidence intervals (CI) and p-values. We conducted a sensitivity analysis in ICU survivors.

Datia	PROSPECT Main RCT 076 Plate #360	Study Day (dd/mm/yyyy)
Pate		
	<u>CLOSTRIDIUM DIFFICILE</u>	ADJUDICATION FORM
Com	mittee Member: DJC ED Timing of a DJD JJJ	utcome: Pre-PROSPECT randomization at apply) Post-PROSPECT randomization, in ICU Post-PROSPECT randomization, post ICU
1. P C	Please provide date of corresponding POSITIVE Clostridium difficile microbiological testing:	
2.	Which test was the Clostridium difficile based upon (P	lease check ALL that apply)?
	ELISA (enzyme-linked immunosorbent assay) [PCR (polymerase chain reaction) [Cell Culture Cytotoxicity Assay	LAMP (loop-mediated isothermal amplification) Other, specify:
3.	Clostridium difficile infection? No Yes, spanning ≥ 3 episodes of unformed stools in ≤24 hours AND rectal tube in place (hard to quantify) Costridium difficile toxin positive stool	ecify:
		momhran aug soFfig
		If present, in the
	OR Histopathological findings of pseudomembra	anous collus Adjudicators Opinion
	OR Diagnosis of toxic megacolon	IS THIS IIKERY ATTINUTABLE C. Diff Infection?
4. 3	Were any of the following present?	No Yes No Yes
	ICU admission for this reason	
	Mental status changes	
	Fever >38.5°C	
	WBC <u>></u> 35.0 or <2.0 x 10 ⁹ /L	
	Lactate > 2.2 mmol/L	
	Septic shock (hypotension with vasopresso	ors) 🗌 🗌 🗕 🛏 🔲 🔲
	lleus or significant abdominal distention	
	Toxic megacolon	
	Bowel perforation	
	End organ failure (i.e., new mechanical ventilation, dialysis)	
5.	Laboratory (If not available on day of event, record w	orst value 48 hours pre to 48 hours post event):
uired /e for <i>le</i>	Highest WBC count 10 ⁹ /L <u>on day of event</u>	Baseline creatinine (umol/L)
ults req f positiv C. diffici	Highest lactate (mmol/L)	Highest creatinine (umol/L) on day of event
Resi only i O	Lowest serum albumin (g/L) <u>on day of event</u>	
6.		n the Adjudicators Opinion (Check one ONLY)
	Not <i>C diff</i> infection Mild to Moderate	Severe Severe with complication (see
	And, <u>if applicable</u> indicate: (Check one ONLY)	

PR Patient ID	OSPECT Main RCT 076 Plate	#361 tient Date of fitials Study Day	Study Day (dd/mm/yyyy) 201 1
	<u>CLOSTRIDIUM</u>	DIFFICILE - ADJUDICATION	FORM
Committee 7. Treati	e Member: X DJC ED ment (ever for this event) Antibiotic therapy started, specify Fecal transplant Colectomy Randomized CDI treatment trial, s	JD ☐ JJ PO IV ene PO IV ene vancomycin ☐ ☐ fidaxomicin ☐ ☐ specify:	ema NONE N/A Other, specify:
8. Clost	ridium difficile Infection Severity	(Check ALL that apply): 🗌 N/A N	OT a C. Difficile Infection
	SHEA	ACG	ESCMID
Mild - Moderate	SHEA - Mild to Moderate (check all that apply):	ACG - Mild to Moderate (check all that apply):	ESCMID - Mild to Moderate (check all that apply):
	WBC ≤ 15x10 ⁹ /L AND Creatinine < 1.5 X premorbid level	Diarrhea PLUS signs and symptoms NOT meeting the criteria for severe or complicated	A clinical picture compatible with CDI and microbological evidence of free toxins and the presence of <i>C. Diff</i> in stool without reasonable evidence of another cause of diarrhea
_	SHEA - Severe	ACG- Severe	Pseudommembraneous colitis as diagnosed during endo- scopy, after collectomy or on autopsy ESCMID - Severe
Severe	 (check all that apply): WBC ≥ 15x10⁹/L OR Creatinine ≥ 1.5 X premorbid level 	(check all that apply): Serum albumin <30g/L PLUS one of the following: WBC ≥ 15x10 ⁹ /L Abdominal tenderness	 (check all that apply): An episode of CDI with (one or more specific signs and symptoms of) severe colitis [see table] or a complicated course of disease with significant systemic toxin effects and shock, resulting in the need for ICU admission, colectomy or death
Severe with Complication	SHEA - Severe with complication (check all that apply): Hypotension Shock leus Megacolon	ACG - Severe with complication (check all that apply): Any of the following attributable to CDI: Hypotension with vasopressors Fever ≥ 38.5°C leus or significant abdominal distention Mental status changes WBC ≥ 35x10 ⁹ /L or <2x10 ⁹ /L Lactate > 2.2 mmol/L Fnd-organ failure (i.e., new	
		Lactate > 2.2 mmol/L End-organ failure (i.e., new mechanical ventilation, dialysis)	

9. Comments:



Category	Signs/symptoms				
Physical examination	Fever (core body temperature >38.5°C). Rigors (uncontrollable shaking and a feeling of cold followed by a rise in body temperature). Haemodynamic instability including signs of distributive shock.				
	Respiratory failure requiring mechanical ventilation.				
	Signs and symptoms of peritonitis.				
	Signs and symptoms of colonic ileus.				
	Admixture of blood with stools is rare in				
	Clostridium difficile infection (CDI) and the				
1.1	correlation with severity of disease is uncertain.				
Laboratory investigations	Marked leucocytosis (leucocyte count >15 × 107L). Marked left shift (band neutrophils >20% of leucocytes).				
	Rise in serum creatinine (>50% above the baseline).				
	Elevated serum lactate (≥5 mM).				
	Markedly reduced serum albumin (<30 g/L).				
Colonoscopy or	Pseudomembranous colitis.				
sigmoidoscopy	There is insufficient knowledge on the				
	correlation of endoscopic findings compatible with CDI, such as oedema, erythema, friability				
land the second s	and ulceration, and the severity of disease.				
Imaging	Distension of large intestine (>6 cm in transverse width of colon)				
	Colonic wall thickening including low-				
	attenuation mural thickening				
	Pericolonic fat stranding				
	Ascites not explained by other causes.				
	The correlation of haustral or mucosal				
	thickening, including thumborinting.				
	pseudopolyps and plaques, with severity of				
	disease is unclear				

Patient ID	PROSPECT Main RCT 076	Plate #362 Patient Initials	Date of Study Day		(dd/mm/yyyy)
Commit	ttee Member: DJC X	ED Timing of (Check all the JJ	- ADJUDICA outcome: at apply)	Pre-PROSPEC Post-PROSPEC Post-PROSPE	<u>1</u> CT randomization CT randomization, in ICU CT randomization, post ICU
1. Ple <i>Cl</i> o	ase provide date of corresp stridium difficile microbiol	oonding POSITIVE ogical testing:	Date (d	id/mm/yyyy)	1
2. W	hich test was the <i>Clostridiu</i> ELISA (enzyme-linked i PCR (polymerase chain Cell Culture Cytotoxicity	<i>m difficil</i> e based upon (l immunosorbent assay) n reaction) y Assay	Please check A LAMP (loc Other, spe	LL that apply) op-mediated iso cify:	? thermal amplification)
3. <i>Ck</i>	Destridium difficile infection Destridium difficile infection Destridium difficile toxio Clostridium difficile toxio	? ☐ No ☐ Yes, sp ed stools in ≤24 hours e (hard to quantify) n positive stool	ecify:		
	R Colonscopic findi R Histopathological R Diagnosis of toxic	ings demonstrating pseud I findings of pseudomembr c megacolon	omembranous c anous colitis	olītis A is	If present, in the djudicators Opinion this likely attributable <i>C. Diff</i> Infection?
4. We	ere any of the following pre ICU admission fo	esent? or this reason	N F	lo Yes 7 1	No Yes ► □ □
	Mental status cha Fever >38.5ºC WBC ≥35.0 or ≪ Lactate > 2.2 mn Septic shock (hy	anges 2.0 x 10 ⁹ /L nol/L potension with vasopress	[[[[] []		▶ □ □ ▶ □ □ ▶ □ □ ▶ □ □
	lleus or significa Toxic megacolon Bowel perforatio End organ failure ventilation, dialys	nt abdominal distention n e (i.e., new mechanical sis)			
auired the for	boratory (If not available o Highest WBC count 10 ⁹ /L <u>on day of event</u>	n day of event, record w	Baseline cr	hours pre to 4 reatinine (umol	8 hours post event): /L)
Results rec only if positi C. <i>diffi</i> c	Highest lactate (mmol/L) on day of event	Lowest serum albumin (g/L) on day of event	Highest cre on day of e	eaunine (umol/l <u>vent</u>	->
	ostridium difficile Infection	Severity Classification,	in the Adjudica	tors Opinion (Check one ONLY)
0. CA				. –	

PR	OSPECT Main RCT 076	#363	Study Day (dd/mm/yyyy)
Patient ID		itials Date of Study Day	2 0 1
Committee	e Member: DJC 🗶 ED]]D []]]]	
7. Treat	ment (ever for this event) Antibiotic therapy started, specify Fecal transplant Colectomy	PO IV ene □ metronidazole □ ▶ □ vancomycin □ fidaxomicin	ema NONE N/A Other, specify:
	Randomized CDI treatment trial,	specify:	
8. Clost	ridium difficile Infection Severity	(Check ALL that apply): N/A N	OT a C. Difficile Infection
	SHEA	ACG	ESCMID
Mild -	SHEA - Mild to Moderate	ACG - Mild to Moderate	ESCMID - Mild to Moderate (check all that apply):
Moderate Severe	 (check all that apply): WBC ≤ 15x10⁹/L AND Creatinine < 1.5 X premorbid level SHEA - Severe (check all that apply): WBC ≥ 15x10⁹/L OR Creatinine ≥ 1.5 X premorbid level 	 □ (check all that apply): □ Diarrhea PLUS signs and symptoms NOT meeting the criteria for severe or complicated □ ACG- Severe (check all that apply): □ Serum albumin <30g/L PLUS one of the following: □ WBC ≥ 15x10⁹/L □ Abdominal tenderness 	 Check all that apply). A clinical picture compatible with CDI and microbological evidence of free toxins and the presence of <i>C. Diff</i> in stool without reasonable evidence of another cause of diarrhea OR Pseudommembraneous colitis as diagnosed during endoscopy, after collectomy or on autopsy ESCMID - Severe (check all that apply): An episode of CDI with (one or more specific signs and symptoms of) severe colitis [see table] or a complicated course of disease with significant systemic toxin effects and shock, resulting in the need for ICU admission, colectomy or death
Severe with Complication	SHEA - Severe with complication (check all that apply): Hypotension Shock leus Megacolon	ACG - Severe with complication (check all that apply): Any of the following attributable to CDI: Hypotension with vasopressors Fever ≥ 38.5°C leus or significant abdominal distention Mental status changes	
		WBC ≥ 35x10 ⁹ /L or <2x10 ⁹ /L Lactate > 2.2 mmol/L End-organ failure (i.e., new mechanical ventilation, dialysis)	

9. Comments:



Category	Signs/symptoms		
Physical examination	Fever (core body temperature >38.5°C). Rigors (uncontrollable shaking and a feeling of cold followed by a rise in body temperature). Haemodynamic instability including signs of distributive shock.		
	Respiratory failure requiring mechanical ventilation.		
	Signs and symptoms of peritonitis.		
	Signs and symptoms of colonic ileus.		
	Admixture of blood with stools is rare in		
	Clostridium difficile infection (CDI) and the		
1.1	correlation with severity of disease is uncertain.		
Laboratory investigations	Marked leucocytosis (leucocyte count >15 × 107L). Marked left shift (band neutrophils >20% of leucocytes).		
	Rise in serum creatinine (>50% above the baseline).		
	Elevated serum lactate (≥5 mM).		
	Markedly reduced serum albumin (<30 g/L).		
Colonoscopy or	Pseudomembranous colitis.		
sigmoidoscopy	There is insufficient knowledge on the		
	correlation of endoscopic findings compatible with CDI, such as oedema, erythema, friability		
land the second s	and ulceration, and the severity of disease.		
Imaging	Distension of large intestine (>6 cm in transverse width of colon)		
	Colonic wall thickening including low-		
	attenuation mural thickening		
	Pericolonic fat stranding		
	Ascites not explained by other causes.		
	The correlation of haustral or mucosal		
	thickening, including thumborinting.		
	pseudopolyps and plaques, with severity of		
	disease is unclear		

PROSPECT Main RCT 076 Plate #364	Study Day
	(dd/mm/yyyy)
Patient 1 Patient ID 1 Initials	Study Day 2 0 1
Committee Member: DJC ED Timing of a	utcome: Pre-PROSPECT randomization Pre-PROSPECT randomization
	Post-PROSPECT randomization, post ICU
1. Please provide date of corresponding POSITIVE <i>Clostridium difficil</i> e microbiological testing:	
2. Which test was the Clostridium difficile based upon (P	lease check ALL that apply}?
ELISA (enzyme-linked immunosorbent assay) [PCR (polymerase chain reaction) [Cell Culture Cytotoxicity Assay	LAMP (loop-mediated isothermal amplification) Other, specify:
3 Clostridium difficile infection? 🗌 No 🗌 Yes so	ecify:
> 3 episodes of unformed stools in <24 hours	
AND rectal tube in place (hard to quantify)	
Clostridium difficile toxin positive stool	
	If present, in the
OR Histopathological findings of pseudomembra	anous colitis Adjudicators Opinion
OR Diagnosis of toxic megacolon	is this likely attributable
4. Were any of the following present?	No Yes No Yes
ICII admission for this reason	
Montal status changes	
Fever >38.5°C	
WBC >35.0 or <2.0 x $10^{9}/$	
actate > 2.2 mmol/	
Santis shock (hypotansion with vacanassa	
Septic shock (hypotension with vasoplesso	
lieus or significant abdominal distention	
Bowel perforation	
End organ failure (i.e., new mechanical ventilation dialysis)	
5 aboratory (if not available on day of event record w	orst value 48 hours pre to 48 hours post evently
- Highost WPC count	
10 ⁹ /L <u>on day of event</u>	Baseline creatinine (umol/L)
Or day of event	on day of event
in 2 G (g/L) <u>on day of event</u>	
6 Clostridium difficile Infection Severity Classification in	n the Adjudicators Opinion (Check one ONLY)
	List, questions 4 and 5)

|

Patient [ID [Committee 7. Treat	OSPECT Main RCT 076 Plate OSPECT Main RCT 076 Plate Pa In CLOSTRIDIUM Member: DJC ED X ment (ever for this event) Antibiotic therapy started, specify Fecal transplant	#365 ttient Date of titials Study Day DIFFICILE - ADJUDICATION JD JJ metronidazole O IV ene metronidazole [tidaxomicin [Study Day (dd/mm/yyyy) 201 FORM Ima NONE N/A Other, specify:
	Colectomy Randomized CDI treatment trial, s	specify:	
8. Clost	ridium difficile Infection Severity	(Check ALL that apply): N/A N	OT a C. Difficile Infection
	SHEA	ACG	ESCMID
Mild -	SHEA - Mild to Moderate	ACG - Mild to Moderate	ESCMID - Mild to Moderate (check all that apply):
Moderate Severe	 WBC ≤ 15x10⁹/L AND Creatinine < 1.5 X premorbid level SHEA - Severe (check all that apply): WBC ≥ 15x10⁹/L OR 	Check all that apply): Diarrhea PLUS signs and symptoms NOT meeting the criteria for severe or complicated ACG- Severe (check all that apply): Serum albumin <30g/L PLUS one of the following:	 A clinical picture compatible with CDI and microbological evidence of free toxins and the presence of <i>C. Diff</i> in stool without reasonable evidence of another cause of diarrhea OR Pseudommembraneous colitis as diagnosed during endoscopy, after collectomy or on autopsy ESCMD - Severe (check all that apply): An episode of CDI with (one or more specific signs and symptoms of) severe colitis [see
	☐ Creatinine ≥ 1.5 X premorbid level	☐ WBC ≥ 15x10 ⁹ /L ☐ Abdominal tendemess	table] or a complicated course of disease with significant systemic toxin effects and shock, resulting in the need for ICU admission, colectomy or death
Severe with Complication	SHEA - Severe with complication (check all that apply): Hypotension Shock leus Megacolon	ACG - Severe with complication (check all that apply): Any of the following attributable to CDI: Hypotension with vasopressors Fever ≥ 38.5°C leus or significant abdominal distention Mental status changes WBC ≥ 35x10°/L or <2x10°/L Lactate > 2.2 mmol/L End-organ failure (i.e., new mechanical ventilation, dialysis)	

9. Comments:



Category	Signs/symptoms		
Physical examination	Fever (core body temperature >38.5°C). Rigors (uncontrollable shaking and a feeling of cold followed by a rise in body temperature). Haemodynamic instability including signs of distributive shock.		
	Respiratory failure requiring mechanical ventilation.		
	Signs and symptoms of peritonitis.		
	Signs and symptoms of colonic ileus.		
	Admixture of blood with stools is rare in		
	Clostridium difficile infection (CDI) and the		
	correlation with severity of disease is uncertain.		
Laboratory investigations	Marked leucocytosis (leucocyte count $>15 \times 10^{-1}$ L).		
	leucocytes).		
	Rise in serum creatinine (>50% above the baseline).		
	Elevated serum lactate (≥5 mM).		
	Markedly reduced serum albumin (<30 g/L).		
Colonoscopy or	Pseudomembranous colitis.		
sigmoidoscopy	There is insufficient knowledge on the		
	correlation of endoscopic findings compatible with CDI, such as oedema, erythema, friability		
	and ulceration, and the severity of disease.		
Imaging	Distension of large intestine (>6 cm in transverse width of colon)		
	Colonic wall thickening including low-		
	attenuation mural thickening		
	Pericolonic fat stranding.		
	Ascites not explained by other causes.		
	The correlation of haustral or mucosal		
	thickening, including thumbprinting,		
	pseudopolyps and plaques, with severity of		
	disease is unclear		

	PROSPECT Main RCT 076 Plate #366 (dd/mm/yyyy)
Pati IC	ent 1 Patient Date of Study Day 201
Com	Imitige Member: DJC ED Timing of outcome: Pre-PROSPECT randomization JD X JJ Check all that apply) Post-PROSPECT randomization, in ICU
1. (Please provide date of corresponding POSITIVE Date (dd/mm/yyyy) Clostridium difficile microbiological testing: 201
2.	Which test was the Clostridium difficile based upon (Please check ALL that apply)?
	ELISA (enzyme-linked immunosorbent assay) LAMP (loop-mediated isothermal amplification) PCR (polymerase chain reaction) Other, specify: Cell Culture Cytotoxicity Assay Other, specify:
3.	Clostridium difficile infection? No Yes, specify:
	Clostridium difficile toxin positive stool
	OR Colonscopic findings demonstrating pseudomembranous colitis
	OR Histopathological findings of pseudomembranous colitis
	OR Diagnosis of toxic megacolon is this likely attributable C. Diff Infection?
4.	Were any of the following present? No Yes No Yes
	ICU admission for this reason
	Mental status changes
	$Fever > 38.5 \circ C \qquad \qquad \Box \qquad \Box \qquad \blacksquare \qquad \blacksquare \qquad \Box \qquad \Box$
	$WBC \ge 35.0 \text{ of } < 2.0 \times 10^{9} \text{L}$
	End organ failure (i.e., new mechanical ventilation, dialysis)
5.	Laboratory (if not available on day of event, record worst value 48 hours pre to 48 hours post event):
auired ive for	Highest WBC count 10 ⁹ /L on day of event Baseline creatinine (umol/L)
iults rec if positi C. <i>diffic</i>	Highest lactate (mmol/L)
Res	(g/L) on day of event
6.	<i>Clostridium difficil</i> e Infection Severity Classification, in the Adjudicators Opinion (Check one ONLY)
	Not C diff infection Mild to Moderate Severe Severe with complication (see
	And, if applicable indicate: (Check one ONLY)
	Recurrence (within 8 weeks of last <i>C diff episode</i> providing symptoms resolved after treatment)

PR	OSPECT Main RCT 076	#367	Study Day
Patient ID	1 Pa	itials Study Day	
	CLOSTRIDIUM	DIFFICILE - ADJUDICATION	FORM
0			
Committee	e member: UDJC UED U		1ma
	Antibiotic therapy started, specify Fecal transplant Colectomy	metronidazole metronidazole vancomycin fidaxomicin	NONE N/A
8 Clast	Randomized CDI treatment thal, s	specity:	IOT a C. Difficile Infection
	SHEA		ESCND
	SHEA - Mild to Moderate	ACG - Mild to Moderate	ESCMID - Mild to Moderate
Mild - Moderate	(check all that apply):	(check all that apply):	(check all that apply):
	WBC ≤ 15x10 ⁹ /L AND Creatinine < 1.5 X premorbid level	Diarrhea PLUS signs and symptoms NOT meeting the criteria for severe or complicated	A clinical picture compatible with CDI and microbological evidence of free toxins and the presence of C. Diff in stool without reasonable evidence of another cause of diarrhea OR
			Pseudommembraneous colitis as diagnosed during endo- scopy, after collectomy or on autopsy
Severe	SHEA - Severe (check all that apply):	ACG- Severe (check all that apply):	ESCMID - Severe (check all that apply):
	WBC ≥ 15x10 ⁹ /L OR Creatinine ≥ 1.5 X premorbid level	Serum albumin <30g/L PLUS one of the following: □ WBC ≥ 15x10 ⁹ /L □ Abdominal tendemess	An episode of CDI with (one or more specific signs and symptoms of) severe colitis [see table] or a complicated course of disease with significant systemic toxin effects and shock, resulting in the need for ICU admission, colectomy or death
Savara with	SHEA - Severe with	ACG - Severe with	
Complication	(check all that apply):	(check all that apply):	
	Hypotension Shock Eus Megacolon	Any of the following attributable to CDI: Hypotension with vasopressors Fever ≥ 38.5°C leus or significant abdominal distention Mental status changes WBC ≥ 35x10 ⁹ /L or <2x10 ⁹ /L Lactate > 2.2 mmol/L End-organ failure (i.e., new	

9. Comments:



Category	Signs/symptoms		
Physical examination	Fever (core body temperature >38.5°C). Rigors (uncontrollable shaking and a feeling of cold followed by a rise in body temperature). Haemodynamic instability including signs of distributive shock.		
	Respiratory failure requiring mechanical ventilation.		
	Signs and symptoms of peritonitis.		
	Signs and symptoms of colonic ileus.		
	Admixture of blood with stools is rare in		
	Clostridium difficile infection (CDI) and the		
	correlation with severity of disease is uncertain.		
Laboratory investigations	Marked leucocytosis (leucocyte count $>15 \times 10^{-1}$ L).		
	leucocytes).		
	Rise in serum creatinine (>50% above the baseline).		
	Elevated serum lactate (≥5 mM).		
	Markedly reduced serum albumin (<30 g/L).		
Colonoscopy or	Pseudomembranous colitis.		
sigmoidoscopy	There is insufficient knowledge on the		
	correlation of endoscopic findings compatible with CDI, such as oedema, erythema, friability		
	and ulceration, and the severity of disease.		
Imaging	Distension of large intestine (>6 cm in transverse width of colon)		
	Colonic wall thickening including low-		
	attenuation mural thickening		
	Pericolonic fat stranding.		
	Ascites not explained by other causes.		
	The correlation of haustral or mucosal		
	thickening, including thumbprinting,		
	pseudopolyps and plaques, with severity of		
	disease is unclear		

Patie	PROSPECT Main RCT 076	Plate #368	Date of	(dd/m	Study Day (m/yyyy) 201
ті (С 1. рі С	iming of outcome: Pre-F iming of outcome: Pre-F CLOST Post- Check all that apply) Post- Post- Post- Issue provide date of correspondence Post- Issue provide	PROSPECT randomization PROSPECT randomization PROSPECT randomization PROSPECT randomization onding POSITIVE gical testing:	- CONSENSU , in ICU , post ICU Date (dd/	mm/yyyy) 201]
	ELISA (enzyme-linked in PCR (polymerase chain Cell Culture Cytotoxicity	nmunosorbent assay) reaction) Assay	LAMP (loop-r	nediated isothern y:	nal amplification)
3. C	Clostridium difficile infection? 2 3 episodes of unforme Image: Stridium difficile in place Image: Stridium difficile i	No Yes, s d stools in ≤24 hours (hard to quantify) positive stool ags demonstrating pseud findings of pseudomemb megacolon	pecify: Iomembranous colit ranous colitis	is If pr Adjudi is this I	esent, in the icators Opinion ikely attributable
4. V	Vere any of the following pres ICU admission for Mental status cha Fever >38.5°C WBC >36.0 or <2	sent? r this reason nges 0 × 10 ⁹ /	No 		
	Lactate > 2.2 mm Septic shock (hyp Ileus or significan Toxic megacolon Bowel perforation	ol/L otension with vasopress t abdominal distention	юrs)		
5. L	End organ failure ventilation, dialysi aboratory (If not available on	(i.e., new mechanical s) I day of event, record 1	worst value 48 ho	urs pre to 48 hou	urs post event):
Results required only if positive for <i>C. difficile</i>	Highest WBC count 10 ⁹ /L <u>on day of event</u> Highest lactate (mmol/L) <u>on day of event</u>	Lowest serum albumin (g/L) on day of event	Baseline creat Highest creati <u>on day of eve</u>	tinine (umol/L) nine (umol/L) 	
6. C	Clos <i>tridium difficil</i> e Infection S	Severity Classification,	in the A djudic ato	rs Opinion (Chec	k one ONLY)
	Not C diff infection	Mild to Moderate	Severe	Severe wi	ith complication (see ions 4 and 5)
A	And, <u>if applicable</u> indicate: (Ch Recurrence (within 8 week	eck one ONLY) s of last C diff episode	Relanse (sa	ame proven strain	\ \

PR Patient ID	ROSPECT Main RCT 076 Plate 1 1	#369 (tient Date of Study Day	Study Day (dd/mm/yyyy) 201 1
7. Treat	<u>CLOSTRIDIU</u> ment (ever for this event) Antibiotic therapy started, specify Fecal transplant Colectomy Randomized CDI treatment trial, s	M DIFFICILE - CONSENSUS F PO IV ene metronidazole [] [] vancomycin [] fidaxomicin [] specify:	EORM_ ema NONE N/A Other, specify:
8. Clost	ridium difficile Infection Severity	(Check ALL that apply): 🗌 N/A N	OT a C. Difficile Infection
	SHEA	ACG	ESCMID
Mild - Moderate	SHEA - Mild to Moderate (check all that apply):	ACG - Mild to Moderate (check all that apply):	ESCMID - Mild to Moderate (check all that apply): A clinical picture compatible with CDI and microbological evidence of free toxins and
	Creatmine < 1.5 X	the criteria for severe or complicated	the presence of <i>C</i> . <i>Diff</i> in stool without reasonable evidence of another cause of diarrhea OR Pseudommembraneous colitis as diagnosed during endo- scopy, after collectomy or on autopsy
Severe	SHEA - Severe (check all that apply): WBC ≥ 15x10 ⁹ /L OR Creatinine ≥ 1.5 X premorbid level	ACG- Severe (check all that apply): Serum albumin <30g/L PLUS one of the following: WBC ≥ 15x10 ⁹ /L Abdominal tendemess	ESCMID - Severe (check all that apply): An episode of CDI with (one or more specific signs and symptoms of) severe colitis [see table] or a complicated course of disease with significant systemic toxin effects and shock, resulting in the need for ICU admission, colectomy or death
Severe with Complication	SHEA - Severe with complication (check all that apply): Hypotension Shock leus Megacolon	ACG - Severe with complication (check all that apply): Any of the following attributable to CDI: Hypotension with vasopressors Fever ≥ 38.5°C lieus or significant abdominal distention Mental status changes WBC ≥ 35x10 ⁹ /L or <2x10 ⁹ /L Lactate > 2.2 mmol/L End-organ failure (i.e., new mechanical ventilation, dialysis)	

9. Comments: ____

Chapter 6 - Conclusions

6.1 Background

This doctoral thesis focuses on diarrhea, bowel protocols and CDI in critically ill patients, which are relatively under-investigated topics in critical care medicine (Table 1). The data presented herein describe the epidemiology of diarrhea and CDI, demonstrating the incidence, prevalence, and outcomes. Evidence on the prevalence of bowel protocols used in a sample of ICUs is also presented, which may precipitate or perpetuate diarrhea or lead to CDI testing.

6.2 Diarrhea during Critical Illness

Chapter 2 describes the methodology for the DICE-ICU study, and chapter 3 outlines the results of this multicenter observational study of diarrhea in the critically ill. This 1109 patient study showed that diarrhea was common, varied based on the definitions applied, and risk factors are potentially modifiable.

The strengths of this study include the large sample size and its multinational sample enhancing the generalizability of the results. The full protocol and statistical analysis were transparently published in a peer review journal. However, the limitations include possible observer bias, possible uncontrolled confounding, and the lack of mechanistic data.

The next step in this research program will be to design an interventional study to decrease the incidence of diarrhea, a nested observational study to evaluate microbiome alterations in the setting of diarrhea, and generation of a definition of diarrhea that can easily and consistently be applied at the bedside.

6.4 Challenges in studying diarrhea during critical illness

Conducting studies of gastrointestinal dysfunction, specifically diarrhea, during critical illness has several methodological challenges. Compared to other physiologic dysregulation that occurs in critical illness, diarrhea and gastrointestinal dysfunction has been under investigated, leading to many scientific problems.

Defining what constitutes diarrhea in this population is a challenge, as clinicians often do not know of pre-existing bowel habits of their patients, therefore, determining a deviation from a patient's norm difficult. There is also a dearth of research of how the gastrointestinal system alters and adapts during critical illness from clinical symptoms, pathophysiologic changes and how this integrates into a patient's presentation. Highlighting the need to not only address the clinical research gap of diarrhea as well as gastrointestinal dysfunction of this population but also bridge the gap in basic science research of mechanisms.

The other challenge is the lack of an accepted universal definition of diarrhea in the critically ill, as presented in chapter 1 and chapter 2. Without a gold standard definition of diarrhea, conflicting results have been observed on the frequency and the impact of diarrhea in this patient population. For DICE-ICU we chose a definition of diarrhea, the WHO definition, that encompassed frequency and consistency. To ensure that this definition was applied consistently at the bedside, in-services were held with bedside nurses to overcome this barrier. In future studies of diarrhea, a consistent definition should be implemented to allow better characterization and determination of the true impact of diarrhea on patient important outcomes and facilitate meta-analysis of results.

6.5 Implications of the use of bowel protocols in the ICU

In chapter 4 the results of a content analyses of bowel protocols in 44 ICUs are presented. The study showed that most ICUs have a bowel protocol, and these protocols varied in terms of initiation criteria, medications, escalation, and discontinuation criteria. However, rarely did the protocols include stool assessment which would be ideal to guide the activation and discontinuation of these protocols in practice.

The strengths of this analysis include the broad cross section of ICUs encompassing both community and academic centers. The analysis also included robust characterizations of bowel protocols from implementation of protocols, pharmacotherapy, and contraindications. However, limitations of this study include a selection bias in that included centers were engaged in a randomized trial of probiotics in the critically ill. This study did not explore how institutional protocols were developed or implemented at the bedside.

The next step in this research program is to develop an evidenced based bowel protocol with multidisciplinary input using medications studied in the critically ill, with refined initiation, escalation and discontinuation criteria, incorporating the diarrhea definition used in DICE-ICU, compared to usual care, assessing the impact of the use of the bowel protocol on patient important outcomes. This will be explored first in a pilot

trial to determine feasibility, followed by a large RCT to determine the efficacy of the protocol.

6.6 Challenges in examining bowel protocols in the ICU

As highlighted in chapter 4, the variability in bowel protocols in the ICU is high. There are multiple challenges in developing and researching bowel protocols. Firstly, the protocols need to not only contain medications, but also instructions on how to implement them (e.g., when, how, who etc). Secondly, instructions need to be clearly actionable by bedside nurses to ensure that the protocol is used appropriately and safely. Thirdly, the protocols would ideally be based on evidence (physiologic rationale, or clinical research); however, very few studies have examined the efficacy of a small number of medications in the critically ill. Lastly, studying a bowel protocol as a multifaceted intervention is challenging because there are so many components, not all of which are used in each patient. That is, the way these protocols are used algorithmically means that when they are initiated, not all nodes or directives are activated; this complexity makes it challenging to evaluate bowel protocols in the realworld setting.

6.7 CDI in the critically ill

Chapter 5 contains the results of an observational cohort study of CDI in ICU patients nested within a clinical trial examining the timing, severity, and treatment for CDI. This study demonstrated that CDI is relatively uncommon in the ICU, and if acquired during critical illness, CDI is mild to moderate in severity.

This study had several strengths including a prospectively enrolled population and rigorous methodology including blinded adjudications and the use of three definitions of CDI severity. However, participants in this study were enrolled in a randomized trial requiring them to be mechanically ventilated, which may limit the generalizability of the results. Given the small number of patients with CDI in the cohort, further analyses of CDI risk factors could not be explored.

The next step in this research program will include development of a CDI ICUspecific scoring system and a prospective cohort study to better examine risk factors for CDI in the critically ill patient population.

6.8 Challenges in investigating CDI in the critically ill

As summarized in chapter 5, studying CDI in the ICU has many challenges. For example, prior CDI studies have either used solely microbiological definitions or definitions with minimal clinical data, as following the daily clinical data can be labour intensive and costly. During this study of CDI, new and expanded methods for microbiological confirmation for CDI were developed (e.g. toxinogenic culture, cell cytotoxicity neutralization assay, toxin A/B, nucleic acid amplification tests), which may or may not influence studies describing the prevalence and incidence of this infection. Also, clinical practice guidelines changed twice for treatment, which may influence reported recurrence rates and severity assessments.

Determining the severity of CDI in the critically ill is difficult, as many markers of CDI disease severity are inherent to the population itself. It would be impossible to

judge true CDI severity without the context of the patients' clinical course, such that detailed clinical data and adjudication are needed.

6.9 Future directions

The studies comprising this doctoral thesis have provided the foundation for future work, as outlined in earlier sections. A more intermediate step at the nexus of critical care and gastroenterology clinical research is the future conduct of a large RCT testing the continuous versus bolus delivery of enteral nutrition in the critically ill and its impact on diarrhea and feeding tolerance in this patient population. Embedded in this trial will be a descriptive translational study examining the microbiota in the critically ill, to better characterize the dysbiosis that occurs in patients who develop diarrhea compared to those who do not develop diarrhea.

Table 1: Overview of chapters and included studies

Chapter	Population/Centers	Objective	Methodology	Status of
				Manuscript
Chapter 1	N/A	N/A	N/A	N/A
Introduction				
Chapter 2 DICE-ICU Protocol Paper	Critically ill adults 18 years of age and older admitted to ICU for greater than 24 hours	The objectives of this study were to determine the incidence and frequency of diarrhea, risk factors for diarrhea and consequences (ICU and hospital mortality, ICU and hospital length of stay) of diarrhea in critically ill adults and validate different stool classification systems.	Prospective Cohort Study Protocol	Published in BMJ Open
Chapter 3 DICE-ICU Results Paper	1109 critically ill adults 18 years of age and older admitted to ICU for greater than 24 hours in 12 centers (8 centers in Canada, 2 in the US, 1 in Poland, and 1 in Saudi Arabia).	The objectives of this study were to determine the incidence of diarrhea defined using the WHO criteria, including the incidence of <i>Clostridioides difficile</i> associated diarrhea (CDAD), to compare the incidence and definitions of diarrhea using the Bristol Stool Chart and Bliss Classification System, to identify diarrhea risk factors,	Prospective Cohort Study Results Report	Submitted

		and to describe the management modifications, consequences, and clinical outcomes associated with		
		diarrhea		
Chapter 4 Bowel Protocols in the Critically III	44 Centers in Canada, US and Saudi Arabia	The objective of this study was to characterize the content of bowel protocols for the prevention of constipation implemented in ICUs in centers that participated in a randomized trial examining the effects of probiotics for critically ill patients to prevent ICU-acquired infections [Probiotics: Prevention of Severe Pneumonia and Endotracheal Colonization Trial (PROSPECT) <u>NCT02462590</u>]. Our secondary objective was to characterize: 1) initiation criteria; 2) medications incorporated; 3) approaches to protocol escalation; 4) discontinuation criteria; 5) stool assessment methods; and 6) contraindications.	Content Analysis of Hospital Documents	Published in the Journal of Critical Care
Chapter 5	86 mechanically	The objectives of this	Nested	Soon to be
	ventilated patients	study were to 1)	Prospective	
<i>Clostridioides</i> <i>difficle</i> infection in	with CDI among 2650 critically ill patients in 44	analyze the incidence and prevalence of CDI in the ICU, 2)	Cohort Study	Submitted

Mechanically Ventilated Critically III Patients	centers in Canada, the US and Saudi Arabia	describe the timing of CDI infection (pre- ICU, in ICU, and post-ICU), 3) assess the severity of CDI infection using 3 scoring systems and the agreement among scores, 4) document CDI treatments used and 5) examine the outcomes of patients with compared to without CDI.		
Chapter 5	N/A	N/A	N/A	N/A
Conclusion				